# SHORT TERM EFFECTS OF HYPOXIA AND ANOXIA ON SEDIMENT

### BIOGEOCHEMISTRY

WIDAD BINTI FADHULLAH

A thesis submitted to the University of East Anglia for the degree of

Doctor of Philosophy

School of Environmental Sciences

University of East Anglia

May 2013

©This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with the author and that use of any information derived there from must be in accordance with current UK Copyright Law. In addition, any quotation or extract must include full attribution.

#### ABSTRACT

Hypoxia and anoxia alter the behaviour, diversity and habitat of the benthic communities and cause accumulation of organic matter and nutrients. Most studies have focused on the effect of anoxia on benthic organisms and little information is available on nutrient sediment-water fluxes and rates of key biogeochemical processes. This study aimed to evaluate the oxygen concentration at which harmful effects start to occur, the rates that the system responds to different levels of oxygen saturation in the overlying water and the recovery times of system processes when re-aeration are restored. Sampling was carried out at Breydon Water, Great Yarmouth in June, 2010 and key biogeochemical processes were monitored in sediment/artificial seawater mesocosms maintained at constant temperature in the laboratory. The experimental work assessed the short term effect of different levels of oxygen by manipulating oxygen into five different treatments; oxic (96% saturation in the overlying water), hypoxia (25% saturation), one day of anoxia (24 hours without oxygen followed by re-aeration of up to 6 days), four days of anoxia (4 days without oxygen and subsequent re-aeration for 3 days) and seven days of anoxia (0% saturation). The effect of these oxygen concentrations on different biogeochemical processes (solute transport, oxygen uptake, nutrient fluxes, porewater chemistry, and enzyme activities) was quantified. Bromide used in solute transport assessments were analysed using inductively coupled plasma mass spectrometry and oxygen was measured using an oxygen minielectrode. Nutrient analysis was carried out using the SKALAR autoanalyser and enzyme activities were measured using microplate based assays. With decreasing oxygen supply, ammonium and phosphate are rapidly released into the water column, and nitrate concentrations decrease, reflecting the reducing conditions. In oxic conditions, rapid transport of water and oxygen into the sediments occurs through irrigation fluxes. Slower transport was observed from the porewater of anoxic treatments, as only of molecular diffusion occurs. Vertical profiles alkaline phosphatase, cellobiohydrolase, ß-glucosidase and chitinase activities in the sediment reflect organic matter inputs decreasing with depth. Urease activity was enhanced following anoxic incubations of 7 days, with an inverse relationship with porewater ammonium concentrations and increased total organic carbon input. The biogeochemical resilience of the system is dependent on the duration of anoxic events. The effect of 1 day anoxia is relatively reversible and the system recovers to normal conditions within the 7 days of treatment. However, hypoxia (25% oxygen saturation) or longer duration (4 days) anoxia followed by re-aeration are enough to cause negative effects on the biogeochemical functioning of the system. So, sediment biogeochemistry has some resilience towards short term anoxic events, but more prolonged hypoxia or anoxic events that are continued for only a few days can have major effects on ecosystem function.

## TABLES OF CONTENTS

### PAGE

Abstract	i
Tables of Contents	ii
List of Tables	vii
List of Figures	ix
Acknowledgements	xiv

### **CHAPTER 1: INTRODUCTION**

1.1	Estuaries - Their Ecological Importance and Exposure to Environmental Pressures	1
1.2	Hypoxia and Anoxia	2
1.3	Solute Transport and Oxygen Fluxes	6
1.4	Organic Matter	8
1.5	Nutrients in Estuaries	11
	1.5.1 Nitrogen Cycling	12
	1.5.2 Nitrogen Fixation	12
	1.5.3 Ammonification and Ammonium Assimilation	13
	1.5.4 Ammonium Oxidation	15
	1.5.5 Nitrate Reduction	15
	1.5.6 Denitrification	16
	1.5.7 Dissimilatory Nitrate Reduction To Ammonium (DNRA)	16
	1.5.8 Phosphorus Cycling	17
1.6	Enzyme Activities in Sediments as Indicators of Biogeochemical Processes	18
	1.6.1 Alkaline Phosphatase (AP)	19

	1.6.2	Cellobiohydrolase (CB)	20
	1.6.3	β-glucosidase (BG)	21
	1.6.4	Chitinase	22
	1.6.5	Urease	24
1.7	The S	ignificance and Aims of This Study	25
1.8	The S	tructure of the Thesis	26

### **CHAPTER 2: METHODOLOGY**

2.1	Introduction	28
2.2	Experimental Design	28
2.3	Sampling	28
2.4	Experimental Setup	31
2.5	Laboratory Protocol	34
2.6	Determination of Physico-Chemical Parameters	34
	2.6.1 Determination of Particle Size Distribution of Sediments	34
	2.6.2 Determination of Overlying Water pH	34
	2.6.3 Determination of Porosity in Sediments	35
	2.6.4 Determination of Total Organic Carbon and Total Nitrogen in Sediments	35
2.7	Bromide Tracer	37
	2.7.1 Total Bromide Incubation, Diffusive and Irrigation Fluxes Measurements	39
2.8	Determination of Oxygen Penetration Depth in Sediments	40
	2.8.1 Measurement of Diffusive Oxygen Fluxes in Sediments	41
2.9	Nutrients Sampling in the Overlying Seawater	41
	2.9.1 Determination of Nitrate and Nitrite Using a SKALAR Autoanalyser	42

	2.9.2	Determination of Ammonium Using a SKALAR Autoanalyser	42
	2.9.3	Determination of Phosphate Using a SKALAR Autoanalyser	43
	2.9.4	Measured Nutrient Fluxes in the Overlying Water	44
2.10	Analys	sis of Porewater Nutrients in the Sediment	44
	2.10.1	Calculated Porewater Diffusive Fluxes in the Sediment	45
2.11	Measu	rement of Enzyme Activities in the Sediment	45
	2.11.1	Linearity Test	45
	2.11.2	Saturation Test	47
2.12	Reage	nts	48
	2.12.1	Substrates	48
	2.12.2	Buffers	49
	2.12.3	Standards	49
2.13	Measu	rement of Potential Enzyme Activity	49
	2.13.1	Fluorometric Assay	50
	2.13.2	Colorimetric Assay	53
2.14	Statist	ical Analysis of Results	55

### **CHAPTER 3: RESULTS**

3.1	Introduction	56
3.2	Physico-chemical characteristics of Breydon Water, Great	56
	3.2.1 Effect of different treatment on overlying water pH during laboratory incubations	57
	3.2.2 Effect of different treatments on porosity during laboratory incubations	57
	3.2.3 Effect of different treatments on total organic carbon (TOC), total nitrogen (TN) and CN ratio taken at the start	59

#### and on day 7 of laboratory incubations

3.3	Effect of different treatments on bromide concentration and total	60
	bromide incubation fluxes in the overlying water and porewater	
	bromide concentrations and diffusive fluxes in the sediments	

- 3.4 Effect of different treatments on oxygen penetration depth and 62 oxygen diffusive fluxes taken at the start and on day 7 of laboratory incubations
- 3.5 Initial nutrient and porewater nutrient concentrations of Breydon 64 Water, Great Yarmouth
  - 3.5.1 Effect of different treatments on nutrients concentrations 65 and measured nutrient fluxes to/from the overlying water during 7 days of laboratory incubations
  - 3.5.2 Interrelationships between nutrient concentrations with 70 dissolved oxygen; and nutrient fluxes with dissolved oxygen in the overlying water during 7 days of laboratory incubations
  - 3.5.3 Effect of different treatments on porewater nutrient 75 concentrations in the sediment during 7 days of laboratory incubations
  - 3.5.4 Effect of different treatments on nutrient concentrations 78 and porewater nutrient concentrations during 7 days of laboratory incubations
  - 3.5.5 Effect of different treatments on porewater diffusive 80 fluxes in the sediment during 7 days of laboratory incubations
  - 3.5.6 Comparison between Measured Nutrient Fluxes and 83 Calculated Diffusive Fluxes
- 3.6 Enzyme Activities in the Sediments taken at the start and on day 83 7 of laboratory incubations
  - 3.6.1 Interrelationships between parameters in the sediment 87

#### **CHAPTER 4: DISCUSSIONS**

4.1	Introduction	92
4.2	Methodological Considerations	92

4.3	Sediment Physico-chemical characteristics, nutrients and enzyme activities of Breydon Water, Great Yarmouth	95
4.4	pH in the overlying water and total organic carbon (TOC) and total nitrogen (TN) in the sediment under five different treatments	99
4.5	Nutrient concentrations and measured nutrient fluxes in the overlying water under five different treatments	101
4.6	Nutrient concentrations and porewater nutrient concentrations in the sediment under five different treatments	105
4.7	Porewater nutrient concentrations and porewater diffusive fluxes; and nutrient fluxes and porewater diffusive fluxes in the sediment under five different treatments	113
4.8	Enzyme activities in the sediment under five different treatments and the relationship of enzyme activities and other parameters in the sediment	116
4.9	Conclusions	119

### **CHAPTER 5: SUMMARY AND FUTURE RECOMMENDATIONS**

5.1	Introduction	121
5.2	Summary of Findings	122
5.3	Selection of Experimental Setup and Research Timeline	125
5.4	Recommendation for future Research	126

### REFERENCES

129

### LIST OF TABLES

Table		Page
1.1	Global distribution of hypoxia in estuaries, coastal and marine ecosystems based on the type of hypoxia (adapted from Diaz and Rosenberg, 2008 and Rabalais et al., 2010)	4
2.1	Passing criteria for standard mode of ICP-MS	38
2.2	ICP-MS operating conditions for bromide measurements	38
2.3	Incubation of sample assay (SA), sample quench (SQ), reference standard (RS), quench coefficient (QC), substrate control (SuC) and sample control (SaC) of AP, CB, BG, and chitinase activities in wells	52
2.4	Incubation of SA, SuC, SaC and extinction coefficient (ExC) for urease activities in wells	54
3.1	Sediment physico-chemical characteristics from samples taken at Breydon Water, Great Yarmouth sampled at 1 to 9 cm depth in June 2010, reported as mean $\pm$ SE, n=5.	57
3.2	Porosity (g/mL) measured at 1 to 9cm depth sediments from five different treatments, oxic, hypoxic, 1DA, 4DA and anoxic taken on day 7. Samples are reported as mean $\pm$ SE, n=5.	58
3.3	Mean total bromide incubation, diffusive and irrigation fluxes ( $\mu$ mol m <sup>-2</sup> h <sup>-1</sup> ) after 3 days of incubation. Samples are reported as mean $\pm$ SE, n=5.	61
3.4	Diffusive oxygen fluxes (mmol $m^{-2} d^{-1}$ ) in the sediment calculated on day 7 of laboratory incubations using Fick's first Law of diffusion. Samples are reported as mean $\pm$ SE, n=5.	63
3.5	Initial water column nutrients and porewater nutrient concentrations ( $\mu$ mol L <sup>-1</sup> ) from samples taken at Breydon Water, Great Yarmouth. Samples are reported as mean $\pm$ SE, n=5.	64
3.6	Eigenvectors extracted based on values greater than 0.6 and above for component 1 and 2 for Principal Components Analysis	72
3.7	Factors extracted using Principal Component Analysis by fixing the number of factors to extract and suppressing factor loadings of less than 0.4.	73

3.8 Component matrices for principal component analysis of sediments in the study by extracting eigenvalues of greater than 0.6 and above and suppressing factor loadings of less than 0.4.

### LIST OF FIGURES

Figure		Page
1.1	Vertical processes of diagenetic reactions in sediments. Source: from Libes, (2009).	10
1.2	Nitrogen cycle through various redox and phase transformations. The boxes contain the nitrogen species and its oxidation number. The arrows represent transformation reactions as follows: 1) nitrogen fixation, 2) solubilisation, 3) ammonification, 4) nitrification, 5) denitrification, 6) anammox, 7) anaerobic nitrification mediated by manganese reduction, 8) DNRA, 9) assimilatory nitrogen reduction. A=anammox microbes (Source: from Libes, 2009).	14
1.3	Chitin synthesis, assimilation and product in bacteria (adapted from Beier, 2010).	24
2.1	Schematic representation of processes carried out on the cores collected from Breydon Water, Great Yarmouth in June 2010.	29
2.2	Sampling during low tide at Breydon Water, Great Yarmouth (source: EDINA Digimap)	30
2.3	Map of sampling site in Breydon Water, Great Yarmouth	31
2.4	Experimental setup for the different aerations conducted in a constant temperature room.	33
2.5	Nutrients standard calibration curve for a) nitrate, b) nitrite, c) ammonium and d) phosphate. Standards are presented as mean $\pm$ SE, n = 3.	43
2.6	Analog fluorescent as a function of incubation time for a) alkaline phosphatase, b) cellobiohydrolase, c) beta-glucosidase and d) chitinase. e) Analog absorbance as a function of incubation time for urease. Values are reported as mean $\pm$ SE, n= 5.	46
2.7	Substrate saturation curves on fluorescence intensity of a) alkaline phosphatase, b) cellobiohydrolase, c) beta-glucosidase d) chitinase and absorbance intensity of e) urease. Values are reported as mean $\pm$ SE, n= 5.	47-48
3.1	Daily hydrogen ion concentrations maintained under different oxygen concentrations in overlying water samples taken during the 7 day laboratory incubations. Samples are reported as mean $\pm$ SE, n=5. Red circles - anoxic treatments, yellow circles - four	58

days without oxygen and re-aerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.

- 3.2 Vertical profiles of a) TOC, b) TN and c) CN ratio taken at the start and on day 7 of laboratory incubations. Samples are reported as mean  $\pm$  SE, n=5. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.
- 3.3 Porewater bromide concentrations from five different 61 treatments, oxic, hypoxic, 1DA, 4DA and anoxic taken at the start and after 3 days incubation with bromide spiked into the overlying water. Values are reported as mean  $\pm$  SE, n=5. No vertical bar indicates standard error is smaller than the symbol size. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.
- 3.4 Oxygen penetration depths from five different treatments, oxic, hypoxic, 1DA, 4DA and anoxic taken at the start and on day 7 of laboratory incubations. Values are reported as mean ± SE, n=5. No vertical bar indicates standard error is smaller than the symbol size. Red circles - anoxic treatments, yellow circles four days without oxygen and re-aerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.
- 3.5 The daily evolution a) oxygen saturation, b) nitrate, c) nitrite d) ammonium and e) phosphate concentrations exposed to different oxygen treatments during the 7 day laboratory incubations. Values are reported as mean  $\pm$  SE, n=5. No vertical bar indicates standard error is smaller than the symbol size. Red circles anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days(1DA), green circles - oxic treatments.
- 3.6 Daily a) nitrate, b) nitrite, c) ammonium and d) phosphate fluxes 69 from five different treatments, oxic, hypoxic, 1DA, 4DA and anoxic. Values are reported as mean  $\pm$  SE, n=5.
- 3.7 Scatterplots of the principal component analysis of dissolved oxygen, nitrate, nitrite, ammonium phosphate and

63

59

67

72

concentrations in the overlying water according to different a) treatments and b) day. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles oxic treatments. Black square - day 1, green square - day 2, orange square – day 3, blue square – day 4, pink square – day 5, red square-day 6, grey square-day 7.

- 3.8 Scatterplots of the principal component analysis of dissolved oxygen, nitrate, nitrite, ammonium and phosphate fluxes in the overlying water according to different a) treatments and b) day. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments. Black square – day 1, green square – day 2, orange square – day 3, blue square – day 4, pink square – day 5, red square-day 6, grey square-day 7.
- 3.9 Relationships of nutrient concentrations to nutrient fluxes 74-75 according to a, c, e and g) treatment and b, d, f and h) day. Samples are reported as mean  $\pm$  SE, n=5. Red circles - anoxic treatments, yellow circles - four days without oxygen and reaerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments. Black triangles - 1 cm depth, green triangles -3 cm depth, orange triangles -5 cm depth, blue triangles -7 cm depth, pink triangles -9 cm depth.
- 3.10 Porewater a) nitrate, b) nitrite, c) ammonium and d) phosphate concentrations taken at the start and on day 7 of laboratory incubations from five different treatments, oxic, hypoxic, 1DA, 4DA and anoxic. Values are reported as mean  $\pm$  SE, n=5. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.
- 3.11 Relationship of a) nitrate and porewater nitrate, b) nitrite and porewater nitrite, c) ammonium and porewater ammonium and d) phosphate and porewater phosphate according to treatments. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.
- 3.12 Porewater diffusive fluxes of a) nitrate, c) nitrite, e) ammonium 81-82 and g) phosphate taken at the start and on day 7 of laboratory

77

73

79

incubations from 5 different treatments, oxic, hypoxic, 1DA, 4DA and anoxic, calculated from porewater concentration gradients. b, d, f and h) Box whisker plot; central line in each box indicates median, ends of box  $25^{\text{th}}$  and  $75^{\text{th}}$  percentiles, error bars and points. Values are reported as mean  $\pm$  SE, n=5.

- 3.13 Vertical profiles of potential enzyme activities of a) alkaline phosphatase, b) cellobiohydrolase, c) beta-glucosidase, d) chitinase and e) urease activities in the sediments exposed to different oxygen concentrations taken at the start and on day 7 of laboratory incubations, reported as mean  $\pm$ SE, n=5. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.
- 3.14 Scatterplots of AP and porewater phosphate concentrations according to different a) treatments and b) depths, urease and porewater ammonium concentrations according to different c) treatments and d) depths, and urease with TOC according to different e) treatments and f) depths, reported as mean ±SE, n=5. Red circles anoxic treatments, yellow circles four days without oxygen and re-aerated for 3 days (4DA), blue circles initial treatments, grey circles hypoxic conditions, purple circles 1 day without oxygen and re-aerated for 6 days (1DA), green circles oxic treatments. Black triangles 1 cm depth, green triangles 3 cm depth, orange triangles 5 cm depth, blue triangles 7 cm depth, pink triangles 9 cm depth.
- 3.15 Scatterplots of the a) principal component analysis of porosity, total organic carbon, total nitrogen, porewater ammonium concentrations. alkaline phosphatase activities. cellobiohydrolase activities. beta-glucosidase activities. chitinase activities and urease activities identified by a) 5 different treatments and b) sediment depths measured. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments. Black triangles - 1 cm depth, green triangles -3 cm depth, orange triangles -5 cm depth, blue triangles -7 cm depth, pink triangles -9 cm depth.
- 3.16 Relationship between chitinase and AP according to a) 9 treatments and b) depth, between chitinase and BG according to c) treatments and d) depth, BG and urease according to e) treatments and f) depth, CB and TOC according to g) treatments and h) depth, chitinase and porewater ammonium concentrations according to i) treatments and j) depth. Red circles - anoxic

89

85

86

90-91

treatments, yellow circles - four days without oxygen and reaerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments. Black triangles - 1 cm depth, green triangles - 3 cm depth, orange triangles - 5 cm depth, blue triangles - 7 cm depth, pink triangles - 9 cm depth.

- 4.1 Framework of biogeochemical processes in the sediments under 107-109 a) oxic, b) hypoxic, c) 1DA, d) 4DA and e) anoxic treatments. The schematic highlights the different response of nutrients and enzyme activities in each treatment. The response is also supported by bromide fluxes, oxygen penetration depth and fluxes and total organic carbon and total nitrogen in each treatment. OW- overlying water,  $NO_3^-$  - nitrate,  $NO_2^-$  - nitrite,  $NH_4^+$  - ammonium,  $PO_4^{3-}$  - phosphate, OPD- oxygen penetration depth, Deeper sed- deeper sediment, Fe (III)-P - iron (III)phosphorus, Fe(II)-P - iron (II)-phosphorus, APA- alkaline phosphatase activity OMorganic matter. CBcellobiohydrolase, BG- beta-glucosidase, → clear relationship ,unclear relationship rightarrow.
- 4.2 Factors affecting enzyme activities in the sediment of this study. C- carbon, PW  $NH_4^+$  - porewater ammonium, BG- βglucosidase, CB- cellobiosidase, TOC- total organic carbon, TN- total nitrogen, AP- alkaline phosphatase, PW  $PO_4^{3-}$  porewater phosphate.

118

#### ACKNOWLEDGEMENTS

First and foremost, I would like to express my utmost gratitude and appreciation to my supervisor, Prof Alastair Grant for his constant support, patience, invaluable guidance and time in finishing this study. Without his help, I would not have reached this level. My thanks are also to my co-supervisor, Prof Julian Andrews who has been understanding and supportive throughout my study.

I would also like to express my gratitude to Emily Sear, Judith Mayne, and Rob Utting for their help in setting up my experiment. Many thanks are also conveyed to Kim Goodey, Emma Hooper, Graham Chilvers and Stephen Humphrey, for their expertise in using analytical instruments. A special thanks also to Alan Jones for his guidance and technical expertise in the world of metabolomics. Not to forget, Jenny Stevenson and Simon Ellis for their assistance during field sampling at Breydon Water, Great Yarmouth. My eternal gratitude is dedicated to Ministry of Higher Education Malaysia and Universiti Sains Malaysia for the financial support via the 'Rancangan Latihan Kakitangan Akademik' program.

My studies would not be as colourful and enjoyable without the support from my friends, especially Sharifah Norkhadijah, Asmida Ismail, Yasmin Yaseem, Maheran Mohamed, Noor Armylisa's Abu Hasan, Ahmad Tarmizi Che Has, Sandhya Sukumaran and Alina Mihailova. Most of all, I can't thank enough my family, especially my parents and parent in-law for encouraging me through thick and thin of this valuable journey. Finally, this study would not have been completed without the constant support, tolerance, motivation and understanding from my beloved husband, Mohd Harizal Senik and for taking care of our beloved daughter, Wardah Imtithaal.

#### **CHAPTER 1 INTRODUCTION**

# 1.1 Estuaries - Their Ecological Importance and Exposure to Environmental Pressures

Estuaries are a semi-enclosed coastal body of water which receive both seawater and freshwater mixing and are subject to daily tidal action (Church, 1975; Hobbie, 2000). They are among the most productive habitats in the world and serve as a receptacle for chemical flows originating from land. They account for high levels of nutrient present in the water column and sediments from seawater and freshwater inflows (Charles, 2004). Recognising their ability to store, bury and recycle chemicals and nutrients from the flow of overlying water to the sediments; estuaries play a vital role in regulating the biogeochemical cycle of nutrients and elements, climate change and organic matter mineralisation (Libes, 2009).

Accelerating agricultural activities, industrial development and population growth near estuaries have caused over enrichment of nutrients and organic matter accumulation (Nedwell et al., 2002; Doney, 2010). These problems are drastically enhanced by human population growth in coastal areas and near estuaries (Meyer-Reil and Koster, 2000). Human activities have resulted in increasing nutrient flows into the water and sediments, dumping or discharging of industrial wastes and municipal sewage sludges and altering the chemical composition of atmosphere through global warming (Libes, 2009). Excessive nutrient inputs, primarily nitrogen and phosphorus further increase the primary production by algae and macrophytes, resulting in eutrophication (Cai et al., 2011; Howarth et al., 2011). Subsequent decay of high plant biomass causes an increase in oxygen consumption which may lead to progressive oxygen depletion within the ecosystem (Meyer-Reil and Koster, 2000; Nedwell et al., 2002). Therefore, the impact of low oxygen conditions (hypoxia) and zero oxygen conditions known as anoxia (Gray et al., 2002; Rabalais et al., 2002; Diaz and Rosenberg, 2008) on sediment biogeochemistry is selected in this study as large number of estuaries is susceptible to this condition.

#### 1.2 Hypoxia and Anoxia

In seawater, anoxia is defined as zero oxygen concentrations (Riedel et al., 2008; Rabalais, 2010) whereas hypoxia is defined as dissolved oxygen concentrations below ~25% of water oxygen saturation or  $\leq 2$ ml of O<sub>2</sub>/l and ~2.8 mg O<sub>2</sub>/l (Gray et al., 2002; Wu, 2002; Larson and Sundback, 2008; Vaquer-Sunyer and Duarte, 2008). Hypoxia is influenced by natural phenomenon of the system itself, by human influences or as a result of both natural and anthropogenically-induced processes (Middelburg and Levin, 2009). Naturally occurring hypoxia is associated with water column stratification, which restricts water exchange in natural intrusions or upwelling of shelf systems, in silled basins and fjords such as the Black Sea (Wu, 2002; Diaz and Rosenberg, 2008). Human influences on coastal hypoxia include growing agricultural and industrial development which exerts increase amount of nutrients in water bodies, accumulation of pollutants and modification of the natural status of estuarine ecosystem (Gray et al., 2002; Middelburg and Levin, 2009).

Hypoxia starts with accumulation of organic matter in the system by production, respiration and decomposition processes. The deposition of organic matter enhances microbial growth, which exerts greater demand for oxygen through respiration. If stratification of the water column occurs, dissolved oxygen concentration will be further depleted. Then, the second phase is initiated where hypoxia occurs for a brief period or short-lived, which causes mortality of benthic aerobic organisms (Vaquer-Sunyer and Duarte, 2008). As time goes on, increasing amounts of nutrients and organic matter in the sediments result in the progression of the third phase of hypoxia, seasonal or periodic hypoxia. This phase particularly affects the animal population, with altering behaviour, physiology and mortality of organisms (Steckbauer et al., 2011). If nutrients and organic matter continue to accumulate and hypoxia persists, the dissolved oxygen concentration will further deteriorate until anoxia happens. During anoxia, microbially generated hydrogen sulphide is released (Diaz and Rosenberg, 2008). As a consequence of hypoxia, the number of fishes reduced and food chain are being disrupted (Karlson et al., 2007). There has recently been increased concern about the oxygen status of the North Sea due to evidence of potentially hypoxic regions and decreasing oxygen levels (Greenwood et al., 2010; Jonasson et al., 2012). The causes of these low oxygen levels are related to isolation of a shallow bottom mixed layer and the remineralisation of the organic matter produced during the spring bloom. In the Humber Estuary, UK, large scale oxygen depletion has been recorded historically in particular at Trent Falls, the upper reaches of the estuary and River Ouse (Uncles et al., 1998). The oxygen depletion is attributed to direct discharges of sewage and industrial waste to the estuary. However, in recent years this event has become irregular, reducing in temporal frequency as well as spatial extent (Cutts et al., 2008). Parts of the Mersey estuary may be particularly vulnerable to oxygen depletion, as changes in water quality is highest in semi-enclosed water bodies with long retention times, and where stratification of the water column occurs (Langston et al., 2006).

Based on the duration and dominant time scales of low oxygen, four types of hypoxia can be identified in the ecosystem, which are 1) permanent, 2) seasonal, 3) periodic and 4) episodic hypoxia. The first type, permanent hypoxia occurs in shelf regions, large fjords and inland seas where strong stratification leads to bottom water hypoxia/anoxia in this region (Gilbert et al., 2010). Table 1.1 showed some of the ecosystems experiencing hypoxia distributed all over the world based on the different types of hypoxia.

The second type, seasonal hypoxia is the most commonly occurring hypoxia, responsible for about half of the known dead zones. Seasonal hypoxia usually occurs once in a year, generally during summer after spring blooms. In summer, water stratification is the strongest and water is the warmest which enhance the declining of oxygen concentration in the water bodies (Diaz and Rosenberg, 2008). Seasonal hypoxia may occur in both stratified temperate estuaries and shelf regions as well as throughout the water column of shallow well-mixed estuaries and tidal rivers, subjected to summer heat strengthen stratification (Kemp et al., 2009). This is the most well studied type of hypoxia in the reported literature (Diaz and Rosenberg, 2008; Selman et al., 2008; Kemp et al., 2009; Turner et al., 2012).

Table 1.1 Global distribution of hypoxia in estuaries, coastal and marine ecosystems based on the type of hypoxia (adapted from Diaz and Rosenberg, 2008 and Rabalais et al., 2010).

System	Year	Hypoxia Type	Impacts	References
Pacific Oceans	2002	Permanent	Widespread foraminiferans	Helly and Levin,
			and meiofaunal nematodes	2004
North Indian Oceans	1998	Permanent	Fish death and plankton	Navqi et al., 2010
			blooms	
Donegal Bay, Ireland	2000	Episodic	Harmful algal blooms	Silke et al., 2005
Tokyo Bay, Japan	1970	Episodic	Decline of Mantis shrimp	Kodama et al., 2006
North Carolina	2004	Episodic/Diel	Fish abundance decreased,	Stevens et al.,
estuaries			replaced with resilience	2006
			estuarine and freshwater	
			species	
Harvey Estuary,	1980	Periodic	Algal bloom	Hearn and
Australia	1000			Robson, 2001
Westerchelde Estuary,	1990	Periodic	Affects	Nobre et al., 2005
Netnerlands	1050 2001	Casaanal	nitrification/denitrification	Us are at al. 2004
Chesapeake Bay, USA	1950-2001	Seasonal	Mortality with annual	Hagy et al., 2004
Northern Gulf of	1970	Saaconal	Blooms of dinoflagellates	Rabalais et al
Mexico	1970	Seasonai	Bioonis of unionagenates	
Dokai and Ise Bays	1990	Seasonal	Reduce shellfish	Ueda et al. 2002
Japan	1770	beabonai	populations	0000 et ul., 2000
Mersey Estuary,	1850	Seasonal	Fish kills	Jones, 2006
England				,
Thames Estuary,	1815	Seasonal	Mortality of fish and	Shand, 2005
England			benthic macroinvertebrates	
Loire Estuary, France	1990	Seasonal	Mortality of migratory	Abril et al., 2003
			species and mullet, Liza	
	1007 7000	~ .	ramada	
Waquoit Bay, USA	1995-2000	Seasonal	Overgrowth of macroalgae	Wenner et al.,
California Comment	2000	Cassanal	Abarras of all fish from	2004 Char et al. 2008
Larga Marina	2000	Seasonal	Absence of all fish from	Chan et al., 2008
			locky leels	
Corpus Christi Bay	1980	Seasonal	Reduced benthic	Montagna and
USA	1700	bousonal	organisms	Froeschke, 2009
Scheldt Estuary.	1990	Seasonal	Fluctuations of fish	Maes et al., 2007
Belgium				·····, ···,
Pearl River Estuary,	2000	Seasonal	High partial pressure of	Dai et al., 2006
China			carbon dioxide and high	
			ammonium concentrations	
Caspian Sea,	1990	Seasonal	Mortality with reduced	Dumont, 1998
Turkmenistan			benthos	
St Lawrence Estuary,	1980	Seasonal	Avoidance of fisheries	Benoit et al., 2006
Canada				

Periodic hypoxia is less severe than seasonal hypoxia, which tends to occur more frequent than seasonal hypoxia, lasting from days to weeks only. This type of hypoxia generally occurs in productive, shallow, slightly deeper (3-8 m) weakly stratified micro tidal coastal systems (Park et al., 2007; Kemp et al., 2009). Relatively, this type of hypoxia happens because of local weather events such as storm and turbulence. It is also influenced by wind and spring neap-tidal cycles, which may increase water stratification for a certain period of time. Daily diel cycles may also caused periodic hypoxia since this processes affects production and respiration, which generally lasts only for a few hours, but has a daily reoccurrence (Kemp et al., 2009).

Episodic hypoxia or diel hypoxia is the least frequent type to occur compared to seasonal and periodic hypoxia, with less than one event per year and it may also happen with years elapsing between events. This type of hypoxia tend to appear and disappear on short (hours to days) time scales in very shallow tidal estuaries (1-5m) (D'Avanzo and Kremer, 1994; Tyler et al., 2009). Normally, this type of hypoxia is triggered by excessive night time respiration of organic matter exceeding oxygen replenishment by air-water exchange during the day by photosynthetic oxygen production (Kemp et al., 2009). When a system has reached this stage, it is an indicator that eutrophication problem within the area is reaching a critical point along with physical factors that stratify the water column, which results in the progression of hypoxia. Episodic hypoxia can be exacerbated by major storm events that deliver large pulses of organic loading. Effect of storm events in estuaries caused a rapid decrease in dissolved oxygen concentrations, which might play a crucial importance in determining the initial response and resilience of the system towards prolonged negative consequences associated with oxygen depletion (Diaz and Rosenberg, 2008).

Large numbers of cases of hypoxia and anoxia have been documented, but some of the systems are already in the state of recovery as a result of measures that have been taken to reduce nutrient concentrations (Rabalais et al., 2010). The complexity of eutrophication processes associated with other environmental pressures such as global warming and ocean acidification (Vaquer-Sunyer and Duarte, 2008; Howarth et al., 2011), may influence the rate of recovery (Mee, 2001). The intensity, duration and frequency of hypoxia and anoxia are the key factors determining the response on coastal waters and marine ecosystems.

At present, no studies in the literature have looked at the effect of short term anoxia and re-aeration on nutrient fluxes associated with enzyme activities in estuarine systems. Therefore, this study focused on the effect of short term hypoxia and anoxia on sediment biogeochemistry, particularly nitrogen and phosphorus cycle and enzyme activities related to carbon, nitrogen and phosphorus cycle. Additionally, the information about solute transport, oxygen fluxes and total organic carbon and total nitrogen within the system may help in understanding the changes in nutrient fluxes, within the sediment-water interface and enzyme activities in the sediment.

#### **1.3** Solute Transport and Oxygen Fluxes

The study of solute transport has received attention because it shows how water is distributed within the system and provides clues of how harmful substances are spread within the system (Ohrstrom et al., 2004). Transport rates of chemicals are often used in biogeochemical studies of sediments to produce estimates of sediment-water fluxes (Berg et al., 2001). Solute transport in sediment affects the distributions of reactants and metabolites, the location and magnitude of early diagenesis reactions and the microbial processes (Meile and Cappellen, 2003; Dornhoffer et al., 2012). Solute exchange is driven by tides and currents which give way to molecular diffusion through porous medium of the sediment (Meile and Cappellen, 2003). In the uppermost layers of marine sediment, however, biologically induced solute transport (bioirrigation) can exceed transport due to molecular diffusion (Kristensen and Hansen, 1999; Meile et al., 2001). An effective means for describing and quantifying solute exchange through various biological and physical processes between the sediment water interfaces is by using tracers (Bradshaw et al., 2006).

Some ions which are used as tracers in seawater are present as conservative ions and some exist as non-conservative ion. When ions are present in nearly constant proportions in the system, these ions are termed as conservative since their concentrations are mostly controlled by physical processes associated with water movement, such as transport by currents, mixing via turbulence, evaporation and rainfall. If the ions are not present in constant proportions and are altered by chemical reactions that occur faster than the physical processes responsible for water movement, these ions are termed as non-conservative (Libes, 2009). Bromide is a conservative tracer that does not undergo sorption or chemical and biological transformations (Keefe et al., 2004). So, it can be used as a non-reactive tracer of the mass flow of water through the sediments (Kazemi et al., 1998). Bromide is commonly spiked into the water as a solute transport tracer in sediment (Forster et al., 1995; 1999; Quintana et al., 2011). The use of bromide as a tracer avoids the regulatory constraints and complexities associated with the use of radioactive tracers (Sayles and Martin, 1995). In this study, bromide is used to assess the effect of oxygen depletion and re-aeration on the water movement of fine-grained sediment by measuring solute transport and related fluxes between the sediment and the water.

The effect of changing the oxygen concentration in the water overlying finegrained sediment was assessed on the flux of oxygen across the sediment-water interface and on the distribution of oxygen within the sediments. This flux provides information on the oxygen penetration depth in fine-grained sediment and how oxygen is consumed. In estuaries, oxygen concentration in the overlying seawater may vary due to water stratification, tidal effects, storm occurrence or hydrodynamic effects (Lewis et al., 2007). Oxygen penetration depth (OPD) was determined from oxygen profiles as the depth below sediment water interface where oxygen permeating from the water column is exhausted (Katsev et al., 2007). The consumption of oxygen by coastal sediments is responsible for extensive hypoxia in estuaries and eutrophicated coastal zones (Zimmerman and Canuel, 2000; Reimers et al., 2012). Oxygen distribution results from a complex interplay between physical processes (diffusion, sedimentation), chemical reactions (oxidation of reduced species) and biogeochemical processes such as photosynthesis or respiration (Rowe et al., 2002; Lansard et al., 2003; Kim and Kim, 2007; Glud, 2008; Glud et al., 2010).

#### **1.4 Organic Matter**

Organic matter is derived from bacteria, phytoplankton and zooplankton in the water column, bacteria and diatom from the upper surface of the sediments and through the production by algae (Cowie and Hedges, 1994; Fernandes et al., 2007). Natural inputs of organic matter originate from land-derived nutrients which are washed off into the coastal ecosystems. External sources of organic matter inputs by humans are largely through increase human population and settlement along coastal and estuarine areas (Kristensen, 2000; Meyers et al., 2002). This leads to increase sewage input to the coastal areas (Mudge and Duce, 2005).

Biogeochemical processes in sediments are influenced mainly by conditions in the water column. Easily degradable organic matter within the water column and the composition of bottom waters are the major factor fuelling the sediment biogeochemistry (Soetart et al., 2000). In the water column, eddy diffusion and particle settling are the transport processes. Organic matter produced from primary production within the water column is deposited to heterotrophic consumers inhabiting sediment as means for getting nutrients and energy. These organic matters are then processed by various microbes, protozoans and metazoans (Kristensen, 2000). Processes which occur in the sediments are different from the water column because of different mechanism of transport processes. In sediments, molecular diffusion and bioturbational mixing prevails. Most of biogeochemical processes involved in sediments are linked with decomposition of organic matter. Organic matter is synthesised in the euphotic zone, where primary production takes place since sunlight may penetrate the area and early diagenesis process can occur (Ingalls et al., 2004). Some of the organic matter which are too large to be grazed and sink to the seabed along with faeces and particulate organic matter are degraded in the water column (Hedges et al., 2001).

A minor portion is subjected to sediment burial, contributing to refractory organic matter (Middelburg and Levin, 2009; Burdige, 2011). Secondary production by heterotrophic organisms assimilates the organic matter by respiration or mineralisation. These two processes consume high amount of oxygen, thus contributing to hypoxic and anoxic conditions within the upper few centimetres of the sediment surface.

Redox reactions are a means of obtaining energy and essential elements for marine and estuarine organisms. This process is controlled by the spontaneous reduction of oxygen and oxidation of organic matter. When oxygen is low, microbes engage in metabolic strategies that involve reduction of oxidised solutes, such as nitrate, Mn (IV), Fe (III), sulphate and organic matter. Therefore, how organic matter is produced and processed based on the most energetically favourable redox reaction is important to understand the different processes occurring in the sediments.

Due to high demand of oxygen in the upper sediments, organic matter degradation in sediments proceeds using the most available oxidant producing the greatest free energy (Figure 1.1). Organic matter degradation follows a certain sequence of reactions, with each successive reaction starting when the previous oxidant is depleted or when chemical and biological conditions favour certain type of organisms to flourish (Bianchi, 2007; Libes, 2009). The rate of decay depends on a few factors, mainly the quality of organic matter, whether labile or refractory, the major composition of organic matter such as protein, cellulose and lignin, at which decomposition stage it is going through and temperature effect (Sullivan et al., 2010). Generally, the formula for chemical composition of organic matter in estuarine sediments, are:

#### $(CH_2O)_x(NH_3)_y(H_3PO_4)_z$ ,

where x, y and z varies according to the origin and age of the material. The composition for freshly produced marine organic matter is usually close to the Redfield ratio: x = 106, y = 16 and z = 1 (Kristensen, 2000; Schneider et al., 2003).

Oxygen limitations in the upper sediments initiate a cascade of alternative electron acceptor use by anaerobic organisms to continue the decomposition process into anoxic layers. In the oxic (oxygen containing) zone, aerobic respiration, nitrification and sulphide oxidation takes place. Anaerobic decomposition involves different types of bacteria for the processes of manganese oxide reduction, denitrification, iron oxide reduction, fermentation, sulphate reduction and carbon dioxide reduction (Figure 1.1). Although the strict vertical distribution of electron acceptors shown in Figure 1.1 is an oversimplification of the true spatial distribution; generally this is the pattern of aerobic and anaerobic decomposition process within zones. The decomposition process starts off with hydrolysis and fermentation of complex polymeric organic molecules into water soluble monomers such as amino acids, monosaccharides and fatty acids. During the process of hydrolysis and fermentation, energy and inorganic nutrients are being released (Libes, 2009). The stoichiometries for denitrification and sulphate reduction are as follows:

Denitrification (**Equation 1.1**):

CH<sub>3</sub>COOH + 1.6 NO<sub>3</sub><sup>-</sup> + 1.6 H<sup>+</sup> → 2CO<sub>2</sub>+ 0.7N<sub>2</sub> + 2.8 H<sub>2</sub>O

Sulfate reduction (Equation 1.2):



Figure 1.1 Vertical processes of diagenetic reactions in sediments. Source: from Libes, (2009).

Methane is the end product when all oxidants have been consumed during organic matter degradation. During anaerobic respiration, various reduced substances are produced such as ammonium, iron (II), manganese (II), hydrogen sulphide and methane. These reduced substances will eventually be re-oxidised since they contain enough energy from the organic matter and mediated by microorganisms (Kristensen, 2000). Ammonium produced from organic nitrogen mineralisation can be oxidised either aerobically or anaerobically through the process of anammox (Jetten, 2008). Reduced iron and manganese in dissolved form is released by dissimilatory reduction of particulate metal oxides. Sulfide produced by sulphate reduction is also reoxidised due to reactions with iron and other metals (Leloup et al., 2007). Since sulphate is the second most abundant anion in seawater, majority of anaerobic mineralisation occurs through sulphate reduction. During sulphate reduction, sulphide reacts with iron to produce iron sulphide whereas another part reacts with organic matter or is involve in oxidative sulphur cycle (Middelburg and Levin, 2009). The end product from this process depends on bioturbation of reduced particulate sulphur and non-steady state diagenetic processes stimulating contact between reduced and oxidised compounds. Reduced sulphur is re-oxidised by oxygen, nitrate, metal oxides or more oxidised sulphur compounds (Riedinger et al., 2010). Majority of methane is oxidised anaerobically involving bacteria and archaea.

Due to the high demand of oxygen within coastal sediments and estuaries, anaerobic mineralisation usually dominates because of high organic carbon loading (Middelburg and Levin, 2009). The majority of oxygen consumption is attributable to reoxidation processes and less than 25% is directly attributable to aerobic respiration processes.

#### **1.5** Nutrients in Estuaries

Since hypoxia and anoxia are largely driven by excessive nitrogen and phosphorus inputs, the following focuses on the role of estuaries in governing nitrogen and phosphorus cycling and the changes caused by oxygen.

#### 1.5.1 Nitrogen Cycling

Nitrogen is one of the most important elements required by biological systems and in estuaries; its availability is frequently implied as one of the factors limiting productivity (Jjemba, 2004; Ward et al., 2007). Given its major role within these ecosystems, there is a need to establish the forms in which nitrogen is present, the routes by which they are transformed, the microorganisms concerned with these processes and factors associated with its presence (Herbert, 1999). Nitrogen is present in many oxidation states with commonly found species in the form of nitrate ion  $(NO_3)$ , nitrogen dioxide gas  $(NO_2)$ , nitrite ion  $(NO_2)$ , nitrous oxide gas  $(N_2O)$ , ammonia gas (NH<sub>3</sub>), and ammonium ion (NH<sub>4</sub><sup>+</sup>) (Thamdrup and Dalsgaard, 2008; Libes, 2009). Nitrogen cycling is largely controlled by redox transformations between the states which are mediated by organism such as phytoplankton, bacteria and probably archaea. With the exception of dissolved nitrogen gas, the principal forms of nitrogen occurring in marine and estuarine ecosystems are nitrate, nitrite, ammonium and organic nitrogen. Sources of nitrogen include precipitation, agricultural and urban run-off, sewage discharges, sludge dumping and biological nitrogen fixation (Herbert, 1999).

The first step in initiating nitrogen cycle is nitrogen fixation (Jjemba, 2004). Most of nitrogen in estuaries is present as nitrogen gas which is biologically inaccessible except to those few microbes, called nitrogen fixers that can break nitrogen's strong triple bond. Nitrogen fixers can be photoautotrophs, photoheterotrophs, or chemoheterotrophs showing that nitrogen fixation may occur both in light and dark. The following sections discuss various processes involved in nitrogen cycling, starting with nitrogen fixation, followed by ammonium assimilation through solubilisation process, ammonification, ammonium oxidation, nitrate reduction, denitrification and dissimilatory nitrate reduction to ammonium.

#### **1.5.2** Nitrogen Fixation

Nitrogen fixation is an enzyme-mediated reduction of nitrogen gas  $(N_2)$  to ammonium, a major source of fixed nitrogen to the biosphere and is particularly

important in nitrogen limited environments (Ward et al., 2007). The stoichiometry of nitrogen fixation requires 8 electrons and 16 molecules of ATP to reduce one molecule of  $N_2$ , as shown below:

#### **Equation 1.3:**

 $N_2 + 8H^+ + 8e^- + 16MgATP$  nitrogenase  $2NH_3 + H_2 + 16MgADP + 16P_i$ 

Nitrogen fixation is energetically expensive, and only prokaryotes can fix nitrogen such as *Trichodesmium* and the cyanobacteria, *Crocosphaera* (Libes, 2009). The reduced nitrogen or ammonium is then incorporated into organic matter and released again when organic nitrogen compounds are degraded (Thamdrup and Dalsgaard, 2008). The enzyme required to facilitate nitrogen fixation is nitrogenase. Iron is an essential cofactor and is required in large amounts, making iron availability a major factor limiting nitrogen fixation. Nitrogenase is inactivated by oxygen, so, aerobic prokaryotes (*Cyanobacteria, Azotobacter, Rhizobium*) form thick cell walls that prohibit oxygen exchange, creating an anoxic intracellular environment (Libes, 2009). Some free-living cyanobacteria protect their nitrogenase against oxygen exposure by conducting nitrogen fixation only at night when photosynthesis is not occurring. Recent investigation has revealed that the rate of nitrogen fixation is underestimated since no estimates have yet been made in hydrothermal systems, anoxic zones in estuarine waters and anoxic microzones within sinking detrital particulate organic matter (Libes, 2009).

#### 1.5.3 Ammonification and Ammonium Assimilation

In the process of nitrogen fixation,  $N_2$  gas is reduced to ammonium which is assimilated into organic nitrogen through anabolic activities of the autotrophs and their consumers (Thamdrup and Dalsgaard, 2008). During the process of ammonium assimilation, ammonium is transported across the cell membrane, mediated by permeases enzyme. Once ammonium has crossed the cell membrane, it can participate in anabolic reactions. For example, ammonium is incorporated into amino acids by first reacting with  $\alpha$ -ketoglutaric acid, followed by transamination reactions (Jjemba, 2004). Other nitrogenous biomolecules are similarly created by these reactions, producing nucleotides, amino sugars, photosynthetic pigments, vitamins and other metabolites. If ammonium concentrations in seawater are low, nitrate and nitrite will be assimilated through the process of assimilatory nitrogen (nitrate or nitrite) reduction. This process is also mediated by permeases and once inside the cell, these dissolved inorganic nitrogen (DIN) species are transformed into ammonium via redox reactions (Libes, 2009). But, phototrophs prefer to assimilate ammonium compared to nitrate and nitrite as less redox energy is required. Some phototrophs may also assimilate dissolved organic nitrogen (DON), such as urea to meet their nitrogen needs (Bianchi et al., 2004).

The process of solubilisation comes into play when particulate organic nitrogen (PON) is decomposed into DON by fragmentation of biomolecules into compounds small enough to be classified as dissolved (Madsen, 2008). The organic nitrogen can be decomposed by heterotrophic bacteria back into ammonium (Jjemba, 2004). The process of cell lysis, hydrolysis of protein, and metabolism by heterotrophs is a three step known as ammonification (Figure 1.2).



Figure 1.2 Nitrogen cycle through various redox and phase transformations. The boxes contain the nitrogen species and its oxidation number. The arrows represent transformation reactions as follows: 1) nitrogen fixation, 2) solubilisation, 3) ammonification, 4) nitrification, 5) denitrification, 6) anammox, 7) anaerobic nitrification mediated by manganese reduction, 8) DNRA, 9) assimilatory nitrogen reduction. A=anammox microbes (Source: from Libes, 2009).

#### 1.5.4 Ammonium Oxidation

Two processes are involved in ammonium oxidation which are nitrification and anaerobic ammonium oxidation or anammox (Libes, 2009). In oxic conditions, NH<sub>4</sub><sup>+</sup> is thermodynamically unstable, chemoautolithotrophic bacteria of the family *Nitrobacteraceae* mediate the stepwise oxidation of ammonium to nitrite and then to nitrate known as nitrification (Madsen, 2008). The first step is conducted by organisms such as *Nitrosomonas* and *Nitrocystis* (an autotrophic organism) by oxidising ammonium to nitrite in an aerobic environment and the second step by *Nitrobacter, Nitrospira*, and *Nitrococcus*, by oxidising nitrite to nitrate (Jjemba, 2004; Madsen, 2008; Thamdrup and Dalsgaard, 2008; Libes, 2009).

Anammox occurs under anaerobic conditions in the water column and sediments when ammonium is oxidised through two pathways (Libes, 2009). First, ammonium is oxidise using nitrite as the electron acceptor and contributes as a dissimilatory nitrogen reduction type, by which DIN is converted to N<sub>2</sub> (Shivaraman and Shivaraman, 2003). This process is an alternative to denitrification in removing fixed nitrogen from the marine and estuaries ecosystem. Second type of anammox involves the oxidation of ammonium using manganese, Mn (IV) as the electron acceptor in anoxic sediments. This causes the manganese to be reduced to Mn<sup>2+</sup> and the nitrogen to be oxidised to N<sub>2</sub> (Libes, 2009).

#### 1.5.5 Nitrate Reduction

In the presence of oxygen, nitrate is the thermodynamically stable form of nitrogen, whereas in the anoxic part of the sediment, it is a reactant in dissimilatory redox reactions that lead to the formation of  $N_2$  or ammonium. The first step in this reductive pathway is the reduction of nitrate to nitrite, which is catalysed by nitrate reductases, which contain a molybdenum cofactor (Thamdrup and Dalsgaard, 2008). This is often treated as part of the denitrification pathway. A catalyse is required because of the high stability of nitrate ion in the environment. From nitrite, three pathways of reduction follows, which are denitrification and anammox, leading to  $N_2$  and dissimilatory nitrate reduction to ammonium (DNRA).

#### 1.5.6 Denitrification

Denitrification is the reduction of nitrate to  $N_2$  by heterotrophic bacteria. The stoichiometry is:

#### Equation 1.4:

#### $(CH_2O)_{106}(NH_3)_{16}H_3PO_4+84.8 HNO_3 \implies 106 CO_2 + 148.8 H_2O + 42.4 N_2 + 16 NH_3 + H_3PO_4$

Denitrification proceeds through a series of steps with nitrate reduced to nitrite (nitrate reduction process) followed by subsequent reduction of nitrite to  $N_2$ . Under some conditions,  $N_2O$  is also produced. Some of the nitrogen that is denitrified is derived from organic matter undergoing oxidation. Two fates of this process occur when first, organic nitrogen present in the reactant organic matter is oxidised to  $N_2$  and another organic nitrogen in the reactant organic matter is ammonified. The relation between anammox and denitrification occurs when some of the ammonium produced during denitrification is used by anammox bacteria and converted to  $N_2$  (Libes, 2009). Two equations to describe the stoichiometry of denitrification are: Equation 1.4 and equation 1.5 below:

#### **Equation 1.5:**

 $(CH_2O)_{106} (NH_3)_{16} H_3PO_4 + 94.4 HNO_3 \rightarrow 106 CO_2 + 177H_2O + 55.2 N_2 + H_3PO_4$ 

#### **1.5.7** Dissimilatory Nitrate Reduction To Ammonium (DNRA)

Under anaerobic conditions, heterotrophic bacteria will use organic matter as an electron donor to drive the reduction of nitrate to ammonium. This process is known as dissimilatory nitrate reduction to ammonium (DNRA). It is coupled to nitrate reduction or nitrite ammonification (Thamdrup and Dalsgaard, 2008). This process is important as a sink of nitrate in coastal and estuarine sediments. Some sedimentary bacteria perform DRNA by using sulphide as an electron donor, for example in giant sulphur bacteria such as *Thioploca* sp. and *Beggiatoa* sp. (Hulth et al., 2005).

As a result of DIN transformations in response to different oxygen levels, we would expect nitrate to be dominant in the surface layers of the sediment in aerobic conditions, whereas in anoxic conditions, ammonium is expected to prevail. Reaeration effect of 1DA (1 day without oxygen followed by re-aeration of 6 days) and 4DA (4 days without oxygen and subsequent re-aeration of 3 days) would cause negative effect to nitrogen cycling, resembling anoxic conditions.

#### 1.5.8 Phosphorus Cycling

Phosphorus (P) availability is crucial in estuaries since it is known as a limiting nutrient, which in turn gives an impact to the primary production and sequestration of atmospheric carbon dioxide in organic matter (Paytan and McLaughlin, 2007). Unlike nitrogen, P cannot be fixed from the atmosphere, and only a minor portion is contributed by dust deposition (Libes, 2009). P occurs naturally in the form of inorganic phosphate minerals and organic phosphate derivatives in rock and soil. In estuaries, organic P only constitutes a small portion of the bulk weight, and is much less abundant than inorganic P (Paytan and McLaughlin, 2007).

The main source of P in estuaries is through river runoff and the sink is through deposition and burial in the sediments (Libes, 2009). P is transported to coastal marine and estuaries in the form of particulate and dissolved phases via riverine flux. Anthropogenic sources of P includes from fertiliser, sewage, soil erosion, livestock and paper pulp manufacturing (Paytan and McLaughlin, 2007).

P burial in the sediments depends on the redox conditions of the sediment. Oxic surface sediments are usually rich in manganese and ferric iron phases which take up large amount of phosphate by adsorption and mineral formation. The opposite is for anoxic sediments, which are depleted of these phases and P is more bound to calcium minerals (Rozan et al., 2002).

In sediments, initial burial of P occurs with organic carbon or P associated with iron oxyhydroxides (Steenbergh et al., 2011). Once in the sediment, labile forms of P are transformed to authigenic forms in a variety of redox states. In oxidising sediments, the majority of the flux of organic phosphorus from the water column is mineralised within the sediments. The released phosphate is partitioned between the porewater and surface adsorption sites. Surface adsorbed phosphate is released into

the porewater to replace the dissolved phosphate that escapes the sediment/water interface and into the water column. If conditions become reducing, phosphate is released to porewaters from the reduction of iron oxides (Belias et al., 2007). Sediments from anoxic conditions desorb or release phosphate from the sediments exceeding the adsorption capacity, whereas in oxic conditions, desorbed phosphate is rapidly readsorbed by iron oxyhydroxides or precipitated as apatite (Paytan and McLaughlin, 2007). In relation to the context of this study, we would postulate that phosphate is immobilised in oxic conditions as a result of Fe (III) binding to phosphate. In contrast, phosphate will be released in periods of anoxic condition due to Fe (III) reduction to Fe (II).

# 1.6 Enzyme Activities in Sediments as Indicators of Biogeochemical Processes

Enzyme activities come from various sources of biotic, abiotic compounds, cells, mineral complexes and the soil solution phase (Taylor et al., 2002). Estuarine benthic environments are characterised by rapid sediment accumulations (Alongi et al., 2005) and high supplies of organic material (Sobczak et al., 2002). Large quantities of riverine materials, organic loads, and autochthonous organic material produced in the estuary are deposited at the sediment surface. The organic matter remineralisation is initiated by microbial extracellular enzymes (Arnosti et al., 2009; Arnosti, 2011), which have been studied extensively in marine environments (Meyer-Reil, 1986; Mayer, 1989; Boetius et al., 2000; Danovaro et al., 2001; Arnosti and Holmer, 2003). The amount and composition of organic matter are cues for the synthesis of enzymes. Specific enzymes in the sediments are produced for the decomposition of specific compounds of organic matter (Meyer-Reil, 1986; Arnosti and Holmer, 2003). Substrate induction is an important mechanism controlling the activity of enzymes (Witte et al., 2003).

Vertical patterns of enzyme activity are variable and related to the specific characteristics of the benthic environment and the specific enzyme considered. Enzyme activities tend to co-vary with changes in substrate content and temperature (Meyer-Reil, 1986; Mayer, 1989; Boetius et al. 2000; Arnosti and Holmer, 2003;

Ziervogel and Arnosti, 2009). Changes in enzyme activity with depth may depend on the characteristics of the enzyme, the sediment organic matter or both (Poremba, 1995). However, enzyme activity can be quite constant with depth irrespective of substrate content (Poremba, 1995; Arnosti and Holmer, 2003). This study provides information on vertical distributions of five types of extracellular enzyme activities alkaline phosphatase (AP), cellobiohydrolase (CB),  $\beta$ -Glucosidases (BG), chitinase and urease activities in fine estuarine sediments to assess changes to the phosphorus, nitrogen cycle and organic carbon decomposition.

#### **1.6.1** Alkaline Phosphatase (AP)

Phosphatases have a broad range of organic substrate specificities, thus hydrolyse a wide variety of organic phosphomonoesters (Shi et al., 2011). Phosphatase enzymes are attached to cell surfaces or freely dissolved in the water column resulting from cell lyses or excretion (Hoppe, 2003). Upon enzyme hydrolysis, organic phosphomonoesters release inorganic phosphate into the water, thereby increasing inorganic phosphate availability for planktonic as well as benthic organism in shallow marine system (Zhang et al., 2007; Koch et al., 2009). The role of AP is important during periods when there are low inorganic phosphorus concentrations in the water column (Wright and Reddy, 2001). AP contributes as an essential component of microbial phosphorus activity, regenerating inorganic phosphate (Wright and Reddy, 2001). The relationship of AP and phosphate is shown below:

#### **Equation 1.6:**



AP has been identified and associated with all major groups of eukaryotic algae (Litchman and Nguyen, 2008), cyanobacteria (Koch et al., 2009), heterotrophic bacteria (Cotner et al., 2000) and submerged macrophytes, including seagrass (Martinez-Crego et al., 2006). Phosphatase activity is widely studied in freshwater lakes (Wright and Reddy, 2001; Yiyong et al., 2008; Chen et al., 2011; Jiang et al., 2011), coastal waters and deep-sea sediments (Paytan and McLaughlin, 2007; Caruso, 2010; Steenbergh et al., 2011), lagoon sediments (Sabil et al., 1995),

estuaries (Koch et al., 2009) and forest soil ecosystem (Caldwell, 2005). The interest in phosphatase activity stems from its potential role in evaluating phosphorus and carbon limitation within these ecosystems (Wojewodzic et al., 2011).

Although numerous investigations have been carried out on AP activities in lake sediment (Zhou et al., 2008; Deng et al., 2009) and coastal and deep sea sediment (Coolen and Overmann, 2000; Danovaro et al., 2001), only one study has assessed the effect of overlying water aeration on AP activity in surface sediment of a freshwater lake (Chen et al., 2011). Taylor et al., (2009) have assessed AP in suspended and sinking organic particles within the anoxic Cariaco Basin. The present study attempted to examine what factor influences AP in fine-grained sediment.

#### 1.6.2 Cellobiohydrolase (CB)

Cellobiohydrolase (CB) depolymerises cellulose to oligomers and monomers (Wittmann et al., 2004; Snajdr et al., 2008). CB activity catalyses cellulose into cellobiose according to the following relationship:

#### **Equation 1.7:**

Cellobiohydrolase Cellobiose Cellobiose

Cellulose is a homopolymer consisting of glucose units joined by  $\beta$ -1,4-bonds (Leschine, 1995; Sakamoto and Toyohara, 2009). Cellobiose is a disaccharide, consisting of two glucose molecules linked by a  $\beta(1\rightarrow 4)$  bond (Leschine, 1995). CB is produced by a wide range of micro-organisms, including fungi, actinomycetes, and bacteria (Rasmussen et al., 2002). The potential activity of this enzyme could reflect the input of cellulose into the sediments (Sakamoto and Toyohara, 2009; Maki et al., 2009).

The majority of the studies available in the literature focus on measuring CB in soil ecosystems. CB plays an important role in the carbon cycle in the rhizosphere of grasses (Sanaullah et al., 2011), in alpine and forest soils ecosystem (Grandy et al., 2007; Kelley et al., 2011). The effect of N deposition has also been assessed to CB activities in litter decomposition and agricultural soils (Saiya-Cork et al., 2002;

Sinsabaugh et al., 2005; Manning et al., 2008). Only one study has assessed CB activities in lake sediments (Hakulinen et al., 2005). The present study investigates CB activity in estuarine sediments.

#### **1.6.3** β-Glucosidase (BG)

 $\beta$ -Glucosidases (BG) are one group of enzymes that bacteria produce to cleave the  $\beta$ linkages of combined carbohydrates, belonging to 2 of the 82 families of enzymes grouped as glycosyl hydrolases (Bhatia et al., 2002; Bhaskar and Bhosle, 2008). The activity of BG is the rate at which the enzyme cleaves  $\beta$ -glycosidic bonds (Bhatia et al., 2002) and thus, can be measured using the rate at which glucose is hydrolysed from methylumbelliferyl molecules. BG catalyses cellobiose to form glucose (MacLellan, 2010) based on the relationship shown:

#### Equation 1.8:

Glucosidases are influenced by temperature and their rates regulated by substrate availability (Talbot et al., 1997). Various researchers have used glucosidase enzyme activity to relate the quality and sources of the organic matter to microbial activity in the near shore, estuarine waters and marine sediments (Koster et al., 1997; Fabiano and Danovaro, 1998; Azua et al., 2003). BG activities are measured because of their role as important energy sources for microorganisms and its involvement in biogeochemical transformation of carbon in decomposition process (Arnosti and Holmer, 2003).

A number of studies have also focused on BG activities because BG is involved in the degradation of cellulose, which is the most abundant organic compound on earth, providing the primary structural components for plants (Hakulinen et al., 2005). It is therefore not surprising that the profound interest in this enzyme has led to extensive studies in grassland soils (Turner et al., 2002), peataccumulating wetlands (Freeman et al., 1997), lowland rivers (Wilczek et al., 2005) and agricultural soils (Wang and Qin, 2006). The interest in shallow coastal waters, estuaries and deep sea sediments has developed later than enzyme studies in soils,
although the organic matter within these ecosystem also originates from autochthonous sources, algal exudates and detritus as well as allochthonous sources, e.g. decaying, dissolved and particulate material from terrestrial vascular plants (Wilczek et al., 2005).

# 1.6.4 Chitinase

Chitin is a homopolysaccharide, a linear polymer made of monomers of N-acetyl glucosamine connected by glucoside  $\beta$ -(1-4) bonds (Warren, 1996). It is the supporting substance of the outer skeleton of a number of animals, insects, arachnids and nematodes (Poulicek et al., 1998) and in fungal cell walls (Boetius, 1995). The main producer of chitin is crustacea and phytoplankton is the second most important source (Donderski and Trzebiatowska, 1999). Bacteria are considered to be the primary mediators of chitin degradation in aquatic environments (Gooday, 1990, Poulicek et al., 1998). The first enzymatic step in the chitin-degradation pathway is performed by chitinases, one of the glycosyl hydrolases families (LeCleir and Hollibaugh, 2006).

Chitin hydrolysis is carried out by heterotrophic bacteria assessed with the MUB-labelled N-acetyl glucosaminidase hydrolysis, an analogue for chitinase activity to break down chitin for acquisition of organic nitrogen (Souza et al., 2010). Chitinase is catalysed based on the following relationship (LeCleir and Hollibaugh, 2006; Beier and Bertilsson, 2011):

chitinase

## **Equation 1.9:**

Chitin ----- N-acetylglucosamine + diacetylchitobiose

The mineralisation of chitin is of great ecological significance as it enters both carbon and nitrogen cycles (Gooday, 1990; Kang et al., 2005; Kelley et al., 2011; Sanaullah et al., 2011). Figure 1.3 shows the relationship between chitin and the carbon and nitrogen cycles (Beier and Bertilsson, 2011). Chitin is hydrolysed prior to assimilation by microbial cells producing N-acetylglucosamine (GlcNAc). Chitin deacetylates into glucosamine or chitosan by the removal of acetyl groups from the molecular chain of chitin. Deamination removes the complete amine group and produces glucose. GlcNAc can be transported across the bacterial cell membranes where they undergo further enzymatic processing to acetate (LeCleir and Hollibaugh, 2006). These compounds may be incorporated into cell material or undergo mineralisation process producing ammonium and carbon dioxide. Ammonium enters the nitrogen cycle by converting into nitrogen through denitrification process. Nitrogen fixation produces ammonium, which can be assimilated into the bacteria. Carbon fixation converts carbon dioxide into organic matter, which can be assimilated by bacteria (Figure 1.3).

Chitinase activity has received far less attention than AP or BG activity although it plays a vital role in degrading chitin, an important polysaccharide to the marine food web (Olander and Vitousek, 2000; Sinsabaugh et al., 2005, Souza, et al. 2010). Most studies have focused on chitinase activity in planktonic, benthic and epiphytic bacteria (Donderski and Tzebiatowska, 1999) bacteria and fungi isolated from shrimp exoskeletons (Swiontek-Brzezinska et al., 2007), marine fish (Goodrich and Morita, 1977) and chitin catabolism in marine bacteria (Keyhani and Roseman, 1999). Chitinase activity has been reported in marine environments (Hoppe et al., 2002), pulp mill-contaminated lake sediment (Wittman et al., 2004), and sediments of the Ythan (Hillman et al., 1989) and Delaware (Kirchman and White, 1999) estuaries.



Figure 1.3 Chitin synthesis, assimilation and product in bacteria (adapted from Beier, 2010). Abbreviations: GlcNAc- N-acetylglucoamine, GlcN- glucosamine, Glc-glucose, NH<sub>4</sub><sup>+</sup>- ammonium, CO<sub>2</sub>-carbon dioxide, N<sub>2</sub>-nitrogen, C- carbon, N- nitrogen

#### 1.6.5 Urease

Urea is a nitrogenous waste product being excreted by many aquatic organisms, including zooplankton, molluscs, fish and mammals (Jorgensen, 2006). In addition, bacteria produce urea during degradation of purines and other nitrogen compounds such as arginine and allantoin (Satoh, 1980; Braga and Torres, 2010). Urease is synthesised by many eukaryotes including some invertebrates and numerous microorganisms (Therkildsen and Lomstein, 1994; Solomon et al., 2010). Urease activity has been found in a variety of different bacterial divisions comprising bacteria with aerobic and anaerobic metabolism (Peers et al., 2000; Solomon et al., 2010). Urease the hydrolysis of urea:

Equation 1.10:

 $CH_4N_2O + H_2O \xrightarrow{urease} CO_2 + 2NH_4$ 

The enzymatic, colorimetric, endpoint method for urea measurement is based on urease-catalysed hydrolysis of urea followed by the quantitation of ammonium by reaction with salicylate and dichloroisocyanate, which can be measured spectrophotometrically (Alef and Nannipieri, 1995). Ureases can exist in two possible states in the sediments. They may be intracellular, present within the cells of ureolytic microorganisms or alternatively extracellular, released from disrupted plant and microbial cells. Extracellular urease is largely adsorbed on clay and organic colloids which have a high affinity for urease. This adsorption on sediment colloids possibly protects ureases from destruction by sediment microorganisms, while the activity of enzymes remains essentially unimpaired (Alef and Nannipieri, 1995). Interest in this enzyme stems from its role in the decomposition of urea fertiliser in agricultural systems (Gianfreda et al., 2005; Rahmansyah et al., 2006). It is also considered an important agent for nitrogen mineralisation in terrestrial and aquatic systems (Kandeler et al., 1999; Sinsabaugh et al., 2000).

# 1.7 The Significance and Aims of This Study

Estuarine nutrient biogeochemical cycles are being modified by anthropogenic impacts increasing pollutions and wastes, reducing species abundance and diversity and causing imbalance to the ecosystem functions (Libes, 2009; Jjemba, 2004; Madsen, 2008). Human intervention has also affected the marine phosphorus cycle by doubling the source of phosphorus through sewage discharges and fertiliser runoff (Paytan and McLaughlin, 2007). Oxygen is one of the important factors mediating these biogeochemical cycles, which may also influence organic matter pathways in the estuaries (Bianchi et al., 2004). As a consequence of this, nutrients and organic matter accumulate in the system increasing the productivity of the system. This problem may further complicate the situation by generating eutrophication, causing algal blooms (Meyer-Reil and Koster, 2000; Howarth et al., 2011). Therefore, the link between oxygen and biogeochemistry of estuaries is a vital issue to look at in order to evaluate the oxygen concentration and at which harmful effects start to occur, the rates that the system responds to different levels of oxygen saturation in the overlying water and the recovery times of system processes when re-aeration are restored.

This will be achieved by investigating the effects of different oxygen saturation in the overlying water on the exchange and transformations of nitrogen and the fluxes of phosphate at the sediment-water interface. The linkage between nutrients in the overlying water and the nutrients in the porewater of the sediments were also investigated. Besides nutrients, five enzyme activities were also measured in the sediment (alkaline phosphatase, cellobiohydrolase,  $\beta$ -glucosidase, chitinase and urease) as enzyme activities can give information on the processes that may underlie an overall change in fluxes and are easier to use in small scale experiments (Marx et al., 2001; Saiya-Cork et al., 2002; DeForest, 2009). Additionally, the effect of different oxygen saturation to the pH, total organic carbon, total nitrogen, bromide and oxygen fluxes were also examined as parameters that may influence the nutrient fluxes and enzyme activities in the treatment.

Based on the aims above, the following questions were addressed to fill in the gaps in the related research:

- 1) Does short term hypoxia and anoxia affect pH in the overlying water and vertical profiles of total organic carbon and total nitrogen in the sediments?
- 2) Are differences in oxygen saturation in the overlying water reflected by changes in bromide solute transport between sediment-water interfaces and the oxygen consumption in the sediments?
- 3) How does oxygen affect nutrient cycling, which in turn influences nutrient fluxes?
- 4) Does the duration of hypoxia and anoxia influence the nitrogen and phosphorus transformations in the estuarine sediment-water system?
- 5) What factors affect enzyme activities in sediments?

#### **1.8** The Structure of the Thesis

This thesis has been divided into 5 chapters:

Chapter 1 provides the introduction to the current knowledge of hypoxia and anoxia distribution, processes affecting nitrogen and phosphorus cycles in estuarine

sediments and the background of the five enzyme activities selected in this study. The aims of the study and significant research questions were also given. Chapter 2 covers the experimental design, the methodology applied for the field sampling and laboratory experiments and the statistical analyses opted. Chapter 3 focuses on the result of sediment physico-chemical characteristics, solute transports, oxygen uptake, nutrients and porewater concentrations and enzyme activities in this study. Chapter 4 discusses the outcome and interpretation of the pattern observed in this study, and finally, Chapter 5 summarises the findings, stressing the importance of oxygen on modifying nutrient cycles and enzyme activities. The final section details the possible further work to be carried out.

#### **CHAPTER 2 METHODOLOGY**

#### 2.1 Introduction

The aim of Chapter 2 is to present the method used in the field sampling and the experimental work of this study. This chapter begins with an explanation of the experimental design, the fieldwork sampling and the experimental setup in the lab. Section 2.6 describes the procedure to determine the physico-chemical characteristics. Section 2.7 and 2.8 explains the method for analysing bromide and oxygen, respectively. This is followed by the determination of nutrients and porewater nutrient concentrations, as well as their fluxes (section 2.9-2.10). Section 2.11 explains the method and calculation for enzyme activities. The final section states the statistical analysis used in this study.

# 2.2 Experimental Design

The experimental design for the sample collection and processing of the sediment cores were shown in Figure 2.1. Samples were collected in June, 2010 using 8.4 cm diameter and 30 cm high Plexiglas cylindrical cores. The samples were used for analysing the sediment physico-chemical characteristics, nutrient and porewater nutrients, oxygen fluxes, and enzyme analysis. Samples were collected from Breydon Water, Great Yarmouth at the start and on day 7 of laboratory incubations. Bromide fluxes were taken on day 4 and day 7 of the laboratory incubation.

# 2.3 Field Sampling

Sediment samples were collected during low tide in June 2010 from Breydon Water estuary, Great Yarmouth (Figure 2.2). Plexiglas tubes were pushed 10 cm into the sediment, leaving 20 cm for the overlying water column. Care was taken as not to disturb the cores. The specification used for handling sediment cores was taken from the NICE (Nitrogen Cycling in Estuary) protocol handbook by Dalsgaard et al., (2000). By using undisturbed sediment, the natural diversity of organism as well as heterogeneity could be retained. Samples were returned to the University within approximately 45 minutes where processing of the cores commenced immediately.

Sediments which were not used straightaway were placed in the dark at 4°C as soon as possible and kept under these conditions until analysis.



Figure 2.1 Schematic representation of processes carried out on the cores collected from Breydon Water, Great Yarmouth in June 2010.

Initial surveys were carried out before selecting the most appropriate sampling location in Great Yarmouth. Breydon Water, Great Yarmouth was selected as the sampling location because of its accessibility and the short walking distance required transporting samples and equipment back to the lab. It is also an area with a low risk of natural hypoxia. The selection of a location without a history of natural hypoxic or anoxic condition was to ensure that the system was not hypoxic and that the only influence of hypoxia comes from the experimental setup in the study. This procedure follows the recommendation of Larson and Sundback, (2008).



Figure 2.2 Sampling during low tide in June 2010 at Breydon Water, Great Yarmouth

Breydon Water is an inland tidal estuary located in Norfolk, East Anglia, England, lying just to the east of Great Yarmouth (Figure 2.3). It is situated at the mouth of the River Yare and its confluence with the River Bure and Waveney. Rivers Yare and Waveney enter Breydon Water before discharging into the North Sea (Baban, 1997). Breydon Water is relatively shallow, with brackish and well mixed water, and a mean tidal range of 1.5 m. It covers an area of 15.34 km<sup>2</sup>, with a mean width of 1-1.2 km and an area of 7 km<sup>2</sup> in high tide whereas during low tide, the water is 10-12 m in width along the entire length of the estuary. The residence time of tidal waters within this estuary including the estuaries of River Yare and Waveney is very short, no more than 1-2 days (Sabri, 1977).

Extensive areas of mud are exposed at low tide and these form the only intertidal flats occurring on the east coast of Norfolk (Baban, 1997). The mudflats are characterised by growths of green algae, *Ulva* sp and two rare species of eelgrass, *Zostera marina* and Z. *noltii*. These plants, together with an abundant invertebrate fauna, particularly the polychaetes, *Nereis diversicolor*, attract large number of ducks

and waders to feed in the estuary, particularly during winter (Norfolk County Council, 2011).



Figure 2.3 Map of sampling site in Breydon Water, Great Yarmouth (source: EDINA Digimap)

# 2.4 Experimental Setup

Upon arrival to the lab, the cores containing 10 cm depth sediments were covered with 900 ml nutrient-free artificial seawater (Tropic Marin salt [Aquarium System]) dissolved in MQ at a salinity matching that of the sampling location (32 ‰). The use of artificial seawater is to allow full control of the experimental design and reduce variation between replicates. Natural seawater could contain high concentrations of suspended or dissolved material that are not appropriate for experimental systems. Water taken from shallow areas of estuaries may have been subjected to large diurnal changes in light, temperature, nutrients or dissolved oxygen that could affect the experimental results. It is possible that the use of artificial seawater could cause rapid nutrient escape from the *in-situ* sediment into the overlying artificial seawater, with the consequences of disrupting natural processes. The artificial seawater was slowly

added into the cores up to 900 ml to avoid sediment resuspension (Rubao et al., 2002). The depth above the sediment should be 10-20 cm to give a volume of 0.5-1 l of overlying water, following the recommendation by the NICE protocol handbook (Dalsgaard et al., 2000).

The cores were covered with a plastic lid with two holes, one with a rubber stopper for water sampling and the other one for the tubing to allow air or nitrogen bubbling (Figure 2.4). Aeration tubes were placed 10 cm above the sediment surface with the bubbling rates at 50 ml/min for oxic treatments, 100 ml/min gas mixtures for hypoxic treatments and 100 ml/min nitrogen gas flow for anoxic treatments. The flow of both gases was regulated through valves and set using a gas flow regulator. The cores were maintained in the absence of light to quantify benthic respiration (dark fluxes). The cores were kept at a constant temperature of 12 °C in a cold room. This is approximately equivalent to the *in-situ* temperature in the study area. Prior to carrying out the experiments, a preliminary check was conducted to ensure the correct oxygen concentrations were obtained for each treatment. The appropriate gas/gas mixture was allowed to bubble in each sediment core for approximately 1 hour. Dissolved oxygen concentrations were then measured and adjustments to the gas flow in each sediment core were made each additional 0.5 hour. This procedure was repeated until the concentration were within the concentrations desired which generally took 4 to 5 hours. Any potential leaking of nitrogen gas from loose connection of the gas cylinders to the aeration tubes and cores were also checked. Leakage of nitrogen into the cold room was also monitored using an oxygen meter to give the alarm if the oxygen concentration in the cold room has reduced.

The experimental setup consisted of five treatments, n = 25; 1) oxic from continuous aeration of the treatment with air pump for 7 days, 2) hypoxic conditions in the overlying seawater obtained by continuous bubbling with 3% oxygen, 97% nitrogen gas mixtures for 7 days, 3) 1 day anoxic (1DA) by bubbling with oxygen free nitrogen, followed by subsequent re-aeration for another 6 days with air pump, 4) 4 days anoxic (4DA) by bubbling with oxygen free nitrogen, followed by reaeration with air pump for 3 days and 5) anoxic bubbled with oxygen free nitrogen for 7 days. These five different treatments were selected as to induce oxic conditions (95% oxygen saturation), hypoxic conditions (25% oxygen saturation), anoxic conditions (0% oxygen saturation), and to determine how fast the system can recover from short term anoxic events (4 day and 1 day anoxic) followed by reaeration over 3 days and 6 days, respectively. The 7 day experiment was selected in order to capture the consequences of episodic hypoxia, which often lasts from several days up to a week, with a typical duration of 5 days (Sagasti et al., 2003). During the laboratory incubations, the cores were sealed with plastic lids to prevent oxygen dissolution to the overlying seawater from the air in the cold room and to maintain the oxygen concentration needed for the different treatments.



Figure 2.4 Experimental setup for the different aerations conducted in a constant temperature room.

## 2.5 Laboratory Protocol

Apparatus used in the laboratory was cleaned of visible dirt then washed and soaked in 'phosphate free' detergent (Decon 90) at least overnight. The apparatus was then rinsed three times and on subsequent occasions with milliQ water (MQ). Next, it was placed in 10% HNO<sub>3</sub> overnight to further remove contaminants, re-rinsed three times with MQ and allowed to air-dry. Apparatus used in association with standards and diluted samples were rinsed with milliQ water before use. All dilutions of samples and reagents were made up in milliQ water. Apparatus used in the field was scrubbed with detergent and rinsed with milliQ water.

### 2.6 Determination of Physico-chemical Parameters

# 2.6.1 Determination of Particle Size Distribution of Sediments

Particle size analysis was measured with 1 g sediment samples at 1, 3, 5, 7 and 9 cm depth for general characteristics of the field site. Sediments were suspended in 70 ml distilled water and wet sieved through a 1.7 mm Fisherbrand test sieve (Fisher Scientific). The slurry was treated with 1 ml of 10 % Calgon (sodium hexametaphosphate) to avoid flocculation (Bartoli et al., 2009). Then, the samples were sonicated in an ultrasonic bath for 1.5 minutes and stirred with a magnetic stirrer to ensure full mixing. Samples were analysed using a laser diffraction method with a Malvern Mastersizer 2000.

## 2.6.2 Determination of Overlying Water pH

pH values was measured with a pH meter (HI 02885) which was calibrated using pH 4, 7 and 9 buffers. The pH electrode was dipped into the overlying water daily for 7 days.

#### **2.6.3** Determination of Porosity in Sediments

Sediment samples were sliced into 2 cm thick sections from the surface to 10 cm depth of the sediment at the start and on day 7 of the laboratory incubations. Sediment slicing on day 7 was carried out under gas mixtures for hypoxic treatments and an oxygen free environment in a glove box for anoxic treatments. Sediment samples were placed in a weighing boat, labelled with a marker and the mass of each boat were carefully determined on an analytical balance (Sartorius). A 12-ml plastic syringe with the tip removed was used to collect a 2-ml subsample from each corer. The syringe was used like a miniature corer, pushing the syringe into the mud while simultaneously pulling back on the plunger. The sub-sampled mud was extruded into appropriate weighing boat and the sediment wet weight was recorded. Sediment samples were dried to constant weight for 24 hour at 70 °C (Hietanen and Lukkari, 2007) and the dry weight of the sediments recorded.

Sediment porosity ( $\phi$ ) was determined as:

# **Equation 2.1:**

 $\phi = \frac{ww-dw}{vol}$ 

Where ww and dw are the wet weight and dry weight of sediment, respectively and vol is the sediment volume.

# 2.6.4 Determination of Total Organic Carbon and Total Nitrogen in Sediments

Sediment samples of approximately 10 g were dried from each depth of 0-2 cm depth intervals until 10 cm depth at 80 °C for 24 hour in a foil tray. Samples were taken at the start and on day 7 of the laboratory incubations. The dried sediments were crushed into fine particles with a mortar and pestle under gas mixtures for hypoxic treatments and oxygen free environment in a glove box for anoxic treatments. The mortar and pestle were cleaned and air-dried with three successive washes of methanol, acidic-methanol, and dichloromethane prior to adding the respective sediment layers to avoid cross-contamination. Powdered samples were stored at room temperature flushed with gas mixtures for hypoxic treatment and nitrogen in

anoxic treatment before being capped in pre-combusted glass scintillation vials until analysis.

Prior to analysis, the powdered sediments were weighed to about 18-22 mg and then folded into a tin capsule for total nitrogen analysis. Sediment samples for total organic carbon analysis were initially treated similarly except that samples were weighed into double tin capsules, one inside the other, and the inner capsules were not sealed until after treatment to remove the inorganic carbon. Inorganic carbon was removed by treating the weighed samples with 8% sulphurous acid according to the method of Verado et al., (1990). Two tin capsules were used in order to prevent any sample lost due to weakening of the inner capsule by the acid.

Three procedural blanks were prepared from tin capsules treated with acid but without sediments. All reagents and glassware used to analyse samples are combusted at 400 °C for a minimum of 4 hours. Acetanilide standards were also weighed to approximately 2 mg and folded in a tin capsule. Samples were analysed using a CHN EA1108 Elemental Analyser (Hansen and Kristensen, 1998). The analytical technique used for the determination of carbon and nitrogen is based on the quantitative dynamic flash combustion method. This method is the complete and instantaneous oxidation of the sample, where all the organic and inorganic substances are converted into combustion products. The samples were placed inside the autosampler drum where they were purged with a continuous flow of helium and then dropped at preset intervals into a vertical quartz tube (combustion reactor) maintained at 1000 °C. When the samples dropped into the furnace, the helium stream was temporarily enriched with a precise volume of pure oxygen. The sample and its container melt and the tin promote a violent reaction called flash combustion in the temporarily enriched atmosphere of oxygen, when the temperature rises to around 1800 °C.

This instrument was calibrated at the beginning, end and every fifteen samples during a run using sulphanilamide standards and empty capsules as blanks. Five replicates of each sediment sample were made from 2 cm depth intervals from the five different treatments, n = 125. Percentage organic carbon and nitrogen values

were calculated using Eager 200 software using the K-factor method. This K value is determined by analysing an organic standard of a known elemental composition.

# 2.7 Bromide Tracer

Movement of bromide tracer into the sediment was assessed from day 4 to day 7 of the laboratory incubation. This procedure is done to determine the effect of oxygen and re-aeration in the overlying water on solute transport between the sediment and water at the end of the incubation. 1 ml of 6.4 mol L<sup>-1</sup> calcium bromide was added to the water column, corresponding to 16 000  $\mu$ mol L<sup>-1</sup> of initial bromide concentration in the overlying seawater. This concentration was selected as the bromide concentration spiked to the cores should be substantially higher than the background value (Martin and Banta, 1992). It is recommend that the concentration should be at least 10 times higher and is in line with the usual concentration of the bromide spike used in solute transport experiments which ranges between 10 000-20 000  $\mu$ mol L<sup>-1</sup> (Martin and Banta, 1992; Forster et al., 1995; 1999; Keefe et al., 2004; Quintana et al., 2007). The background porewater concentration in the cores was 700 to 900  $\mu$ mol L<sup>-1</sup>.

On day 4 to day 7 of the incubation, 20 ml overlying water samples were taken with a syringe for the determination of bromide concentration. In the case of hypoxic and anoxic treatments, the syringe was flushed for a few seconds with gas mixtures and nitrogen gas before taking water samples for analysis. The water samples were filtered and stored in 30 ml universal bottles at -20 °C until further analysis using inductively coupled plasma mass spectrometer (Thermo X5 series I ICP-MS, Thermo Electron Corporation, USA).

All samples for ICP-MS analysis were sampled through a Cetec 500 autosampler. ICP-MS was tuned using the standard mode with a solution containing lithium, indium and uranium at 10  $\mu$ g L<sup>-1</sup> each. The multielement tune A from Thermo Electron Corporation containing barium, beryllium, bismuth, cerium, cobalt, indium, lithium and uranium at 10 mg L<sup>-1</sup> was used to prepare the tuning solution. The passing criterion for this tune is shown in Table 2.1. The operating conditions applied for bromide concentrations are listed in Table 2.2.

Isotope	Counts per second			
<sup>-7</sup> Li	>1 000			
<sup>115</sup> In	>10, 000			
<sup>238</sup> U	>10,000			
<sup>7</sup> Li/ <sup>5</sup> Bkg	>2 000			
<sup>115</sup> In/ <sup>101</sup> Bkg	>10,000			
<sup>156</sup> Ce O/ <sup>140</sup> Ce	<0.02			
<sup>238</sup> U/ <sup>220</sup> Bkg	>10,000			

Table 2.1 Passing criteria for standard mode of ICP-MS

Table 2.2 ICP-MS operating conditions for bromide measurements

Radio Frequency power	1100W
Argon gas flow rates:	
-plasma	15.0 L/min
-auxiliary	0.8 L/min
-make up	1. 1 L/min
Sample flow rate	0.85 mL/min

Bromide was monitored at mass to charge ratio, m/z 79 and 81. Linear calibration curves were found for standards in the range of 200 to 1000 µmol L<sup>-1</sup> prior to sample measurements using the software Plasma Lab. Porewater bromide concentrations were diluted to 1:20 using a Varian SP55 autodilutor.

The experiment was terminated after 3 days of incubation with bromide. Core slicing was carried out in an oxygen free environment using a glove box to assess the movement of bromide throughout the 3 days of incubation in each treatment. Before extruding the sediments, the water above the sediment was removed with a syringe. The rest of the sediment was sectioned at 2 cm intervals by extruding it upwards from the core tube and sectioning it with a thin plastic slicer. A clean slicer was used for each sectioning to avoid contamination between slices (Bradshaw et al., 2006). 10 ml porewater samples were centrifuged from the sediment at 4,500 rpm for 10

minutes and transferred to 15 ml Corning polypropylene centrifuge tubes stored at - 20 °C and analysed with the same procedure as seawater.

# 2.7.1 Total Bromide Incubation, Diffusive and Irrigation Fluxes Measurements

Total bromide incubation fluxes were calculated based on the equation:

#### **Equation 2.2:**

$$Flux = (C_{f} - C_{i}) \times V$$
  
A x t

Where;

Flux = flux of bromide and nutrients ( $\mu$ mol m<sup>-2</sup> h<sup>-1</sup>)

$$C_f$$
 = final concentration of bromide and nutrients (µmol L<sup>-1</sup>)

 $C_i$  = initial concentration of bromide and nutrients (µmol L<sup>-1</sup>)

A = area of sediment surface in core  $(m^2)$ 

V = volume of water in core (L)

t = incubation time (h)

The diffusive bromide fluxes ( $\mu$ mol m<sup>-2</sup> h<sup>-1</sup>) were calculated from the bromide profile with depth according to Fick's first law of diffusion:

#### **Equation 2.3:**

 $J = -\phi Ds (\partial C / \partial x)$ 

Where  $\phi$  is porosity

Ds is the whole sediment diffusion coefficient

 $\partial C / \partial x$  is the concentration gradient (mol m<sup>-4</sup>)

The sediment diffusion coefficient, Ds, is determined according to:

 $D_s = D_o/\theta^2 (m^2 h^{-1})$ 

Where  $D_o$  is the effective temperature-corrected diffusion coefficient at 12 °C for bromide, yielding a value of  $1.4389 \times 10^{-9} \text{ m}^2 \text{ s}^{-1}$  (Li and Gregory, 1974; Boudreau, 1997). For conversion of seconds to hour,  $D_s$  are multiplied by 3600.

 $\theta^2 = 1$ - ln ( $\phi^2$ ),  $\phi$  is the porosity which accounts for tortuosity correction according to Boudreau and Meysman, (2006).

 $\partial C/\partial x$  is determined from linear regression of the concentration profiles in the sediment (mol m<sup>-4</sup>).

Irrigation fluxes were calculated from the difference between total bromide incubation fluxes with the diffusive bromide fluxes. Since the sediment was not sieved, we assumed that there was active solute transport through irrigation.

# 2.8 Determination of Oxygen Penetration Depth in Sediments

Oxygen profiles were measured with a dip-type oxygen minielectrode (Microelectrodes, Inc., USA) at the start of the incubation and just before the end of the incubations. Five oxygen profiles to depths were measured from each core. The minielectrode consists of an Ag reference electrode with a tip diameter of 3 mm and up to 8.6 cm in length. The oxygen minielectrode was connected to an oxygen adapter (O2-ADPT) and a voltage reading device (HI 9025 pH meter). Oxygen vertical profiles of the surface sediments were measured in increments of 0.5 mm depths because dissolved oxygen concentration shows an abrupt change within the upper 1 cm (Kim and Kim, 2007). The 90% response time is 150 seconds.

To measure oxygen profiles within the sediment, the electrode was attached to a rod, to allow penetration into the sediment, which was lowered using a retort stand equipped with a precision screw slide to ensure measurement of 0.5 mm vertical resolution. A hand lens is used to magnify the view of the minielectrode tip and the tip of the probe was carefully lowered to the sediment surface. Measurements were made at 0.5 mm vertical intervals starting from 1 mm above the sediment followed by 0.5 mm intervals down to a depth in the sediment where milivolt (mV) reading reached near zero, indicating a zero oxygen concentration. Before measurement, the minielectrode was calibrated using a two point calibration, one with 0 % gas bubbled with nitrogen for approximately 30 minutes and the other with 20.9 % oxygen (ambient air). At 20.9 %,  $-300 \pm 7.9$  mV reading was obtained and at 0 % oxygen,  $-0.7 \pm 0.4$  mV was recorded, n= 5. The mV readings obtained from the calibration process were then converted to percent oxygen of the overlying seawater in the treatment. The percent oxygen saturation was converted to oxygen concentrations using values from the table provided by Ramsing and Gundersen, (2001) (available at http://www.unisense.com/).

#### 2.8.1 Measurements of Diffusive Oxygen Fluxes in Sediments

Diffusive oxygen fluxes or sediment oxygen consumption within the sediment was estimated from the oxygen gradient within the sediment utilising Fick's 1<sup>st</sup> Law using Equation 2.3. This procedure assumes molecular diffusion-limited rates in the sediment (Berner, 1980). But, the values were multiplied by 86 400 to convert the flux data from hours to day. For this calculation, we used the temperature-dependent oxygen diffusion coefficient according to Ramsing and Gundersen, (2001), yielding the value of  $D_o=1.5702 \times 10^{-9} \text{ m}^2 \text{ s}^{-1}$  at 12 °C and a salinity of 32.0 ‰ (available at <u>http://www.unisense.com/</u>). Porosity was measured in centimetre scale (section 2.6.3), which is different than the sediment oxygen profiles which are measured at milimetre scale (section 2.8), therefore, porosity was estimated from extrapolation (using polynomial curve fit) of the measured porosities (Hulth et al., 1994).  $\partial C/\partial x$  is determined from the linear portion of the oxygen gradient in the top layer of the sediment.

# 2.9 Nutrients Sampling in the Overlying Seawater

To measure nutrient fluxes (nitrate, nitrite, ammonium and phosphate), 20 ml of overlying water were sampled daily for 7 days to capture rapid changes to the system and to assess nutrient changes for benthic respiration (dark fluxes). The water samples were filtered using 0.45  $\mu$ m cellulose filters (Jiang et al., 2006; Belias et al., 2007). At the end of the incubation, the height of water in the cores and the volume of water over the cores were noted. The 20 ml nutrient samples were transferred to

30 ml universal bottles and kept at -20 °C and were analysed by the standard autoanalyser method using SKALAR autoanalyser.

## 2.9.1 Determination of Nitrate and Nitrite Using a SKALAR Autoanalyser

The method is based on the cadmium reduction method. The sample was buffered at pH 8.2 and passed through a column containing granulated copper cadmium to reduce the nitrate to nitrite. The nitrite (originally present plus reduced nitrate) was determined by diazotising with sulphanilamide and coupling with N-(1-naphtyl) ethylenediamine dihydrochloride to form a highly coloured azo dye. The pink coloured dye produced was measured at an absorbance of 540 nm.

Working standards, ranging from concentrations of 0 to 20  $\mu$ mol L<sup>-1</sup> sodium nitrate (Figure 2.5 a) and 0 to 5  $\mu$ mol L<sup>-1</sup> sodium nitrite (Figure 2.5 b) were prepared fresh daily and run before the analysis of samples every day.

## 2.9.2 Determination of Ammonium Using a SKALAR Autoanalyser

The automated procedure is based on the modified Berthelot reactions. Ammonium was chlorinated to monochloroamine which reacted with phenol. After oxidation and oxidative coupling, a green coloured complex was formed. The reaction was catalysed by nitroprusside; sodium hypochlorite was used for chlorine donation. The absorbance of the coloured complex was measured at 630 nm.

Working standards were prepared from concentrations of 0 to 200  $\mu$ mol L<sup>-1</sup> ammonium chloride (Figure 2.5 c). A 10 fold serial dilution was carried out for porewater ammonium concentrations. The working standards were prepared fresh daily from the stock solution before the ammonium analysis was carried out.

#### 2.9.3 Determination of Phosphate Using a SKALAR Autoanalyser

Phosphate was determined by the reaction of ammonium heptamolybdate and potassium antimony (III) oxide tartrate in an acidic medium with diluted solutions of phosphate. This step formed an antimony-phospho-molybdate complex. This complex was reduced to an intensely blue-coloured complex by L (+) ascorbic acid and measured at 880 nm absorbance.Working standards of 0 to 5  $\mu$ mol L<sup>-1</sup> (Figure 2.5 d) were prepared fresh daily.



Figure 2.5 Nutrients standard calibration curve for a) nitrate, b) nitrite, c) ammonium and d) phosphate. Standards are presented as mean  $\pm$  SE, n = 3.

#### 2.9.4 Measured Nutrient Fluxes in the Overlying Water

Fluxes ( $\mu$ mol m<sup>-2</sup> h<sup>-1</sup>) were determined from the concentration changes, incubation time and the measured height of the water column in core during incubation. Nutrient fluxes were calculated using equation 2.2. Positive flux values represent a concentration increase in the overlying water, from the flux of the sediment to the overlying water. Negative flux values represent a decrease in the concentration of overlying water, which may be an indication of a flux into the sediment, or loss to other forms.

## 2.10 Analysis of Porewater Nutrients in the Sediment

All sectioning of cores, centrifugation and handling of unpreserved porewaters was done in an oxygen free environment. This process was carried out in a glove box which is flushed with oxygen free nitrogen (OFN) to minimise oxidation of reduced species in the porewaters (Gillespie, 2005). Samples taken from the cores immediately upon arrival to the lab, 45 minutes represented the start condition. The cores were extruded carefully to minimise compaction at the top and leakage of water from the base. Porewater was obtained by centrifugation for 10 min at 3000 rpm in 50 ml sterile polypropylene centrifuge tubes, which were pre-flushed with OFN before use and sealed in the glove box. Before and after the 7 day incubation period, sediment samples were taken to determine pore water solute concentrations by cutting the cores into 0-2, 2-4, 4-6 and 6-8 cm sections. From each section, roughly around 70 g of sediment were obtained which produced between 5-8 ml of porewater. The porewater was then filtered using a 0.2 µm disposable filter into 15 ml sterile polypropylene tubes.

The supernatant was analysed for nitrate, nitrite, ammonium and phosphate. Once extracted, these samples were frozen and later analysed within a week. They were defrosted for about 2 hours at room temperature before analysis was carried out using the SKALAR autoanalyser. The method used for porewater nutrients analysis is the same using the SKALAR method mentioned under section 2.9.1 to 2.9.3.

#### 2.10.1 Calculated Porewater Diffusive Fluxes in the Sediment

Theoretical diffusive fluxes were calculated from porewater gradients using equation 2.3. The effective diffusion coefficient (Ds) of a nutrient in the sediment was corrected for a temperature of 12 °C from molecular diffusion coefficients,  $D_o$  given by Boudreau, (1997). This yields a value of  $1.33 \times 10^{-9}$  m<sup>2</sup> s<sup>-1</sup> for nitrate,  $1.34 \times 10^{-9}$  m<sup>2</sup> s<sup>-1</sup> for nitrite,  $1.36 \times 10^{-9}$  m<sup>2</sup> s<sup>-1</sup> for ammonium and  $0.41 \times 10^{-10}$  m<sup>2</sup> s<sup>-1</sup> for phosphate. To estimate the gradient close to the sediment-water interface, different calculations were applied depending on the vertical profile of the nutrients considered. For nitrate and nitrite concentrations, the gradient was calculated using the third order polynomial curve fitting. For ammonium and phosphate concentrations, a linear regression over the upper 9 cm depth was considered.

# 2.11 Measurement of Enzyme Activities in the Sediment

Linearity test and substrate saturation test were carried out to validate the measurement of enzyme activities (section 2.11.1 to 2.11.2). Section 2.12 describes the reagents used for enzyme activities measurement. Section 2.13 explains the two types of assays used in the measurement of enzyme activities.

#### 2.11.1 Linearity Test

The validation process indicated that fluorescence intensity of the fluorochrome analogs, released due to alkaline phosphatase, cellobiohydrolase,  $\beta$ -glucosidase, and chitinase activity in the sediment, increased linearly with time (Figure 2.6 a-d). This linear association shows that fluorescence intensity increases with the amount of fluorochrome in solution and that enzyme activity was not induced or repressed within the 4 h incubation. Absorbance intensity for urease increased linearly with time for up to 20 hours of incubation in Figure 2.6 e.



Figure 2.6 Analog fluorescent as a function of incubation time for a) alkaline phosphatase, b) cellobiohydrolase, c) beta-glucosidase and d) chitinase. e) Analog absorbance as a function of incubation time for urease. Values are reported as mean  $\pm$  SE, n= 5.

## 2.11.2 Saturation Test

Substrate saturation curves were carried out for all 5 enzymes measurement to determine which substrate concentration to be used in this study. For all the enzymes measured using fluorometric methods, 200  $\mu$ M substrate was sufficient to saturate the enzyme. This is shown in Figure 2.7 a-d for alkaline phosphatase, cellobiohydrolase,  $\beta$ -glucosidase and chitinase. This concentration was proposed as the saturated substrates concentration of these 4 enzymes by Sinsabaugh et al., (2000); Saiya-Cork et al., (2002) and DeForest, (2009). Urease was saturated at 20 mM as shown in Figure 2.7 e. Alef and Nannipieri, (1995) have reported substrate saturation of urease between 5-40 mM, depending on the sediment used.





Figure 2.7 Substrate saturation curves on fluorescence intensity of a) alkaline phosphatase, b) cellobiohydrolase, c) beta-glucosidase d) chitinase and absorbance intensity of e) urease. Values are reported as mean  $\pm$  SE, n= 5.

#### 2.12 Reagents

# 2.12.1 Substrates

Methylumbelliferyl substrates (MUB) were used for 4 of the enzyme measurement, 4-methylumbelliferyl (MUB)-phosphate, 4-MUB-  $\beta$ -D cellobioside, 4-MUB- $\beta$ -Dglucopyranoside and 4-MUB-N-acetyl-  $\beta$ -D glucosaminidase. All substrate analogues were purchased from Sigma-Aldrich Co. Ltd in crystalline form. The use of MUB substrates has a close relationship to naturally occurring processes (Hoppe, 1993). All substrates were made up to 10 ml with sterile deionised water and the resulting stock solutions of 10 mM were kept at 4°C. Working solutions for each substrate was diluted to 1000, 100 and 10  $\mu$ M with buffer in sterile 30 ml universal bottles to establish substrate saturation curves. Solutions of MUB-standards and MUB-linked substrates were prepared fresh during the day of the analysis, to prevent rapid changes of MUB-standard and MUB-linked substrates that can occur within the first 24 hours (DeForest, 2009). The final substrate concentrations in the plates were 2, 5, 8, 20, 50, 80, 100, 200, 300, 500 and 700  $\mu$ M.

# 2.12.2 Buffers

All enzyme assays were carried out in buffered conditions in order to reduce the effects of pH changes on the results. The use of buffer also stabilises the fluorescence intensity which is highly dependent on pH (Alef and Nannipieri, 1995) and allow comparisons to be made with other system (German et al., 2011). A 50 mM acetate buffer solution was used by mixing sodium acetate trihydrate (certified ACS grade, crystalline, Sigma-Aldrich) with deionised water for fluorescence assays. Various studies for enzyme measurements have used 50 mM acetate buffer in their study for sediments (Niemi and Vepsäläinen, 2005; Sinsabaugh et al., 2005). The pH was adjusted to 7.8 for all enzyme measurements using either sodium hydroxide (NaOH) or hydrochloric acid (HCl). Generally most of the sediments in this study exhibited pH ranges of 7.5 to 8.0 within the sediment porewaters, therefore average range of pH 7.8 was selected for all enzymes. For alkaline phosphatase (AP) activity measurement in the sediment, usually a near neutral pH was used (Alef and Nannipieri, 1995) and AP was found to react optimally in pH range of 7.6 - 9.6(Cunha et al., 2010). Niemi and Vepsäläinen, (2005) have observed a pH optimum of 7.5 or higher for alkaline phosphatase activity, chitinase and  $\beta$ -glucosidase activity.

#### 2.12.3 Standards

A 10 mM stock solution of 4-MUB was prepared by dissolving 0.1762 g of 4-MUB in 100 ml of methanol. This stock solution was diluted to concentrations, ranging from 1-10  $\mu$ M with acetate buffer and prepared fresh on the day of the analysis. These concentrations are good for the sediment samples in this study.

### 2.13 Measurement of Potential Enzyme Activity

Initial enzyme activities were measured from sediment cores immediately upon arrival to the laboratory. The effect of different oxygen concentrations on enzyme activities was measured at the end of the incubation on day 7. Substrates, buffers, standards and incubation assays in the microplates were purged with gas mixtures for hypoxic treatments and nitrogen gas for anoxic treatments prior to use. Each sediment core was placed over a core extruder and sectioned with a metal spatula at 2 cm intervals down to 10 cm. Enzyme assays were conducted in five replicates for each depth interval (0-2, 2-4, 4-6, 6-8 cm). For clarity, each depth interval was depicted as 1, 3, 5, 7 and 9 cm on the y-axis of graphs. The enzyme activity assay gives the maximum potential activity of enzymes occurring in sediment rather than the actual enzyme activity (Alef and Nannipieri, 1995). Enzyme activity is determined by measuring the increase in fluorescence signal over a fixed time interval (Boetius et al., 2000).

#### 2.13.1 Fluorometric Assay

1.0 g of sediment suspensions was mixed with 125 ml 50 mM acetate buffer at a pH of 7.8 in a 250 ml beaker using an ultrasonicator water bath for 15 minutes. Sediment slurries are necessary to reduce diffusion barriers and ensure that enzymes have adequate access to substrate (Allison et al., 2007). Ultrasonication provides better homogenisation of the samples and released extracellular enzymes previously immobilised in humic colloids (De Cesare et al., 2000). It also completely break up macro aggregates (>250 µm) but does not disrupt smaller particles (Stemmer et al., 1998). The resulting suspensions were continuously stirred using a magnetic stirrer plate while samples were dispensed into 96-well microplates. The combination of using both ultrasonication and magnetic stirrer is adopted to reduce variability in enzyme assays using microplate technology and to homogenise the sediment slurry thoroughly as to represent the initial material. Before pipetting the assay solution containing the reaction product into new plates, the particles were ensured to be settled during the incubation period. This step is to prevent particles from sample homogenate causing light beam scattering of the microplate reader (Allison et al., 2007).

The dispensing of buffer, sample, standards and substrates followed a strict order and position on the well plate. First, 200  $\mu$ l of buffer was dispensed in wells that served as the blank, reference and substrate control. Next, 50  $\mu$ l of buffer was dispensed in columns that served as the sample controls for the sediment. Sample controls and blank was analysed to account for naturally occurring fluorescence of

the buffer or sediment within the well plate. Sediment turbidity exerts a quenching effect on the fluorescence intensity of MUB. Therefore, a quench correction procedure was applied to the sediment samples. Then, 200  $\mu$ l of sediment samples was dispensed into wells that served as the quench, sample control and sample assay (Saiya-Cork et al., 2002; DeForest, 2009). The steps in the sediment samples, references and substrate in each well are summarised in Table 2.3. Fifty microliters of a 10  $\mu$ M MUB solution was dispensed into wells that served as reference standard and quench standard. Lastly, 50  $\mu$ l of a 200  $\mu$ M MUB-linked substrate was dispensed into plates that served as a substrate control and the sample assays. A substrate concentration of 200  $\mu$ M was used according to the substrate marked the start of the incubation period and time was noted. The microplates were incubated in the dark at 12 °C for up to 4 h (Saiya-Cork et al., 2002; DeForest, 2009). The selection of the assay temperature is equivalent to the time of sampling and incubations.

The samples to be assayed, buffer, substrates and standard concentrations of 250  $\mu$ l are pipetted into 96 well, 300  $\mu$ l black microwell plates (Nunclon Inc.) and fluorescence measured with a microplate reader (Spectramax M2). The fluorescence intensity was measured with 365 nm-excitation and 450 nm emission filters (Saiya-Cork et al., 2002). This analytical equipment allows incubation of the microplate at constant temperature. For the measurements, the microplate reader was programmed to shake the microplate for 5 s in order to homogenise the reaction medium. After finishing the readings, the computer software (Softmax Pro) was used to calculate the relative fluorescence intensity increase in each well. Since the rate of fluorescence increased was measured rather than the absolute amount of fluorescence at the end of the incubation period, a time zero blank was not needed but anyway, only accounted for minimal background fluorescence intensity (Marx et al., 2001). Initial experiments demonstrated that background fluorescence did not change during the incubation time. Calculation of enzyme activity is based on subtraction from control and for standards, with quench and reference standard, based on DeForest, (2009).

Enzyme activities were expressed in units of nmol  $g^{-1} h^{-1}$  and calculated from the raw data to final values by the following incubations and formula:

Table 2.3 Incubation of sample assay (SA), sample quench (SQ), reference standard (RS), quench coefficient (QC), substrate control (SuC) and sample control (SaC) of AP, CB, BG, and chitinase activities in wells

SA : Mean fluorescence of 200 $\mu$ L sample + 50 $\mu$ L substrate				
SQ : Mean fluorescence of 200 $\mu$ L sample +50 $\mu$ L standard				
RS : Mean fluorescence of 200 $\mu$ L buffer + 50 $\mu$ L standard				
QC : SQ/RS				
SuC : Mean fluorescence of 200 $\mu$ L buffer + 50 $\mu$ L standard				
SaC : Mean fluorescence of 200 µL sample +50 µL buffer				

Formula for enzyme activities calculation:

## **Equation 2.4:**

Activity (nmol  $g^{-1} h^{-1}$ ) = <u>NF</u> ×<u>125 ml total sediment slurry</u> EmC × 0.2 ml sediment slurry per well × Time (h) × sediment in fresh sample (g)

Where:

Equation 2.5:  $NF = (\underline{SA - SaC}) - SuC$ (QC)

Equation 2.6: EmC (relative fluorescence unit nmol<sup>-1</sup>) =  $\underline{RS}$ 0.5 nmol per well

NF = net fluorescence SA = sample assay SaC = sample control SuC = substrate control QC = quench coefficient EmC = Emission coefficient RS = Reference standard

The 125 ml refers to the volume of sediment slurry and 0.5 nmol is the amount of MUB-standard added to a well. Fluorescence values were means from six analytical wells subtracted from the sample control.

Inhibition of the reaction and maximisation of the fluorescence intensity through alkalinisation of the medium was not carried out because at pH 7.8 used in this study, the fluorescence intensity and high sensitivity were detected. The elimination of this procedure further minimise the time consumed in sample preparation and analysis, offering a better potential for use in routine enzyme measurements.

#### 2.13.2 Colorimetric Assay

Urease activity was measured colorimetrically in clear, flat bottomed polystyrene (Nunclon, Inc.) 96-well, 300 µl microplates. Urease activity was calculated from ammonium concentration using 96 clear microplates after 8 hours of incubation at 12 °C (Sinsabaugh et al. 2000). The incubation time and temperature selected were based on the linearity test (Figure 2.6) and the temperature during sampling. A sample suspension was created by homogenising 1.0 g sediment in 100 ml of 50 mM, pH 7.8 borate buffer, using an ultrasonicator and a magnetic stirrer. More concentrated suspensions have more activity but they generate a higher absorbance background (Sinsabaugh et al., 2000). pH 7.8 buffer was used to approximate the pH of bulk sediment in this study and to control the intensity of the absorbance. The optimum pH for urease activity is within 7-10 (Alef and Nannipieri, 1995). While stirring vigorously, 200 µl aliquots were withdrawn and dispensed into the microplate wells using a pipette. Three microplate columns (24 wells) were used per sample. Aliquots (10 µl) of substrate solution (400 mM urea in deionised water), were dispensed into the wells of two columns (16 wells) yielding a final urea concentration of 20 mM based on the substrate saturation curve (Figure 2.7).

The third column of wells acted as negative controls; each well received 10  $\mu$ l of water. Each microplate also contained a substrate control: a column of wells that each contained 200  $\mu$ l of buffer and 10  $\mu$ l of substrate solution. Ammonium released was quantified by dissolving 8.5 g of sodium salicylate in 85 ml of water and adding 0.06 g of sodium nitroprusside and diluted to 100 ml. Cyanurate reagent was prepared by dissolving 2.4 g of sodium hydroxide in 90 ml of water, and when it has cool, 0.5 g of sodium dichloroisocyanurate was added and diluted to 100 ml. The reagents were kept in amber bottles and wrapped with aluminium foil. The sample in the microplate (150  $\mu$ l) was treated with 60  $\mu$ l of salicylate reagent, followed by 60  $\mu$ l of cyanurate reagent and 30  $\mu$ l of water.

Standards, prepared from ammonium chloride, were added to the microplate. Standards ranging from 1 to 10  $\mu$ M were used for the calibration curve, which are good within the incubation time and analysis of the sediment samples in this study. Wells in standard columns contain 200  $\mu$ l of standard and 10  $\mu$ l of water. The wells containing standards constituted the positive controls for the assay. Standards were also prepared using sample suspensions which allows for the potential adsorption of ammonium to particles. In this case, 10  $\mu$ l aliquots of ammonium chloride used as standard, ranging from 20 to 200  $\mu$ M, were added to wells containing 200  $\mu$ l of sample suspension. The absorbance of each well was then read at 610 nm using a spectrophotometric microplate reader. Well to well variation can be minimised by adequate homogenisation, vigorous mixing of samples while dispensing, good pipetting technique (Sinsabaugh et al., 2000) and by running a larger number of replicate samples and controls (i.e. 16). Enzyme activities were expressed in units of nmol g<sup>-1</sup> h<sup>-1</sup> and calculated by the following formula and summarised as below in Table 2.4 (Sinsabaugh et al., 2000; Saiya-Cork et al., 2002):

Table 2.4 Incubation of SA, SuC, SaC and extinction coefficient (ExC) for urease activities.

SA : Mean absorbance of 200 $\mu$ L sample + 10 $\mu$ L substrate
SuC : Mean absorbance of 200 $\mu$ L buffer + 10 $\mu$ L standard
SaC : Mean absorbance of 200 $\mu$ L sample + 10 $\mu$ L buffer
ExC : Slope of absorbance versus concentration of $NH_4^+$ under the conditions of the
assay

Formula for enzyme activities calculation:

**Equation 2.7:** Activity (nmol  $g^{-1} h^{-1}$ ) =

<u>FA ×100 ml</u> ExC×(0.2ml)×Time (h)×Sediment (g)

Where

**Equation 2.8:** FA = SA - SuC - SaC

FA = final absorbance SA = sample assay SuC = substrate control SaC = sample control ExC = extinction coefficient (nmol) Absorbance is the absorbance at 450 nm, 100 ml refers to buffer volume and 0.2 ml is the sample volume in the microplate.

## 2.14 Statistical Analysis of Results

One way ANOVA was conducted to test the effect of different treatments on total bromide incubation fluxes, porewater oxygen diffusive fluxes and porewater nutrient diffusive fluxes. When there were significant interactions, post hoc Tukey test was carried out (Field, 2009). Differences between treatments and day of pH and nutrient fluxes and differences in porosity, total organic carbon, total nitrogen, CN ratio, porewater nutrient concentrations, porewater bromide concentrations and enzyme activities between treatments and depths were tested using two-way ANOVA. Post hoc Bonferroni test was conducted to identify between which treatments, day and depth, significant differences occurred. Linear regressions and non-linear regressions between enzyme activities with porewater nutrients and enzyme activities and sediment physico-chemical characteristics were also assessed. The relationship between dissolved oxygen with nutrient concentrations and dissolved oxygen with nutrient fluxes in the overlying water were processed using principal component analysis (PCA). In the sediments, porosity, total organic carbon, total nitrogen, porewater ammonium concentrations, alkaline phosphatase, beta-glucosidase, cellobiohydrolase, chitinase and urease activities were also assessed using PCA. Statistical analyses were performed with software package SPSS 19.0 (SPSS Inc., Chicago, IL, USA) and GraphPad Prism 6.0 software (GraphPad Prism Software, San Diego, USA).

#### **CHAPTER 3 RESULTS**

#### **3.1** Introduction

This chapter presents all the data obtained in this study following the methodology described in Chapter 2. This chapter begins with a description of the physicochemical characteristics of the sampling area and the effect of different treatments to the physico-chemical characteristics. Section 3.3 describes the effect of different treatments to the bromide concentration in the overlying water and the porewater. Section 3.4 explains the effect of different treatments to the oxygen penetration depth and oxygen diffusive fluxes. Section 3.5 explains the initial nutrient and porewater nutrient concentrations of Breydon Water and the effect of different treatments to the water column nutrients and porewater nutrients concentrations (nitrate, nitrite, ammonium, and phosphate), followed by the nutrient fluxes and porewater diffusive fluxes. The final section explains the enzyme activities in the sediment (alkaline β-glucosidase, cellobiohydrolase, chitinase and urease). phosphatase, The interrelationships between the different parameters measured in this study were also presented in this chapter.

## 3.2 Physico-chemical Characteristics of Breydon Water, Great Yarmouth

The overlying water during the sampling carried out in June 2010 had a salinity of 32 %, a pH of 7.83  $\pm$  0.06 at a temperature of 12.4  $\pm$  0.02 °C and a dissolved oxygen saturation of 75 %. Table 3.1 summarises the sediment characteristics in Breydon Water, Great Yarmouth. The porosity of the sediment ranged from 0.54 to 0.77, decreasing with depth due to sediment compaction that occurred below 5 cm depth. The sediment was muddy with high total organic carbon (TOC) and total nitrogen (TN) at the surface, which decreased with depth, showing a pattern of organic matter mineralisation.

Depth (cm)	Porosity (g/mL)	TOC (%)	TN (%)	CN ratio	Particle size (%)		
					<8µm	8-64 µm	> 64 µm
1	$0.77\pm0.01$	$3.86\pm0.11$	$0.44\pm0.02$	$8.88 \pm 0.34$	$70.3\pm6.8$	$15.8\pm6.1$	$7.21\pm2.3$
3	$0.73\pm0.02$	$3.59\pm0.15$	$0.45\pm0.02$	$8.03\pm0.57$	$66.4\pm7.7$	$10.3\pm4.6$	$5.43 \pm 1.8$
5	$0.70\pm0.01$	$3.27\pm0.09$	$0.35\pm0.01$	$9.27\pm0.28$	$65.7\pm6.0$	$11.7\pm5.4$	$8.90\pm3.7$
7	$0.58\pm0.02$	$2.83\pm0.07$	$0.34\pm0.01$	$8.26\pm0.44$	$63.4\pm8.4$	$9.3\pm2.6$	$6.62\pm0.7$
9	$0.54\pm0.03$	$2.21\pm0.08$	$0.29\pm0.03$	$8.08 \pm 0.97$	$71.4\pm5.0$	$12.1\pm3.3$	$6.84\pm3.4$

Table 3.1 Sediment physico-chemical characteristics from samples taken at Breydon Water, Great Yarmouth sampled at 1 to 9 cm depth in June 2010, reported as mean  $\pm$  SE, n=5.

# 3.2.1 Effect of Different Treatments on Overlying Water pH during Laboratory Incubations

pH measures the active hydrogen ions concentrations; therefore data are presented in terms of their hydrogen ion concentrations in the overlying water. This data is measured to evaluate the effect of different treatments and days on pH as shown in Figure 3.1. Hydrogen ion concentrations are higher in the oxic treatment; indicating that the pH is lower in the oxic treatment relative to the anoxic treatment. Conversely, low concentrations of hydrogen ions in anoxic treatment; means that the water is more alkaline and has a higher pH than oxic. Statistically significant difference was found between treatments (two-way ANOVA, p<0.0001) but were not significant with days (two-way ANOVA, p>0.05).

# 3.2.2 Effect of Different Treatments on Porosity during Laboratory Incubations

Generally, all treatments showed higher porosity at the sediment surface and decreasing pattern with depths (two-way ANOVA, p< 0.0001, depth effect, Table 3.2). But, no significant differences were detected among treatments due to no changes in the sediment profiles (two-way ANOVA, p > 0.05, treatment effect). Similar with initial treatments (Table 3.1), the influence of depth is likely due to compaction, leaving denser solid particles below 5 cm depth in the sediments.


Figure 3.1 Daily hydrogen ion concentrations maintained under different oxygen concentrations in overlying water samples taken during the 7 day laboratory incubations. Samples are reported as mean  $\pm$  SE, n=5. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.

Table 3.2 Porosity (g/mL) measured at 1 to 9 cm depth sediments from five different treatments, oxic, hypoxic, 1DA, 4DA and anoxic taken on day 7 of laboratory incubations. Samples are reported as mean  $\pm$  SE, n=5.

Depth (cm)/ Treatments	1	3	5	7	9
Oxic	$0.78 \pm 0.11$	$0.76\pm0.02$	$0.72\pm0.01$	$0.60 \pm 0.03$	$0.52\pm0.05$
Hypoxic	$0.78 \pm 0.02$	$0.72\pm0.05$	$0.69\pm0.05$	$0.55\pm0.05$	$0.50\pm0.05$
1DA	$0.80\pm0.04$	$0.71 \pm 0.06$	$0.61\pm0.06$	$0.53\pm0.05$	$0.49\pm0.05$
4DA	$0.80\pm0.06$	$0.71\pm0.05$	$0.61\pm0.04$	$0.53\pm0.03$	$0.49\pm0.04$
Anoxic	$0.81 \pm 0.06$	$0.75\pm0.04$	$0.70\pm0.01$	$0.60\pm0.01$	$0.52\pm0.01$

Abbreviations, 1DA = 1 day anoxic, 4DA = 4 day anoxic

## 3.2.3 Effect of Different Treatments on Total Organic Carbon (TOC), Total Nitrogen (TN) and CN ratio Taken at the Start and on Day 7 of Laboratory Incubations

During the 7 days of the experiment, all treatments showed rapid reductions in sediment TOC and TN relative to initial conditions (Figure 3.2 a and b). On average, the reduction of TOC in the anoxic treatment is 0.1% and 1% in the oxic treatment relative to initial condition. The loss of TOC and TN affects the mol/mol CN ratio particularly in the oxic and 4DA treatments at 7 cm depths relative to other depths (Figure 3.2 c).



Figure 3.2 Vertical profiles of a) TOC, b) TN and c) CN ratio taken at the start and on day 7 of laboratory incubations. Samples are reported as mean  $\pm$  SE, n=5. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles- initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.

## 3.3 Effect of Different Treatments on Bromide Concentration and Total Bromide Incubation Fluxes in the Overlying Water and Porewater Bromide Concentrations and Diffusive Fluxes in the Sediments

At the start of the incubation with Br spiked into the overlying water on day 4, Br concentration in the overlying water was 16 287  $\pm$  10 µmol L<sup>-1</sup>. After 3 days of incubation until day 7, Br concentrations in the overlying water were 14 900  $\pm$  6 µmol L<sup>-1</sup> for oxic, 15 200  $\pm$  2 µmol L<sup>-1</sup> for 1DA, 15 900  $\pm$  2 µmol L<sup>-1</sup> for hypoxic, 16 400  $\pm$  4 µmol L<sup>-1</sup> for 4DA and 16 600  $\pm$  2 µmol L<sup>-1</sup> for anoxic treatments. The difference in Br concentrations in the overlying water after 3 days of incubation reflects differences between treatments in the extent of movement of Br into the sediments. There is also a smaller contribution caused by changes in the volume of overlying water due to evaporation (Quintana et al., 2007).

The background porewater Br concentration at the start of the incubation ranged from 700- 900  $\mu$ mol L<sup>-1</sup>. After 3 days of incubations, Br was transported from the overlying water to the sediments in all treatments; concentrations decreasing with increasing depth in the sediment (two-way ANOVA, p< 0.001, depth effect, Figure 3.3). The vertical profiles of bromide were significantly different among treatments (two-way ANOVA p< 0.001, treatment effect). The significant treatment and depth effect resulted from higher irrigation fluxes into the sediment (Table 3.3). Br penetrated up to 9 cm depth in the oxic, 1DA and hypoxic treatment, however Br concentrations were low from 5 cm depths downwards, relative to those in the oxic treatments. The 4DA treatment showed zero bromide transport beyond 7 cm. The anoxic treatment revealed a steep decrease, reaching the background level at 5 cm depth (Figure 3.3). Reduced bromide exchange in the hypoxic, 4DA and anoxic treatments were driven by the higher diffusive fluxes compared to irrigation fluxes (Table 3.3).

Total bromide incubation fluxes were calculated using equation 2.2 from the difference of Br concentration in the overlying water at the start of the incubation and during the 3 days of incubation. The calculation also includes the volume of water in the cores and the incubation time (Table 3.3). There were statistically

significant difference of total bromide incubation fluxes between treatments (oneway ANOVA, p<0.001). Total bromide incubation fluxes from the overlying water in the oxic and 1DA treatments were significantly higher than in the hypoxia, 4DA and anoxic treatments (Tukey, p<0.001). This results from higher irrigation fluxes in the oxic and 1DA treatments (Table 3.3). Total incubation fluxes in the hypoxic treatment were significantly lower than in the oxic and 1DA treatment (Tukey, p< 0.001), showing lower irrigation fluxes (Table 3.3).



Figure 3.3 Porewater bromide concentrations from five different treatments, oxic, hypoxic, 1DA, 4DA and anoxic taken at the start and after 3 days incubation with bromide spiked into the overlying water. Values are reported as mean  $\pm$  SE, n=5. No vertical bar indicates standard error is smaller than the symbol size. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.

Table 3.3 Mean total bromide incubation, diffusive and irrigation fluxes ( $\mu$ mol m<sup>-2</sup> h<sup>-1</sup>) after 3 days of incubation. Samples are reported as mean ± SE, n=5.

Treatments	Total incubation fluxes ( $\mu$ mol m <sup>-2</sup> h <sup>-1</sup> )	Diffusive $(\mu m old m^{-2} h^{-1})$	Irrigation (µmol m <sup>-2</sup> h <sup>-1</sup> )	Ratio Total/Diffusive
	4	4 /	4	
Oxic	$1475 \pm 5$	$537 \pm 25$	937	2.7
Hypoxic	$1307 \pm 28$	$889\pm41$	417	1.5
1DA	$1450 \pm 24$	$809 \pm 30$	611	1.8
4DA	$1335 \pm 30$	$1298\pm22$	90	1
Anoxic	$1388\pm24$	$1386 \pm 14$	2	1

Abbreviations, 1DA = 1 day anoxic, 4DA = 4 day anoxic

## 3.4 Effect of Different Treatments on Oxygen Penetration Depth and Oxygen Diffusive Fluxes Taken at the Start and on Day 7 of Laboratory Incubations

Vertical profiles of oxygen showed decreasing trends with depth as oxygen are consumed within the sediment (Figure 3.4). The oxygen profiles were collected beginning at 1 mm above the sediment-water interface. Oxygen concentrations declined steeply at 0.5 mm above the sediment-water interface in the oxic and 1DA treatments (Figure 3.4). This portion of the profile indicates the development of a slight benthic boundary layer due to the apparent vertical profile of the bottom waters accompanied by a linear decline of oxygen into the sediment.

Initially, oxygen penetrated up to 4 mm into the sediments (Figure 3.4). After being bubbled with air in the oxic treatment, oxygen penetrated up to 5.5 mm depth on day 7 with an oxygen concentration of 265  $\mu$ mol L<sup>-1</sup> in the overlying water (Figure 3.4 and Table 3.4). Diffusive oxygen fluxes were significantly different among treatments (one-way ANOVA, p<0.001). This difference is influenced by higher concentration gradient in the oxygenated treatments than in the low oxygen treatments. The highest diffusive oxygen fluxes were found in the oxic treatments (Table 3.4), followed by the 1DA treatments (6.87 ± 0.06 mmol m<sup>-2</sup> d<sup>-1</sup>). Low oxygen treatments in the hypoxic, 4DA and anoxic treatments showed limited oxygen penetration beyond depths of 2.5 mm and low diffusive fluxes (Figure 3.4 and Table 3.4). This pattern reflects the negative effects of anoxia to the sediment biogeochemistry. The oxygen penetration depth indicates the thickness of the oxic zone and is determined mainly by organic carbon degradation and the supply of oxygen from the bottom water (Cai and Sayles 1996; Kim and Kim 2007).



Figure 3.4 Oxygen penetration depths from five different treatments, oxic, hypoxic, 1DA, 4DA and anoxic taken at the start and on day 7 of laboratory incubations. Values are reported as mean  $\pm$  SE, n=5. No vertical bar indicates standard error is smaller than the symbol size. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.

Table 3.4 Diffusive oxygen fluxes (mmol  $m^{-2} d^{-1}$ ) in the sediment calculated on day 7 of laboratory incubations using Fick's first Law of diffusion. Samples are reported as mean  $\pm$  SE, n=5.

Treatments	DO in the overlying water (% saturation)	DO in the overlying water (µmol L <sup>-1</sup> )	Flux (mmol $m^{-2} d^{-1}$ )
Oxic	95	265	$7.90\pm0.04$
Hypoxic	25	74	$0.95\pm0.29$
1DA	93	276	$6.87\pm0.06$
4DA	22	44	$0.81\pm0.05$
Anoxic	0	0.6	$0.01\pm0.01$

Abbreviations, 1DA = 1 day anoxic, 4DA = 4 day anoxic

## 3.5 Initial Nutrient and Porewater Nutrient Concentrations of Breydon Water, Great Yarmouth

Table 3.5 shows the nutrient concentration in the overlying water and porewater nutrient concentrations of Breydon Water, Great Yarmouth. The nutrients in the overlying water were dominated by ammonium, followed by nitrate and to a lesser extent by nitrite and phosphate. Porewater nitrate displayed a gradual decrease with depth. There was no change in porewater nitrite and an increase of porewater ammonium with depth. The high concentrations of ammonium in the porewater compared to nitrate and nitrite could be explained by organic matter mineralisation process taking place. Higher nitrate concentrations in the overlying water compared to porewater, illustrated that porewater nitrate were either transformed to other dissolved inorganic nitrogen (DIN) or released to the overlying water. The constant nitrite concentrations with depth, suggests diffusion is taking place. Ammonium profiles displayed drastic reduction of ammonium at the sediment surface relative to that in the overlying water. This might show extensive exchange at the sedimentwater interface, influenced by organic matter degradation process. Similar with ammonium, porewater phosphate concentrations increased with depth, possibly as a result of the absence of oxygen in deeper sediments.

Table 3.5 Initial water column nutrients and porewater nutrient concentrations ( $\mu$ mol L<sup>-1</sup>) from samples taken at Breydon Water, Great Yarmouth. Samples are reported as mean  $\pm$  SE, n=5.

	Depth (cm)	Nitrate $(\mu mol L^{-1})$	Nitrite $(\mu mol L^{-1})$	Ammonium $(\mu mol L^{-1})$	Phosphate $(\mu mol L^{-1})$
Overlying water	-	$15.07 \pm 1.54$	$0.35 \pm 0.14$	$44.17\pm7.89$	$0.32 \pm 0.04$
Porewater	1	$0.57\pm0.09$	$0.48\pm0.01$	$2.02 \pm 1.64$	$0.43\pm0.02$
	3	$0.27\pm0.04$	$0.48\pm0.01$	$53.80 \pm 4.64$	$0.57\pm0.03$
	5	$0.07\pm0.02$	$0.48\pm0.02$	$482.57 \pm 17.15$	$0.72\pm0.03$
	7	$0.06\pm0.01$	$0.48\pm0.01$	$786.67 \pm 10.22$	$1.20\pm0.15$
	9	$0.05\pm0.01$	$0.49\pm0.01$	$812.16\pm3.85$	$2.41\pm0.46$

# 3.5.1 Effect of Different Treatments on Nutrients Concentrations and Measured Nutrient Fluxes to/from the Overlying Water During 7 days of Laboratory Incubations

This section begins with the description of different oxygen saturation supplied in the five treatments (Figure 3.5 a). This is followed by the nutrient concentrations measured in the overlying water (Figure 3.5 b-e). The measured nutrient fluxes (Figure 3.6) were calculated from the daily nutrient concentration changes in the overlying water in Figure 3.5. The calculation also includes the incubation time and the volume of overlying water in the core during incubations according to equation 2.2 under section 2.7.1.

The dissolved oxygen saturation in the overlying water of oxic, hypoxic and anoxic treatments was maintained at 95 %, 25 % and 0 % respectively throughout the 7 days experiment (Figure 3.5 a). Recovery processes from anoxia were assessed in two other treatments, with varying length of exposure to anoxia, 1 day and 4 days (1DA and 4DA, respectively). In the 1DA treatments, re-aeration was carried out for 6 days and 3 days in the 4DA treatment. DO % was maintained at 0 % on day 1 and increased to 93 % after re-aeration was supplied in the 1DA treatment. In contrast, DO saturation only reached up to 22 % in the overlying water at the end of the experiment in the 4DA treatments (Figure 3.5 a). That indicates that there was a very substantial oxygen demand after the period of anoxia in the 4DA treatments.

Nitrate concentrations were high initially in all 5 treatments and started to decrease with low oxygen conditions (in the hypoxic, 4DA and anoxic treatments in Figure 3.5 b). A minor shift of increasing nitrate was observed in the 1DA treatment after re-aeration was introduced to the treatments on day 3 and day 4 (Figure 3.5 b).

On the other hand, at the start of the incubation, nitrite concentrations in the water column of all treatments were low and increased in the oxic and 1DA treatments (Figure 3.5 c). Nitrite concentration also increased in the hypoxic treatment after day 3. Only a slight increased of nitrite concentration was observed in

the 4DA treatments after day 5. Nitrite remains low in the anoxic treatment (Figure 3.5 c).

Ammonium concentrations increased at first in all treatment and decreases in the oxic treatment (Figure 3.5 d). In the oxic treatment, decreasing ammonium concentration was measured simultaneous with increasing water column nitrite (Figure 3.5 c and d). The pattern suggests ammonium oxidation to nitrite in the overlying water in the oxic treatment.

In the 1DA treatment, ammonium concentration was lower than in the hypoxic treatment (Figure 3.5 d). This pattern implies the effect of re-aeration in decreasing the ammonium concentrations.

In the hypoxic treatment, ammonium concentration increased and nitrate reduced (Figure 3.5 b and d). The pattern inferred from the hypoxic treatments, reflects nitrate reduction to ammonium.

Substantial increased of ammonium concentration was measured in the 4DA and anoxic treatments (Figure 3.5 d). At the same time, very low nitrate and nitrite concentrations in the 4DA and anoxic treatment suggest a net source of ammonium in these two treatments.

The concentration of phosphate was very low in the oxic treatment, but continues to increase in the hypoxic and anoxic treatment (Figure 3.5 e). The low phosphate concentration in the oxic treatment might reflect phosphate binding with iron oxides. Increasing phosphate concentrations in the hypoxic and anoxic treatment shows desorption from iron oxides, releasing phosphate to the water column. Phosphate concentration remains low in the 1DA treatments and decrease from day 4 onwards. In the 4DA treatments, phosphate concentration increased until day 4 and decreased from day 5 onwards. The pattern in the 1DA and 4DA treatment shows that the effect of re-aeration was probably sufficient to capture the released phosphate back to the sediment surface.



Figure 3.5 The daily evolution of a) oxygen saturation, b) nitrate, c) nitrite, d) ammonium and e) phosphate concentrations exposed to different oxygen treatments during the 7 day laboratory incubations. Values are reported as mean  $\pm$  SE, n=5. No vertical bar indicates standard error is smaller than the symbol size. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.

Daily nutrient fluxes in Figure 3.6 a-d were calculated from the water column nutrient concentration changes in Figure 3.5 b-e. Generally, the magnitude and/or direction of all measured nutrient fluxes in Figure 3.6 a-d differed significantly with day and treatment (two-way ANOVA, p<0.0001).

On day 2, nitrate fluxes (Figure 3.6 a) were significantly lower in the 4DA and anoxic treatments compared to the oxic treatments (Bonferroni corrected, p<0.0001). On day 7, there was a significant nitrate flux into the overlying water in the 4DA treatments in Figure 3.6 a (Bonferroni corrected, p<0.01).

On day 2, nitrite flux was significantly lower in the 1DA treatment compared to the oxic treatment in Figure 3.6 b (Bonferroni corrected, p < 0.05). On days 4 and 6, Figure 3.6 b shows that nitrite fluxes were significantly lower in all other treatments than in the oxic treatments (Bonferroni corrected, p = 0.0001-0.01).

Figure 3.6 c shows that substantially higher ammonium fluxes out to the overlying water in the 4DA and anoxic treatments from day 3 onwards (Bonferroni corrected, p<0.01). This pattern demonstrates the dominance of ammonium, consistent with very low nitrate and nitrite fluxes in low oxygen conditions (Figure 3.6 a-c). High release of ammonium in the overlying water might be due to decreases in the sediment demand for ammonium, i.e for nitrification and assimilation.

But, on day 7, ammonium flux was into the sediments in the 4DA treatment in Figure 3.6 c (Bonferroni corrected, p<0.0001). Nitrate flux was simultaneously observed into the overlying water on day 7 in the 4DA treatments (Figure 3.6 a). This shows ammonium conversion to nitrate by the end of the experiment.

On the first day, phosphate flux was to the overlying water in all except the oxic treatments (Figure 3.6 d). Phosphate flux to the overlying water is greatest on day 2 in the 4DA treatments (Figure 3.6 d), potentially as a response to drastic oxygen fluctuations. Phosphate flux into the overlying water continues the following day until day 7 in the anoxic treatment. This flux presumably results from the reduction of iron (III) oxides.



Figure 3.6 Daily a) nitrate, b) nitrite, c) ammonium and d) phosphate fluxes calculated from the water column nutrient concentration changes in Figure 3.6 b-e from five different treatments; oxic, hypoxic, 1DA, 4DA and anoxic. Values are reported as mean  $\pm$  SE, n=5. Red bar- anoxic treatments, yellow bar- four days without oxygen and reaerated for 3 days (4DA), grey bar- hypoxic conditions, purple bar- 1 day without oxygen and re-aerated for 6 days (1DA), green bar- oxic treatments.

69

## 3.5.2 Interrelationships between Nutrient Concentrations with Dissolved Oxygen; and Nutrient Fluxes with Dissolved Oxygen in the Overlying Water during 7 days of Laboratory Incubations

Principal Component Analysis (PCA) was used to summarise the data on nutrient concentration and dissolved oxygen in the overlying water (Figure 3.7 a - b). Two components accounted for 88% of the overall variability of the data. The first component, which accounted for 66% of the variability of the data, was associated with nitrate, ammonium, phosphate and dissolved oxygen (Table 3.6). Component 1 contrasted sample with increasing nitrate concentration and dissolved oxygen and decreasing ammonium and phosphate concentrations. The water chemistry in each treatment evolves along its own trajectory (Figure 3.7 a).

PCA was also use to show the evolution of nutrient fluxes in response to changing dissolved oxygen in the overlying water during the 7 days laboratory incubations (Figure 3.8 a - b). Component 1 was positively correlated with dissolved oxygen and negatively correlated with ammonium and phosphate fluxes whereas nitrate and nitrite fluxes was positively correlated with both components in Table 3.7 (component 1 44.17%, component 2 18.8%).

Nitrate concentrations are high when dissolved oxygen saturations are high and started decreasing when dissolved oxygen saturations were low (Figure 3.7 a). The largest nitrate uptake into the sediment occurred when the dissolved oxygen was low as a response to oxygen depletion during the experiment (Figure 3.8 a). A scatterplot between nitrate concentration and nitrate fluxes also shows high concentrations and fluxes in the presence of oxygen (Figure 3.9 a). In the hypoxic treatments, nitrate is partly reduced to nitrite with most of the DIN converted to ammonium (Figure 3.9 a, c and e).

Nitrite concentration accounted for the second component (22% of the variability of the data) in Table 3.6. Nitrite concentrations were higher in the presence of oxygen and low in anoxic conditions (Figure 3.7 a). The highest nitrite flux into the overlying water was at high dissolved oxygen saturation (Figure 3.8 a).

This is consistent with the scatterplots of nitrite concentrations and nitrite fluxes in Figure 3.9 b. Nitrite accumulates in the oxic treatment but in the anoxic treatment, this is reduced further to ammonium (Figure 3.9 c and e).

Ammonium is high when the dissolved oxygen is low (Figure 3.7 a). Ammonium fluxes out into the overlying water when oxygen was low (Figure 3.8 a). This indicates ammonium accumulation in the absence of oxygen. Ammonium is inevitably produced from both nitrate and nitrite reduction in the hypoxic, 4DA and anoxic treatments (Figure 3.9 a, c and e).

In the presence of oxygen, phosphate remains relatively low. Hypoxia and anoxia, in turn increased phosphate concentrations in the overlying water (Figure 3.7 a). The greatest phosphate flux into the overlying water occurred when oxygen was low, potentially because of iron (III) reduction to iron (II) (Figure 3.8 a – b). Reaeration was sufficient to maintain phosphate at low concentrations in the 1DA treatment and led to decreased phosphate concentrations in the 4DA treatment (Figure 3.9 g – h). Overall, principal component analysis revealed a distinct pattern of evolution of nutrients in the water in response to different treatments during the 7 days of incubation.



Figure 3.7 Scatterplots of the principal component analysis of dissolved oxygen, nitrate, nitrite, ammonium and phosphate concentrations in the overlying water according to different a) treatments and b) day. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments. Black square – day 1, green square – day 2, orange square – day 3, blue square – day 4, pink square – day 5, red square-day 6, grey square-day 7.

Table 3.6 Eigenvectors extracted based on values greater than 0.6 and above for component 1 and 2 of Principal Components Analysis

Variables	Component		
variables	1	2	
Nitrate concentration	.905	241	
Nitrite concentration	.368	.906	
Ammonium concentration	902	.198	
Phosphate concentration	930	.155	
Dissolved Oxygen	.845	.245	



Figure 3.8 Scatterplots of the principal component analysis of dissolved oxygen, nitrate, nitrite, ammonium and phosphate fluxes in the overlying water according to different a) treatments and b) day. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments. Black square – day 1, green square – day 2, orange square – day 3, blue square – day 4, pink square – day 5, red square-day 6, grey square-day 7.

Variables	Component		
variables	1	2	
Dissolved Oxygen	.794		
Ammonium fluxes	754		
Phosphate fluxes	667		
Nitrate fluxes	.543	.625	
Nitrite fluxes	.520	.534	

Table 3.7 Factors extracted using Principal Component Analysis by fixing the number of factors to extract and suppressing factor loadings of less than 0.4.





Figure 3.9 Relationships of nutrient concentrations to nutrient fluxes according to treatment (a, c, e and g) and day (b, d, f and h). Samples are reported as mean  $\pm$  SE, n=5. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments. Black square – day 1, green square – day 2, orange square – day 3, blue square – day 4, pink square – day 5, red square-day 6, grey square-day 7.

## 3.5.3 Effect of Different Treatments on Porewater Nutrient Concentrations in the Sediment During 7 days of Laboratory Incubations

In general, the concentrations of porewater nitrate and nitrite were variable with depth, but distinct increasing patterns were observed in porewater ammonium and phosphate concentrations with depth (two-way ANOVA, p < 0.0001, depth effect in Figure 3.10 a - d).

A significant difference in the concentration of porewater nitrate was measured between the treatments (two-way ANOVA, p < 0.0001). There was a significant interaction between treatments and depth effects on porewater nitrate concentrations (two-way ANOVA, p < 0.0001). The significant interaction shows that the relationship with depth varied between treatments. Porewater nitrate concentration peaked at 1 cm depth in the oxic and initial treatments and decreased with depth (Figure 3.10 a). Porewater nitrate concentrations decreased beyond 3 cm depth in the anoxic treatment. But, there was little change with depth in the hypoxic, 1DA and 4DA treatments. Porewater nitrate concentration is lower in the 4DA and anoxic treatments throughout the sediment depth than in the oxic, 1DA and hypoxic treatment (Figure 3.10 a). This pattern results from conversion of porewater nitrate to ammonium in the anoxic treatment (Figure 3.10 a and c). However, it is unclear whether porewater nitrate in the 4DA treatment is converted to porewater ammonium.

There were significant effects of treatment and depth on porewater nitrite concentrations (two-way ANOVA, p < 0.001, Figure 3.10 b). Porewater nitrite concentrations did not show major changes with depth but increasing concentration was observed at 5 cm depth in the 4DA treatments (Figure 3.10 b). Porewater nitrite concentrations decreased at 9 cm depth in the 1DA, hypoxic and 4DA treatments. Porewater nitrite concentrations also decreased beyond 5 cm depth in the anoxic treatments (Figure 3.10 b). The patterns observed in the hypoxic, 1DA, 4DA and anoxic treatments at 9 cm depth; suggest nitrite reduction to ammonium in the porewater of these treatments.

Porewater ammonium concentrations significantly increased from initial values, particularly below 3 cm depth in all treatments. The significant increase shows extensive ammonium accumulation due to organic matter mineralisation.

Porewater phosphate concentrations increased extensively below 5 cm depth in the hypoxic, 4DA and anoxic treatments. Porewater phosphate concentrations remained fairly constant in the oxic and 1DA treatments (Figure 3.10 d). The pattern may be explained by the release of phosphate that is sorbed with iron (III) in low oxygen.



Figure 3.10 Porewater a) nitrate, b) nitrite, c) ammonium and d) phosphate concentrations taken at the start and on day 7 of laboratory incubations from five different treatments, oxic, hypoxic, 1DA, 4DA and anoxic . Values are reported as mean  $\pm$  SE, n=5. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.

# 3.5.4 Effect of Different Treatments on Nutrient Concentrations and Porewater Nutrient Concentrations During 7 days of Laboratory Incubations

The water column nitrate concentrations in the oxic, hypoxic, 1DA, 4DA and anoxic treatment are similar (Figure 3.11 a). But, porewater nitrate concentration is the highest in the oxic treatment followed by the 1DA, hypoxic, 4DA and anoxic treatments (Figure 3.11 a). The highest porewater nitrate concentration in the oxic treatment might imply nitrification whilst low porewater nitrate concentration in the anoxic treatment might suggest conversion into ammonium (Figure 3.11 c).

Low nitrite concentrations occur in both the overlying water and porewater. Porewater nitrite is the highest in the oxic and 1DA treatments relative to those in the anoxic treatment. This might suggest porewater nitrite conversion to porewater ammonium in the 4DA and anoxic treatment (Figure 3.11 b).

Ammonium has the highest concentration in comparison to the other nutrients in both the overlying water and porewater (Figure 3.11 c). Ammonium is highest in the porewater of the 4DA and anoxic treatments due to organic matter mineralisation. Low ammonium concentrations in the oxic and 1DA treatment may imply nitrification.

Low phosphate concentrations in both the overlying water and porewater in the oxic and 1DA treatment is due to the buffering mechanism of iron oxides (Figure 3.11 d). Higher porewater phosphate concentrations in the hypoxic, 4DA and anoxic treatment relative to those in the oxic and 1DA treatment might be due to the reduction mechanism of insoluble  $FePO_3^-$  into soluble  $Fe_3(PO_4)^2$  (Figure 3.11 d).



Figure 3.11 Relationship of a) nitrate and porewater nitrate, b) nitrite and porewater nitrite, c) ammonium and porewater ammonium and d) phosphate and porewater phosphate according to treatments. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.

# 3.5.5 Effect of Different Treatments on Porewater Diffusive Fluxes in the Sediment During 7 Days of Laboratory Incubations

The porewater diffusive fluxes were calculated from the porewater concentration gradients in Figure 3.10 a –d. The calculated diffusive fluxes are all very low (Figure 3.12 a - h), because of the low porewater concentrations in the sediments (Figure 3.10 a - d). An exception was for ammonium, which has high porewater concentrations (Figure 3.10 c) and not surprisingly had the highest diffusive fluxes (Figure 3.12 e). Overall, the data variability in the diffusive fluxes as well as the non-significant diffusive fluxes limits the strength of the conclusion that can be derived from this result.

Porewater nitrate diffusive fluxes were directed into the overlying water in all except the anoxic treatments (one-way ANOVA, p < 0.001, Figure 3.12 a). Porewater nitrate diffusive fluxes were significantly lower in the hypoxic, 4DA and anoxic treatments than in the oxic and initial treatment (Tukey p< 0.001). Lower porewater nitrate diffusive fluxes in the hypoxic, 4DA and anoxic treatments could indicate decrease nitrification. This is consistent with the conversion of nitrate to ammonium from the fluxes inferred from concentration changes in the overlying water of these treatments (Figure 3.6 a – c).

Porewater nitrite, ammonium and phosphate diffusive fluxes show fluxes directed to the sediments in all treatments in Figure 3.12 c, e and g (one-way ANOVA, p<0.001). No significant differences were observed for porewater nitrite and ammonium diffusive fluxes between the different treatments (Tukey, p > 0.05; Figure 3.12 c and e).

Phosphate diffusive fluxes were significantly lower in the oxic and 1DA treatments and higher in the 4DA and anoxic treatments compared to initial. Phosphate diffusive flux directed into the sediment could be the consequence of adsorption on iron (III) oxyhydroxides, which might explain the pattern in the oxic and 1DA treatments (Figure 3.12 g). Phosphate diffuses into the sediment in the 4DA treatment due to re-aeration. Re-aeration may cause phosphate binding to iron (III)

oxyhydroxides. Box plots of the diffusive fluxes (Figure 3.12 b, d, f and h) summarise the variability across the dataset.





📟 Initial 🚥 Oxic 🚍 Hypoxic 🛄 1DA 🜌 4DA 🖾 Anoxic

Figure 3.12 Porewater diffusive fluxes of a) nitrate, c) nitrite, e) ammonium and g) phosphate calculated from concentration gradients in the cores at the start and on day 7 of laboratory incubations from 5 different treatments, oxic, hypoxic, 1DA, 4DA and anoxic, calculated from porewater concentration gradients. b, d, f and h) Box whisker plot; central line in each box indicates median, ends of box  $25^{th}$  and  $75^{th}$  percentiles, error bars and points. Values are reported as mean  $\pm$  SE, n=5.

## 3.5.6 Comparison between Measured Nutrient Fluxes and Calculated Diffusive Fluxes

In general, DIN fluxes were dominated by ammonium fluxes in the overlying water and ammonium diffusive fluxes in the sediments than other nitrogen species (Figure 3.6 c and 3.12 e). There was a small contribution of nitrate and nitrite diffusive fluxes to the overall DIN flux. Similarly, small fluxes were measured for phosphate in the overlying water and the porewater (Figure 3.6 d and 3.12 g).

Apparent disagreement was observed between the measured nutrient fluxes in the overlying water and the calculated porewater diffusive fluxes (Figure 3.6 a-d and 3.12 a-g). The pattern and direction of the fluxes were variable, making it difficult to interpret. The inconsistency might be influenced by biological and chemical effects dominating the treatments. The diffusive fluxes showed variable patterns, suggesting that surface processes occurring at the sediment-water interface may determine the extent and direction of the fluxes.

## 3.6 Enzyme Activities in the Sediment Taken at the Start and on Day 7 of Laboratory Incubations

All enzyme activities decreased significantly with depth and were influenced by treatments (two-way ANOVA, p < 0.001, for both depth and treatment effect, Figure 3.13 a-e). Significant interactions between treatment and depth were found for AP, CB and chitinase (two-way ANOVA, p < 0.001). This results from higher enzyme activities at the surface of oxic and 1DA treatments relative to those at depths compared with the initial pattern (e.g. AP, CB and chitinase activities). As for BG, activities at the sediment surface and 3 cm depths in the 4DA and anoxic treatment were significantly lower than those measured at the beginning of the incubation. In contrast, urease activities in the 4DA and anoxic treatments were significantly higher than at the start of the incubation at all depths except at 9 cm depths (Bonferroni corrected, p < 0.001).

AP was strongly correlated with porewater phosphate concentrations (Figure 3.14 a - b). This relationship might be due to a response to porewater phosphate concentrations or to overall bacterial activity, which in turn is controlled by the availability of oxygen. Anoxia elevates porewater phosphate concentrations as observed in the deepest part of the anoxic treatment. AP is also low near the surface of the anoxic treatment and decreased with depth. Hence, it is more likely that this significant inverse relationship is due to a response to porewater phosphate concentrations as oxygen concentration is low throughout the anoxic treatment.

BG, CB and chitinase activities were consistently high on the surface and decreased with depth (Figure 3.13 b - d). This pattern indicates the presence of labile organic matter on the surface, which supports high microbial activity. Microbial activity reduces with depth as the more labile components were broken down.

Urease activities were inversely correlated with porewater ammonium concentration (Figure 3.14 c-d). Urease activities explained 38% of the variance of porewater ammonium concentrations when fitting the relationship using simple linear regression. Figure 3.14 c-d shows that high porewater ammonium concentrations were associated with low urease activities particularly at the deepest part of the sediment. This relationship suggests that urease might be partly inhibited by porewater ammonium concentrations. Urease activities were also positively correlated with TOC as shown in Figure 3.14 e-f. This relationship suggests that bacteria with urease enzymes are enhanced by TOC. On the other hand, high urease activities at the sediment surface of anoxic treatments might be due to the death of aerobic benthic organism. The availability of organic matter is enhanced with the death of these organisms as a food source for anaerobic bacteria with urease activities. Potentially, the response of urease is associated with the influence of both porewater ammonium concentrations and TOC in the sediment.



Figure 3.13 Vertical profiles of potential enzyme activities of a) alkaline phosphatase, b) cellobiohydrolase, c) beta-glucosidase, d) chitinase and e) urease activities in the sediments exposed to different oxygen concentrations taken at the start and on day 7 of laboratory incubations, reported as mean  $\pm$ SE, n=5. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments.



Figure 3.14 Scatterplots of AP and porewater phosphate concentrations according to different a) treatments and b) depths, urease and porewater ammonium concentrations according to different c) treatments and d) depths, and urease with TOC according to different e) treatments and f) depths, reported as mean  $\pm$ SE, n=5. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments. Black triangles - 1 cm depth, green triangles - 3 cm depth, orange triangles - 5 cm depth, blue triangles - 7 cm depth, pink triangles - 9 cm depth.

### 3.6.1 Interrelationships between Parameters in the Sediment

PCA was used to investigate the relationship between porosity, total organic carbon (TOC), total nitrogen (TN), porewater ammonium concentrations, alkaline phosphatase activities (AP), cellobiohydrolase activities (CB), beta-glucosidase activities (BG), chitinase and urease activities measured in the sediments. The first two components explained 86.8% (component 1 70.2%, component 2 16.6%) of the variability observed in the sediments data. Table 3.8 shows the component matrices of factor loadings for each of the original variables.

Factor 1 was strongly correlated with increasing CB, BG, chitinase, AP, porosity and urease; and decreasing porewater ammonium concentration. TOC and TN were positively correlated with both factor 1 and 2. Figure 3.15 shows a plot of factor 1 against factor 2 with data points identified by treatments and by sediment depths.

In general, both plots show the sequence of each treatments from anoxic: 4DA: initials: hypoxic: 1DA and oxic (Figure 3.15 a). The anoxic and 4DA treatments are grouped together and surprisingly initial treatments followed closely to the 4DA treatments. Oxic treatments are clustered together with 1DA treatments, implying the effect of 1 day without oxygen to the physico-chemical characteristics, porewater ammonium concentrations and enzyme activities closely resemble those in oxygenated conditions. Enzyme activities also clearly varied with depths (Figure 3.15 b).

AP, BG, CB and chitinase activities were highly correlated with one another and their activities increased with increasing oxygen concentrations (for example in Figure 3.16 a, c and e). But, urease activities showed a weaker correlation with other enzymes and inversely related to oxygen concentrations (e.g. figure 3.16 e). The oxic treatment leads to increased AP, BG and chitinase activity relative to initial values (e.g. Figure 3.16 a and c). In contrast, urease activities shifted to higher activities in the anoxic treatments relative to initial values (e.g. Figure 3.16 e). The consistent pattern for AP, BG, CB and chitinase is a shift of enzyme activities with organic matter from the surface to increasing depth. AP, CB and chitinase activities were high at the sediment surface and generally decreased with depth, suggesting a decrease in substrate quality with depth (e.g. Figure 3.16 a-h). CB activities were weakly related to the TOC (Figure 3.16 g-h). Chitinase activities were inversely related to porewater ammonium concentrations, suggesting its function in the nitrogen cycle. Porosity decreases with depth because of sediment compaction and is correlated with other parameters largely due to the influence of depth.

Variable -		Component	
		1	2
Cellobiohydrolase		.917	
Beta-glucosidase		.895	
Chitinase		.888	
Alkaline phosphatase		.872	420
Porewater ammonium		865	
concentrations			
Porosity		.862	
Urease		.833	
Total Nitrogen		.689	.627
Total Organic Carbon		.683	.651

Table 3.8 Component matrices for principal component analysis of sediments in the study by extracting eigenvalues of greater than 0.6 and above and suppressing factor loadings of less than 0.4.



Figure 3.15 Scatterplots of the a) principal component analysis of porosity, total organic carbon, total nitrogen, porewater ammonium concentrations, alkaline phosphatase activities, cellobiohydrolase activities, beta-glucosidase activities, chitinase activities and urease activities identified by a) 5 different treatments and b) sediment depths measured. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles - initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments. Black triangles - 1 cm depth, green triangles - 3 cm depth, orange triangles - 5 cm depth, blue triangles - 7 cm depth, pink triangles - 9 cm depth.





Figure 3.16 Relationship between chitinase and AP according to a) treatments and b) depth, between chitinase and BG according to c) treatments and d) depth, BG and urease according to e) treatments and f) depth, CB and TOC according to g) treatments and h) depth , chitinase and porewater ammonium concentrations according to i) treatments and j) depth. Red circles - anoxic treatments, yellow circles - four days without oxygen and re-aerated for 3 days (4DA), blue circles initial treatments, grey circles - hypoxic conditions, purple circles - 1 day without oxygen and re-aerated for 6 days (1DA), green circles - oxic treatments. Black triangles - 1 cm depth, green triangles - 3 cm depth, orange triangles - 5 cm depth, blue triangles - 7 cm depth, pink triangles - 9 cm depth.

#### **CHAPTER 4 DISCUSSION**

### 4.1 Introduction

The aim of Chapter 4 is to discuss the results and interpretation of the data presented in Chapter 3. Integration of all parameters measured was done to investigate the underlying mechanisms involved in each treatment. This chapter begins with the significance and limitations of the methodology selected in Chapter 2 and continue with a general assessment of the sampling area. The discussion is then narrowed down to processes responsible for the patterns inferred from the nutrients and enzyme activities in each treatment.

### 4.2 Methodological Considerations

Before embarking on a thorough discussion of the data, it is worth discussing the significance and the limitations of the experimental technique selected. Laboratory incubation of sediment cores rely on careful sediment extraction and transport to laboratory (Mortimer et al., 1998). Cores are overlain with a specific volume of water and samples are taken at set intervals during the incubation period (Tengberg et al., 2003; Jenkins, 2005). Laboratory incubations of sediment cores was selected in this study as cores need to be aerated and bubbled with nitrogen to induce different levels of anoxia and hypoxia. Removal of the cores from the estuary does lead to some disturbance of the cores, but for muddy sediments which only have to be transported a short distance to the laboratory, disturbance is small.

Enhanced solute transport in excess of molecular diffusion is commonly calculated using a 1-dimensional and non-steady state mass conservation equation (Forster et al., 1995; Berg et al., 2001). Solute transport models are often used to simulate the exchange of solutes between sediment and water and solute transport within sediment inhabited by benthic macroorganisms (Matisoff and Wang, 1998). The generally accepted models are the enhanced diffusion model, the cylindrical burrow irrigation model and the nonlocal exchange model (Martin and Banta, 1992; Berg et al., 2001; Meile et al., 2005). But, the limitation of our data prevented calculation of enhanced diffusion coefficient irrigation models, which requires

information about the burrows (Meile et al., 2005). Therefore, we opted to use the difference between measured fluxes and diffusive fluxes as the irrigation fluxes and to compare the differences between oxic and oxygen depleted treatments (section 2.7.1).

Using tracers that occur in the field can be a useful technique, as it follows natural sediments and benthic communities to be studied undisturbed. This in turn may provide a more realistic picture of the processes occurring. Some previous studies have used naturally occurring isotopes, <sup>234</sup>Th, <sup>7</sup>Be, and <sup>210</sup>Pb to study downward mixing (Berg et al., 2001; Shull and Mayer, 2002). But, this is done to assess particle mixing in the sediment. In situ measurements are again preferable compared to laboratory incubations (Forster et al., 1999). But, these require specific and complex instrumentation and the variability between replicates can cause problems of interpretation (Bradshaw et al., 2006). Another problem is that the short half lives of some of these isotopes requires measurement to be made immediately, which was not possible in this study. Moreover, our focus on the experimental manipulation of the effects of different concentrations of oxygen would be very difficult to carry out in the field. Adding bromide to the system being studied creates an artificial set-up in which conditions are different to those found in the field, but this allowed full control of the experimental design and the use of multiple replicates and controls.

For the estimation of oxygen diffusive fluxes from the concentration gradient (section 2.8.1), the linear concentration change in the diffusive boundary layer (DBL) is commonly used because there is no need to have data on sediment porosity (Tengberg et al., 2004). However, the DBL procedure has experimental uncertainties due to the difficulty of determining the exact position of the sediment-water interface which is not always obvious. In this study, the concentration gradient was taken from the upper millimetre of the sediment and the porosity data extrapolated using polynomial curve fitting to get porosity measurement in millimetre scale (section 2.8.1).
Measurement of nutrient fluxes from sediments to the overlying water can be carried out either by *in-situ* benthic chambers, concentration gradients in sediment pore waters or through laboratory incubations of sediment cores (Hammond et al., 2004; Jenkins, 2005; Macreadie et al., 2006). However, it should be noted that any measurement of fluxes will only give the estimation of the ability of the sediment to release or take up the nutrients as they require a containment of water overlying the sediments. This therefore does not mimic the dynamic continuous physical movement of water over the mudflats, with periods of drying out and inundation of water. The chemical changes in water overlying the sediments may also alter fluxes. Despite these limitations, measurement of nutrient fluxes is useful as a basis to understand nutrient biogeochemistry. Moreover, the vast majority of studies of fluxes in the literature are of this type (e.g. Liu et al., 2003; Rasheed et al., 2006; Belias et al., 2007) facilitating comparison of the data produced in this study.

Measurements of *in-situ* benthic fluxes involve placing enclosed sediment corers on the surface of the sediments at the sampling area containing both sediment and water (Haag et al., 2005; Wild et al., 2005). The variations in nutrient concentrations in the overlying water within the chamber are measured and used to infer nutrient flux. Benthic chambers are considered the best method for estimating nutrient flux since they involve few assumptions and minimal perturbation to the sediment-water interface. However, the downside of this method is that they are time consuming and expensive to construct. They also require logistically elaborate deployments with significant occupational hazard (Jenkins, 2005).

Nutrient concentration gradients across the sediment-water interface can be obtained by sampling sediment pore waters at discrete depths (Denis and Grenz, 2003). This concentration-depth data may then be used to calculate nutrient diffusive fluxes on the basis of Fick's Law of Diffusion (Lavery et al., 2001; Liu et al., 2003). Fick's First Law is used to model sediment-water nutrient flux, assuming that diffusion is the dominant transport mechanism and to predict those fluxes (Liu et al., 2003). According to this law, flux is proportional to the slope of the concentration gradient (Jenkins, 2005). In this study, the nutrient fluxes in the overlying water and porewater diffusive fluxes in the sediment were measured (section 2.9.4 and 2.10.1)

as comparisons between the two can provide information about the relation between fluxes at the sediment-water interface and nutrient cycling within the sediment column (Clavero et al., 2000).

Microplate fluorimetric and colorimetric methods were used for measurement of enzyme activities (section 2.13.1 and 2.13.2) because they allow large number of sediment samples and enzymes to be analysed within a short time (Wirth and Wolf, 1992; Marx et al., 2001). This microplate method also enables well-known colorimetric assays to be scaled down for high-throughput analyses (Allison et al., 2007). It also allows reaction product formation to be analysed straight away in the microplate using fluorometrically labelled substrates, without the need to extract and purify the product through centrifugation and filtration prior to analysis (Saiya-Cork et al., 2002).

However, a limitation in the use of substrate proxies lies in the competition for available enzyme between the MUB substrates and any naturally occurring substrates (Liu et al., 2008). In this study, the problem was minimised by conducting assays at saturation concentrations of the fluorometric substrate (section 2.11.2). Secondly, MUB substrates are dimeric, rather than polymeric, so the extent to which their enzyme hydrolysis is representative of that of natural biopolymers is uncertain. Nevertheless, this form of enzyme activity measurement is very valuable and has generated major advances in the understanding of the functioning of enzymes in various environments (German et al., 2011; De et al., 2012) and facilitates comparison of data across environments (Arnosti, 2011).

## 4.3 Sediment Physico-chemical Characteristics, Nutrients and Enzyme Activities of Breydon Water, Great Yarmouth

The sampling site of Breydon Water is adjacent to the North Sea (Figure 2.3), hence the saline water. The pH readings of the overlying water in Breydon Water (section 3.2) fall within the typical range for estuarine systems of 7.6 - 8.5 (Olausson and Cato, 1980; Hinga, 2002). The porosity of the sediment decreases with depth (Table 3.1) because of compaction and precipitation of diagenetic materials (Gluyas and Coleman, 1992). Compaction happens when water is squeezed out, leaving denser solid particles within the sediments (Zonneveld et al., 2010). Initial carbon and nitrogen content in the sediment (Table 3.1) were high as it is typical for estuarine sediment (Hansen and Kristensen, 1998). TOC content in estuaries and productive ecosystems are in the range of 1 to 4% DW in areas where there are high organic matters content in sediments (Bianchi, 2007). High accumulation of organic matter on the surface of the sediment indicated high rates of mineralisation on the surface of fine grained sediment with less penetration deeper into the sediment except that due to the presence of invertebrates (Soetart et al., 1997). The CN ratio of 8.0 to 9.3 (Table 3.1) at the start of the incubations implies a mixture of both terrigenous and marine organic matter (Rasheed, 2004). An increase of CN ratio with depth down to 5 cm (Table 3.1) reflects the faster mineralisation of nitrogen than carbon (Kristensen and Blackburn, 1987; Meyers, 1994). This is also often linked to preferential loss of marine relative to terrigenous organic matter, which is often low in nitrogen (Hedges and Keil, 1995). The oxygen penetration depth of 4 mm in this study (Figure 3.4) is typical in coastal sediment compared to the centimetre or decimetre oxygen penetration depth in deep ocean sediments (Rasmussen and Jorgensen, 1992; Glud et al., 1994; Cai and Sayles, 1996; Mouret et al., 2010).

Nitrate concentrations are higher in the overlying water than the porewater (Table 3.5). This pattern illustrates that most of the regenerated nitrate was not consumed at the sediment-water interface, but was released by nitrification (Liu et al., 2003). Nitrite concentration was low in both the overlying water and porewater, suggesting conversion to nitrate and ammonium. The increase in porewater ammonium concentrations compared to porewater nitrate and nitrite concentrations, particularly from 3 to 9 cm depth, could be due to organic matter mineralisation processes (Kristensen, 2000; Al-Rousan et al., 2004; Jantti and Hietanen, 2012). Ammonium production deeper in the sediments could be through the process of ammonification, where nitrogen is incorporated into inorganic forms via microbially mediated reactions (Dong et al., 2009). The surface of the sediment was depleted of ammonium compared to the overlying water (Table 3.5), suggesting the consumption of ammonium near the sediment surface. The process of nutrient assimilation by diatoms might explain the rapid ammonium consumption near the sediment surface

as sources of nitrogen (Rasheed, 2004). It is also possible that availability of nitrogen from ammonium allows heterotrophs to break down organic carbon in the surface layers where there is oxygen available (Galloway et al., 2004). Ammonium is the preferable inorganic nitrogen to be assimilated over nitrate and nitrite by microphytobenthos, particularly diatoms (Canfield et al., 2010). The main reason is because less redox energy is required to incorporate its nitrogen into biomolecules, primarily amino acids and the nucleotide bases (Libes, 2009). The appearance of a diffusive boundary layer (Roy et al., 2002; Kelly-Gerreyn et al., 2005) from 0.5 mm to the sediment surface (Figure 3.4) might have changed porewater ammonium concentrations dramatically near the sediment surface. It should be noted that the depth resolution of 2 cm intervals might not have captured the extensive changes at the sediment surface. Phosphate concentrations increase with depth as a response to low dissolved oxygen in deeper sediment (Conley et al., 2009). With low oxygen, Fe (III) is reduced to Fe (II) releasing phosphate ions in the deeper sediment (Belias et al., 2007).

Alkaline phosphatase activity (AP) in the sediments ranges from 32.17 to 86.40 nmol  $g^{-1} h^{-1}$  (Figure 3.13 a), within the range reported for intertidal mudflats of the German Wadden Sea by Coolen and Overmann, (2000), in the Vellar Estuary, India (20 to 138 nmol  $g^{-1}h^{-1}$ ) (Ayyakkannu and Chandramohan, 1971) and within the range of 2 to 70 nmol  $g^{-1}h^{-1}$  in the Venice Lagoon, Italy (DeGobbis et al., 1986). AP was inversely related to porewater phosphate concentrations (Figure 3.14 a-b) potentially because autrotroph need less AP when free phosphate concentrations is high (Jones, 2002; Zhou et al., 2008; Steenbergh et al., 2011; De et al., 2012).

β-glucosidase (BG) activities of sediments in Breydon Water, Great Yarmouth (3.9 to 27.0 nmol g<sup>-1</sup> h<sup>-1</sup> in Figure 3.13 b), is similar to other studies conducted in intertidal marine sediments. King, (1986) reported BG activity of 23.2 nmol g<sup>-1</sup> h<sup>-1</sup> in intertidal sediments. In the sediments of a eutrophic lake, a study by Mallet et al., (2004) exhibited BG activity ranging between 1.0-26.1 nmol g<sup>-1</sup> h<sup>-1</sup>. Mallet and Debroas, (2001) has also shown BG activity of 30.7 nmol g<sup>-1</sup> h<sup>-1</sup> in superficial sediments of a eutrophic lake. To date, only one previous study has been conducted measuring cellobiohydrolase (CB) activity in lake sediments (Hakulinen et al., 2005). The majority of measurements of CB activity are in forest soils (Saiya-Cork et al., 2002; DeForest, 2009) and agricultural soils (Niemi and Vepsäläinen, 2005). So, this study is the first attempt to measure CB in estuarine sediments (Figure 3.13 c). Initial CB activities were between 1.0 and 22.7 nmol g<sup>-1</sup> h<sup>-1</sup> from sediments of Breydon Water, Great Yarmouth. Comparison with other systems showed CB activity was lower from this study, reflected by different sampling location and sources. Hakulinen et al., (2005) reported CB value of 410 nmol g<sup>-1</sup> h<sup>-1</sup> in Lake Uurainen and 60 nmol g<sup>-1</sup> h<sup>-1</sup> in Lake Jamijarvi, Finland. The higher values in this lake sediment reflected a forested catchment lake as the main source of cellulose degradation. A study by Sanaullah et al., (2011) in a grassland site, France reported CB activities ranging from 10 to 23 µmol g<sup>-1</sup> h<sup>-1</sup>, a thousand fold higher than the activities reported in this study. This is expected since grassland provides more cellulose inputs.

Chitinase activities in sediments (Figure 3.13 d) were relatively low compared with those of other enzymes. This may be due to the heavy seaweed growth, *Ulva* sp. which covered the mudflats at the time of sampling as this will input organic matter that does not contain chitin to the sediments (section 2.3). It may also be that the sediment contains other carbon sources than chitin. LeCleir and Hollibaugh, (2006) has reported chitinase activity in sediments of a salt lake, Mono Lake of 3.3-81.1 nmol g<sup>-1</sup> h<sup>-1</sup> and 137-874 nmol g<sup>-1</sup> h<sup>-1</sup> for 4°C and 20°C incubations, respectively which were contributed by high numbers of *Artemia* in the lake and the chitin rich exuvia they produce during the maturation process. Bacteria capable of chitin hydrolysis were generally rare in estuarine and marine environment (Kollner et al., 2012). This is probably related to a high availability of very diverse food substances, many of which can be metabolised more easily than chitin (Podgorska and Mudryk, 2003).

Urease activity was detected from 1 to 9 cm depth, recording the highest in the upper sediment (313 nmol  $g^{-1} h^{-1}$ ) and decreasing with depth by a factor of 10 (Figure 3.13 e). The method used in detecting urease activity was adapted from the protocol commonly used in soils of forest systems (Sinsabaugh et al., 2000; Saiya-

Cork et al., 2002; Sinsabaugh et al., 2005). Urease activities were between 700 to 2960 nmol  $g^{-1}$  h<sup>-1</sup> in four English river sediment; River Lathkill, River Wye (Derbyshire), River Swale and Porter Brook in Sheffield (Duddridge and Wainwright, 1982). A study by Sabil et al., (1995) on urease activity in Venice Lagoon sediment reported values of 1140 to 4400 nmol  $g^{-1}$  h<sup>-1</sup> using a spectrophotometric method.

Surprisingly, the overall pattern of enzyme activities in initial treatment resembles the pattern observed in the 4DA treatment (Figure 3.15 a-b). What might be happening? Initial sampling was carried out during high oxygen saturation of 75 % which was different from the 4DA treatment. Possibly, this reflects the rapid changes of microbial communities during the early stage of sediment handling prior to transferring to the cold room. The vertical profiles of AP, CB, BG, chitinase and urease activity were consistently higher in the presence of high TOC at the sediment surface (Figure 3.14 e and Figure 3.16 g) and decreased in deeper sediments with decreasing organic matter (Figure 3.16 a-h). In the sediments, the highest enzyme activity occurred in the 0-5 cm surface layer and decreased rapidly with depth (King, 1986). High inputs of organic matter are followed by enhanced microbial activity of heterotrophic bacteria (Cavari and Hadas, 1979), inducing the synthesis of enzymes (Riadas and Pinkas, 1997).

# 4.4 pH in the Overlying Water and Total Organic Carbon (TOC) and Total Nitrogen (TN) in the Sediment under Five Different Treatments

The pH response showed an increase of 0.2 units, equivalent to an average difference of 4 nmoles  $L^{-1}$  hydrogen ion concentrations in the anoxic treatments relative to the oxic treatment (Figure 3.1). If we examine hydrogen ion concentrations, a big difference was observed between the anoxic and oxic treatments which might be overlooked if data are presented as pH units. Nevertheless, both these measurements showed a move towards more alkaline pH in the anoxic treatment. The more alkaline pH in anoxic condition does not cause major physiological or biogeochemical changes in the treatments and does not relate to enhanced acidification problems (Howarth et al., 2011). A decline of 0.2 units may cause physiological changes to the calcifiers or non-calcifiers inhabiting the system (Mucci et al., 2011). It is likely that the ecological and biogeochemical consequences of anoxia are much larger than the effects of a change of 0.2 pH units.

TOC and TN showed decreasing patterns with depth (Figure 3.2 a and b). This pattern reflects reduced organic matter lability with depth (Souza, 2009). All treatments showed rapid usage in sediment TOC and TN compared to the initial condition (Figure 3.2 a and b). This implies that up to half of the organic matter is broken down in 7 days, even at depths well below the depth of oxygen penetration. During organic matter degradation, organic carbon and nitrogen are remineralised (Kristensen, 2000). This leads to a decreasing TOC and TN with depth until non-degradable fraction remains. But, degradation of this proportion of the organic matter is unlikely to occur over a timescale of only 7 days. Presumably, the rapid use of organic carbon reflects various benthic organisms and microorganisms depending on the limited organic carbon source since primary production by photoautotrophs is inhibited in the closed dark incubation.

The highest loss of TOC and TN at all depths in the sediment (1 to 9 cm depth) is in the oxic treatment at the end of the 7 day laboratory incubation (Figure 3.2 a and b). This would imply enhanced preservation of organic carbon in the anoxic relative to the oxic treatment, although the difference is quite substantial to occur within only 7 days. Lower degradation of TOC and TN in the anoxic treatments (Figure 3.2 a and b) can be related to restricted mixing (Lee, 1992) as shown by less exchange of Br in Figure 3.3 in the anoxic treatments and possibly, lack of oxygen. Presumably, the lower decrease of TOC in the anoxic treatment compared to the oxic treatment may be caused by the destruction of macrofauna and the disruption of the microbial loop via changes in bacterial grazers (Bianchi et al., 2000). In the absence of grazers who feed on bacterially derived organic matter, thus organic carbon may be less efficiently remineralised and preferentially preserved (Burdige, 2007). Bioturbation and the macrofaunas contribution to the catabolism of sedimented organic matter are also absent (Burdige, 2007).

The CN ratio in the oxic treatments at 7 cm depth (Figure 3.2 c) is closest to the Redfield ratio of 6.6, that implies labile organic carbon (Heilskov and Holmer, 2001) but this is not possible at 7 cm depth where the sediment is less likely exposed to rapid changes and the sediment have been deposited at that depth for quite some time. Higher CN ratio in the hypoxic and anoxic treatment, suggests a shift to refractory organic matter relative to the oxic treatment, but organic matter diagenesis would not happen in such a short timescale. A decrease of CN ratio suggests preferential organic matter processing of carbon compared to nitrogen, presumably due to the consumption of carbon by heterotroph and incorporation of nitrogen during degradation of organic matter (Meyers, 1997; 2003). Overall, this is a paradox and we could not point out the reason for what is happening. The TOC and TN pattern require further study, perhaps by measuring sulphate reduction rates, the microbial communities or isotopic carbon and nitrogen composition and biomarkers of carbohydrate, amino acid etc to get a better picture.

# 4.5 Nutrient Concentrations and Measured Nutrient Fluxes in the Overlying Water under Five Different Treatments

The present study demonstrates the importance of oxygen in regulating the nitrogen and phosphorus cycles. This section begins by explaining the possible underlying mechanisms involved in nutrient concentrations and nutrient fluxes. Secondly, section 4.6 assesses the relationship of nutrient concentrations and porewater nutrient concentrations; thirdly, section 4.7 covers the relationship of porewater nutrient concentrations and porewater diffusive fluxes, as well as nutrient fluxes and porewater diffusive fluxes; fourthly, section 4.8 explains the patterns observed in the enzyme activities and fifthly, attempts to relate between nutrients and enzyme activities. The relationships between bromide, oxygen fluxes, TOC and TN in the treatments are also discussed as these variables influence nutrient exchange and enzyme activities.

The following discusses the processes influencing nutrient concentrations and nutrient fluxes in each treatment. In the overlying water of oxic treatments, there was a loss of 2  $\mu$ mol L<sup>-1</sup> nitrate concentrations simultaneous with an increase of 4  $\mu$ mol L<sup>-</sup>

<sup>1</sup> nitrite concentrations (Figure 3.5 b and c). This pattern shows a gain of 2  $\mu$ mol L<sup>-1</sup> nitrite concentration, implying nitrate reduction to nitrite. At the same time, ammonium concentrations increase at first, then decreases after day 2. By day 7, 8  $\mu$ mol L<sup>-1</sup> ammonium concentration was loss, which may be partly due to ammonium oxidation (nitrification) from ammonium to nitrite (Camargo and Alonso, 2006; Bianchi, 2007; Beman et al., 2008), explaining the gain in nitrite concentrations. In terms of total nitrogen, nitrogen is lost from the overlying water. The decrease of ammonium concentrations in the overlying water (Figure 3.5 d); causes a decrease of ammonium flux (Figure 3.6 c), into the sediment. Nitrite flux was also directed into the sediment by the end of the experiment in the oxic treatment (Figure 3.6 b). The direction of the fluxes is a response of improved oxygen supply in the treatments, indicating nitrification (Rasheed et al., 2006; Belias et al., 2007).

In the hypoxic treatment, the decreasing nitrate concentrations in the overlying water was consistent with moderately increasing nitrite and ammonium concentrations during the 7 days experiment (Figure 3.5 b-d). This pattern shows that nitrate reduction to nitrite and ammonium was carried out (Camargo and Alonso, 2006; Piehler and Smyth, 2011). Simultaneous nitrite and ammonium was released into the overlying water. Nitrate flux was directed into the sediment in the hypoxic treatment (Figure 3.6 a-c). This pattern shows how the concentration of both nitrites and ammonium under hypoxia is regulated by the availability of nitrates (Belias et al., 2007). This process appears to be an important sink of nitrate in coastal and estuarine sediments (Kemp et al., 2009; Washbourne et al., 2011).

Our re-aeration experiment revealed contrasting responses between the 1DA and 4DA treatments. The small increase of nitrate towards the end of the experiment in the 1DA treatment (Figure 3.5 b) indicates that some nitrification from nitrite to nitrate was occurring (Figure 3.5 b-c). Re-aeration caused ammonium to be oxidised to nitrate and nitrite through nitrification (Belias et al., 2007; Canfield et al., 2010). The increase of nitrite and decrease of ammonium after day 3 in the 1DA treatment (Figure 3.5 c and d) resulted from ammonium oxidation to nitrite in an aerobic environment (Madsen, 2008). The increase of nitrite concentrations over time

(Figure 3.5 c) is consistent with the nitrite flux out into the overlying water in the 1DA treatment (Figure 3.6 b).

After re-aeration in the 4DA treatment, a slight increase in nitrite concentrations was measured in combination with low nitrate concentrations (Figure 3.5 b and c). This pattern indicates that the system does not recover quickly from anoxia. Continuously increasing ammonium concentrations were observed (Figure 3.5 d) despite the system being continuously aerated during the last 3 days of the experiment. This may result from continuing organic matter decomposition in anoxic conditions (Kemp et al., 2005; Prokopenko et al., 2011). It could also be that it takes some time for the ammonium to diffuse out of the sediments as observed from the direction of the ammonium diffusive flux into the sediment in the 4DA treatment (Figure 3.12 e). Higher nitrate flux into the sediment in the 4DA treatments on day 2 (Figure 3.6 a) is consistent with the decreasing nitrate concentrations in the overlying water (Figure 3.5 b). This pattern was probably caused by increased nitrate reduction to ammonium in the sediment as a response to oxygen depletion (Larson and Sundback, 2008; Caffrey et al., 2010). Nitrate release into the overlying water was observed in the 4DA treatment at the end of the experiment (Figure 3.6 a). This release was coupled with ammonium flux into the sediment (Figure 3.6 c) which could indicate that nitrifying bacteria were unable to meet their ammonium demand from the sediment (Soetart et al., 2006). A flux of nitrogen into the sediment would enable bacteria in the sediment to take up nitrogen from the porewater.

In the anoxic treatment, very low nitrate and nitrite concentrations were coupled with progressively increasing ammonium concentration (Figure 3.5 b-d). This pattern indicates the dominance of nitrate and nitrite reduction into ammonium (Belias et al., 2007; Jantti and Hietanen, 2012). Ammonium release to the overlying water (Figure 3.6 c) is commonly observed in anoxic condition (Neubacher et al., 2011; Hietanen et al., 2012).

In the oxic treatments, the decreasing pattern of phosphate over time may be attributed to the high removal of phosphate at the oxic sediment-water interface. At this sediment-water interface, phosphate can be bound to the surface of carbonate grains of the sediments (Rasheed, 2004). Re-aeration after 1 day of anoxia (Figure 3.6 d) was probably sufficient for the capture of released phosphate back to the sediment surface. This was also evident from the quickly decreasing concentration of phosphate in the water column in this treatment (Figure 3.5 e). Similar sorption of phosphate from the water column to the sediment during a positive redox-shift in lab experiments has been described by Hupfer and Lewandowski, (2008). After the shift from anoxic to oxic conditions, the sediment started to oxidise and bind phosphate more efficiently (Kemp et al., 2005; Hietanen and Lukkari, 2007; Jordan et al., 2008). In anoxic conditions, phosphate is continuously released due to iron mobilisation at the sediment-water interface (Golterman, 2001; Hupfer and Lewandowski, 2008; Middelburg and Levin, 2009).

Overall, distinct patterns of nutrient response in each treatment were observed (Figure 3.7 a-b). Nitrite increases in oxic conditions as a result of ammonium oxidation to nitrite (Figure 3.9 c and e) whilst nitrate decreases through nitrate reduction to nitrite. Hypoxia caused a continuous increase of ammonium through nitrate reduction (Conley et al., 2009; Bianchi et al., 2010). Under reducing conditions, ammonium increases (Hietanen et al., 2012). The re-aeration experiment showed that lower ammonium concentrations were observed in the 1DA treatment in comparison to the hypoxic treatment (Figure 3.5 d and 3.9 e). The lower ammonium concentrations in the 1DA treatment might be due to the re-oxygenation, which leads to nitrification (Conley et al., 2007). The extensive increase of ammonium in the 4DA treatment closely resembled the anoxic treatment (Figure 3.5 d); indicating increased ammonification of organic matter in the sediment (Conley et al., 2007). The hypoxic and anoxic treatments led to an increase of phosphate concentration in the overlying water as phosphate bound to iron oxides is released when iron (III) is reduced (Hietanen and Lukkari, 2007). Low phosphate concentrations in the oxic and 1DA treatments (Figure 3.9 g) reflect phosphate adsorption or coprecipitation with iron (III) in oxic conditions (Rasheed et al., 2006).

Figure 3.8 a-b shows the relationship between dissolved oxygen and nutrient fluxes. When oxygen saturation decreases, nitrate fluxes were into the sediment, indicating that nitrate production by nitrification was low relative to nitrate removal

by processes such as denitrification (Caffrey et al., 2010). In contrast, at high oxygen saturation, nitrite fluxes were into the overlying water (Figure 3.9 c), reflecting oxidation of ammonium to nitrite. Large release of ammonium and phosphate to the overlying water in low oxygen condition (Figure 3.9 e and g) indicates that the sediments in short term anoxic treatments act as a source of ammonium and phosphate to the system, processes that would exacerbate eutrophication problems in the field (Kemp et al., 2009).

## 4.6 Nutrient Concentrations and Porewater Nutrient Concentrations in the Sediment under Five Different Treatments

A framework to integrate all the parameters measured in this study is proposed in Figure 4.1 a-e. This section refers to the framework to link between nutrient and porewater nutrient concentrations in each treatment. As the overlying water consisted of nutrient free artificial seawater at the start of the laboratory incubations, the nutrients measured in the overlying water were clearly released from the sediment to the overlying water (Figure 3.5 b-e and Figure 3.11 a-d). Extensive exchange of overlying water and porewater in the sediment is shown by the vertical profile of porewater bromide concentrations (Figure 3.3). The fluxes of bromide was measured as a tracer of overlying water and porewater exchange on the assumption that lack of oxygen will cause the activity of benthic infauna to cease and thereby limit exchange to that accomplished by molecular diffusion.

The present study indicates that the process responsible for enhanced water column and porewater exchange is irrigation. In the oxic treatment, bromide irrigation flux was 937  $\mu$ mol m<sup>-2</sup> h<sup>-1</sup> compared to the lower bromide diffusive flux of 537  $\mu$ mol m<sup>-2</sup> h<sup>-1</sup> (Table 3.3). Vertical profiles of bromide also showed the deepest bromide exchange up to 9 cm depth in the oxic treatment (Figure 3.3). The higher irrigation flux in the oxic treatment, suggests that biologically driven sediment irrigation occurs when the burrows of benthic infauna, which are filled with overlying seawater exchange solutes with porewaters; are flushed frequently due to the infaunas activities. During the field sampling and porewater extraction from the sediment of this study, the ragworm, *Nereis diversicolor* were observed. So, most

probably higher irrigation was related to the presence of this polychaete. But, further study is needed on the exact number of the worms to get a concrete picture. Hedman et al, (2011) have reported that irrigation caused by the movement of benthic invertebrates can cause physical transport of bromide enhancing overlying water exchange with porewater.

Extensive irrigation between the overlying water and porewater can have an effect to the diagenetic reactions by transporting oxygen deeper (Green et al., 2002; Quintana et al., 2007). This is consistent with oxygen penetration depth (OPD) in the oxic treatments increasing from 4 mm to 5.5 mm after 7 days of laboratory incubations (Figure 3.4). A significantly higher diffusive oxygen flux in the oxic treatment (Table 3.4) could be related to bacterial activity in the sediment (Rowe et al., 2008), using oxygen for organic matter degradation and for oxidation of reduced compounds (Giles et al., 2007; Mouret et al., 2010). Figure 4.1 a shows the link between irrigation and OPD in the oxic treatment.

Nitrification occurs in the oxidised sediment surface (Lavery et al., 2001) as shown in the upper sediment surface of oxic sediments (Figure 4.1 a). Porewater ammonium profiles show increasing concentration with depth as it is produced from the degradation of organic matter (Mortimer et al., 1998). Below the oxic zone, porewater nitrate decreases with depth. This decrease indicates nitrate conversion into porewater ammonium concentrations through the process of nitrate reduction in the deeper sediment (Figure 4.1 a). Porewater ammonium in the deeper sediment diffuses upward and was consumed near to the sediment surface. Porewater nitrate in the deeper sediment might undergo denitrification and be released as nitrogen gas from the sediment. Nitrate generated in nitrification is reduced in surrounding anaerobic zones via denitrification to gaseous nitrogen (Kemp et al., 2009). But, there is an unclear relationship of nitrate to nitrite in the porewater of the sediment (Figure 3.10 a, Figure 3.11 a-b and Figure 4.1 a). Where has nitrite come from in the porewater? Nitrite is the intermediate in both nitrification and denitrification and therefore fluxes are difficult to interpret, since these processes are often closely coupled in marine sediments (Dong et al., 2009). Irrigation released ammonium to the overlying water (Figure 4.1 a). Ammonium release from the sediment is enhanced by bioirrigation and excretion by macrofauna (Mortimer et al., 1998). Ammonium is then oxidised to nitrite; and nitrate also reduces to nitrite in the overlying water of oxic treatment, accumulating nitrite in the overlying water (Figure 4.1 a).

In the oxic treatment, phosphate is released to the sediments from organic matter mineralisation (Figure 4.1 a). Phosphate is then bound to iron (III) (Denis and Grenz, 2003), causing low concentrations of phosphate in the sediment porewater and the overlying water (Figure 3.11 d). Irrigation at the sediment-water interface may release phosphate to the overlying water (Figure 4.1 a).











Figure 4.1 Framework of biogeochemical processes in the sediments under a) oxic, b) hypoxic, c) 1DA, d) 4DA and e) anoxic treatments. The schematic highlights the different response of nutrients and enzyme activities in each treatment. The response is also supported by bromide fluxes, oxygen penetration depth and fluxes and total organic carbon and total nitrogen in each treatment. OW- overlying water,  $NO_3^-$  nitrate,  $NO_2^-$  - nitrite,  $NH_4^+$  - ammonium,  $PO_4^{3-}$  - phosphate, OPD- oxygen penetration depth, Deeper sed- deeper sediment, Fe (III)-P – iron (III)-phosphorus, Fe(II)-P – iron (II)-phosphorus, AP- alkaline phosphatase activity OM- organic matter, CB- cellobiohydrolase, BG- beta-glucosidase,  $\longrightarrow$  clear relationship, unclear relationship  $\stackrel{\square}{\rightarrow}$ .

Oxygen penetrated less than 2.5 mm depth into the sediment in the hypoxic treatment, corresponding to low oxygen diffusive fluxes (Table 3.4 and Figure 4.1 b). Both the penetration of oxygen into the sediment and consumption of oxygen within the sediment decrease in hypoxic conditions (Neubacher et al., 2011). The decrease is explained by a reduction in the concentration gradient and the narrowing of the oxic layer (Diaz and Rosenberg, 2008). Ammonium is released to the sediment through organic matter mineralisation (Figure 4.1 b). Organic matter degradation produces porewater nutrients, ammonium and phosphate (Clavero et al., 2000). As the OPD is reduced, ammonium diffuses down into the deeper sediment (Figure 4.1 b). Nitrite reduction to ammonium occurs in the deeper sediment, and ammonium is then released to the overlying water through diffusion. There is an unclear link of nitrate to ammonium in the sediment as nitrate profiles were almost constant with depth (Figure 3.10 a and c; Figure 4.1 b).

Higher bromide diffusive fluxes of 889  $\mu$ mol m<sup>-2</sup> h<sup>-1</sup> was measured compared to the sediment irrigation fluxes of 417  $\mu$ mol m<sup>-2</sup> h<sup>-1</sup> in the hypoxic treatment (Table 3.3). Possibly, reduced irrigation during hypoxia releases ammonium into the overlying water in this treatment (Figure 4.1 b). In oxygen depleted treatments such as hypoxia; molecular diffusion dominates, influenced by the reduction of macrofauna (Foster et al., 1995). Burrowing and irrigation of macrofauna are suppressed following hypoxia (Rasmussen and Jorgensen, 1992). Nitrate is also reduced to nitrite and ammonium in the overlying water (Figure 4.1 b). Hypoxia suppresses nitrification and caused nitrate to be reduced to ammonium (Kemp et al., 2009). The overall framework implies that nitrogen is retained in the form of ammonium in the hypoxic treatment (Figure 4.1 b). The release of phosphate into the overlying water is through reduced irrigation (Figure 4.1 b) and Fe(III) reduction to Fe (II) (Emeis et al., 2000; Belias et al., 2007), releasing in turn phosphate ions into the overlying water and the porewater of hypoxic treatment (Figure 3.11 d and Figure 4.1 b).

In spite of the aeration supplied throughout the remaining 6 days of the experiment to the overlying water in the 1DA treatments, there is unclear evidence of nitrification in the surface sediments (Figure 3.10 a-c and Figure 4.1 c). There is also

unclear relationship between nitrate to nitrite and ammonium in the porewater (Figure 3.10 a-c and Figure 4.1 c). Increasing ammonium concentration was observed in the deeper sediment through nitrite reduction to ammonium (Figure 4.1 c). Despite the continuous air supply, nitrite reduction to ammonium is still occurring because full oxygenation of the sediment is not taking place (Belias et al., 2007). Consistently, vertical oxygen profiles in the 1DA treatment (Figure 3.4) only reached 5.5 mm depth in the sediment (Figure 4.1 c), causing the buildup of inhibitory metabolites such as ammonium below this depth. The sediment surface is generally oxidised only within the upper 3 mm for fine- grained sediment (Revsbech et al., 1980; Glud, 2008). Changes of oxygen concentration to the water above the sediments may alter the penetration depth of oxygen within the sediment (Rasmussen and Jorgensen 1992). With less oxygen penetration in the sediment, nitrification rates are slow, increasing ammonium concentration (Kemp et al., 2009). Ammonium is released to the overlying water through diffusion from the deeper sediment and irrigation from the sediment surface (Figure 4.1 c).

Total bromide incubation fluxes were 1.8 times the bromide diffusive fluxes (Table 3.3), showing the influence of irrigation in this treatment. Bromide irrigation in the 1DA treatment ( $611 \mu mol m^{-2} h^{-1}$ ) is also higher than in the hypoxic treatment ( $417 \mu mol m^{-2} h^{-1}$  in Table 3.3). Consistent with this, higher diffusive oxygen flux and deeper oxygen penetration depth was also observed in the 1DA treatment relative to that in the hypoxic treatment (Figure 3.4 and Table 3.4). In the 1DA treatment, as oxygen concentration was increased (Table 3.4), anaerobic metabolites could be chemically re-oxidised causing an increase in diffusive oxygen flux (Murrell and Lehrter, 2010). Presumably, benthic infauna present in the cores were able to withstand 24 hours without oxygen and the benthic community compensated for a lack of oxygen by pumping more overlying water through the sediment (Foster et al., 1995). Gamenick et al. (1996) has found that 50% of *Nereis diversicolor* adult and juvenile can survive up to 60 hours without oxygen. This result suggests that brief anoxic events up to 24 hours can be tolerated by benthic infauna.

In the 1DA treatment, ammonium released from irrigation and diffusion from the sediment is oxidised to nitrite and nitrate through nitrification in the overlying water (Figure 4.1 c). Low nitrite concentration relative to that in the nitrate concentration (Figure 3.11 a and b) also point towards nitrification. In the presence of oxygen, ammonium tends to be oxidised to nitrate (Kemp et al., 2009). Overall, nitrate is retained in the 1DA treatment. Phosphate concentration is low in both the overlying water and porewater in the 1DA treatment (Figure 3.11 d) due to the formation of iron (III)-phosphate complex (Rasheed 2004; Kemp et al., 2009). This formation in the sediment inhibits the transport of phosphate into the overlying water in the 1DA treatment (Figure 4.1 c).

After 3 days of aeration, the oxygen concentration did not seem to recover in the 4DA treatments (Figure 3.5 a). Ammonium accumulates in the porewater from nitrite reduction to ammonium in the deeper sediment and is released to the overlying water through diffusion (Figure 4.1 d). With less oxygen, less water exchange between the overlying water and sediment (Figure 3.3) thus, less bromide going into the sediments (Martin and Banta, 1992). The shape and smoothness of the bromide profiles indicated that diffusion or diffusive-like transport process (Berg et al., 2001) was the dominant process in the sediment of 4DA treatment. This is not surprising if all the infauna are dead, releasing ammonium in the treatment. However, there is an unclear relationship of nitrate to nitrite and ammonium in the porewater with increasing depth of the sediment (Figure 3.10 a-c and Figure 4.1 d). Figure 3.11 a-c suggests nitrate and nitrite reduction to ammonium in the porewater and overlying water. Overall, nitrogen is stored as ammonium in the 4DA treatment (Figure 4.1 d). Oxidised iron minerals in the sediment are reduced with low oxygen conditions releasing bound phosphate (Conley et al., 2007). The different response between the 1DA and 4DA treatment points out that the different duration of anoxic exposure may act as an important factor in determining the response of nitrogen and phosphorus cycle (Conley et al., 2007; Katsev et al., 2007; Riedel et al., 2008; Middelburg and Levin, 2009).

In the anoxic treatment, porewater nitrate and nitrite are reduced to ammonium in the sediment (Figure 3.11 a-c and Figure 4.1 d). Anoxic conditions favour ammonium release from the sediment by anaerobic breakdown of organic matter rather than the release of nitrate and nitrite (Kristensen, 2000; Middelburg and Levin, 2009). Ammonium is also released to the overlying water in anoxic conditions through diffusion (Figure 4.1 e). In the absence of dissolved oxygen, phosphate also accumulates in the sediment (Figure 3.11 d and Figure 4.1 e). In anoxic condition, phosphate is released during iron mobilisation at the sediment-water interface (Golterman, 2001; Hupfer and Lewandowski, 2008; Middelburg and Levin, 2009). Studies have shown that anoxic condition results in the release of phosphate (Rozan et al., 2002; Skoog and Arias-Esquivel, 2009; Ozaki et al., 2011) from the sediment to the porewater and the overlying water. Irrigation flux is low relative to diffusive flux in the anoxic treatment (Table 3.3). The lowest bromide exchange as expected is shown in the anoxic treatments, which are primarily driven by molecular diffusion (Libes, 2009). With the absence of infauna which irrigates the sediment through burrows, no oxygen penetration was observed in the anoxic treatments (Figure 3.4), thereby supporting the dominance of molecular diffusion within this treatment (Forster et al., 1995; Glud et al., 1996). In accordance, the lowest oxygen diffusive flux was also measured in the anoxic treatments (Table 3.4). The absence of oxygen rapidly suppresses the oxygen penetration depth into the sediment (Abell et al., 2011).

# 4.7 Porewater Nutrient Concentrations and Porewater Diffusive Fluxes; and Nutrient Fluxes and Porewater Diffusive Fluxes in the Sediment under Five Different Treatments

This section begins with the relationship between porewater nutrient concentrations and porewater diffusive fluxes in each of the treatments. This is followed by the discussion about nutrient fluxes and porewater diffusive fluxes. Decreased porewater nitrate and increased porewater ammonium concentrations in the oxic treatment indicate the occurrence of nitrate reduction into ammonium with increasing depth in the sediment (Figure 3.10 a and c). But, where has nitrite come from in the porewater? Vertical porewater nitrite with depth (Figure 3.10 b) remains relatively the same that limits our interpretation. Looking at the diffusive fluxes in the oxic treatment, simultaneous porewater nitrate flux into the overlying water (Figure 3.12 a), and the flux of nitrite and ammonium into the sediments was measured (Figure 3.12 a, c and e). These fluxes suggest that nitrification is occurring in the oxidised sediment surface layer (Lavery et al., 2001). However, comparison between the porewater nutrient concentrations and porewater diffusive fluxes in the oxic treatment shows an inconsistency between these 2 parameters. Porewater nutrient concentrations suggest nitrate reduction to ammonium whereas porewater diffusive fluxes indicate nitrification.

In the case of 1DA treatment, fairly constant vertical profiles of porewater nitrate concentrations with depth were observed (Figure 3.10 a), that limits our interpretation. However, nitrite reduction to ammonium is clearly observed at 9 cm depth (Figure 3.10 b). A linear increase in ammonium concentrations over space is related to the degradation of organic matter (Denis and Grenz, 2003). Porewater nitrate diffusive fluxes show a similar pattern to the oxic treatment (Figure 3.12 a). The pattern might be due to nitrification in the 1DA treatment. Nitrite diffusive flux into the sediment simultaneous with nitrate diffusive flux directed into the overlying water is indicative of nitrite consumption by nitrification (Mortimer et al., 1998). Again, an inconsistency is observed between porewater nutrient concentrations and porewater diffusive fluxes. The occurrence of nitrification suggested by the porewater diffusive fluxes is not consistent with the pattern of nitrite reduction to ammonium observed from the vertical porewater nutrient concentrations in the 1DA treatment.

In the hypoxic, 4DA and anoxic treatments, the majority of porewater DIN is in the form of ammonium throughout the core with low porewater nitrate and nitrite concentrations (Figure 3.10 a-c). The reduction of porewater nitrite to ammonium occurred in the hypoxic and 4DA treatments (Figure 3.10 b and c). In the anoxic treatment, porewater nitrate and nitrite are reduced to ammonium, demonstrating anaerobic degradation of organic matter (Mortimer et al., 1998; Rasheed et al., 2006; Middelburg and Levin, 2009). Under low oxygen conditions, ammonium is released (Abell et al., 2011).

Looking at the porewater diffusive fluxes in the hypoxic, 4DA and anoxic treatment, porewater nitrate diffusive fluxes were directed to the overlying water in the hypoxic and 4DA treatments but were into the sediments in the anoxic treatment

(Figure 3.12 a). On the other hand, the fluxes of nitrite and ammonium were into the sediment (Figure 3.12 c and e). This pattern indicates decreased nitrification (Denis and Grenz, 2003) in the hypoxic and 4DA treatments. In the anoxic treatment, nitrate uptake into the sediment was observed, suggesting denitrification (Mortimer et al., 1998). But, the diffusive fluxes of nitrite and ammonium in the anoxic treatment were into the sediments, suggesting nitrification. One possible explanation is that ammonium uptake in these 3 treatments might reflect ammonium consumption by bacteria living at/near the sediment water interface (Maksymowska-Brossard and Piekarek-Jankowska, 2001). Core incubation was carried out in the dark, which would inhibit ammonium uptake by benthic photosynthetic autotrophs (Lerat et al., 1990).

Phosphate concentration increased extensively in particular in the 4DA and anoxic treatments (Figure 3.10 d). This pattern suggests reduction of ferric oxyhydroxide phases and release of iron-bound phosphorus to the porewater under low oxygen condition (Steenbergh et al., 2011). All phosphate diffusive fluxes were into the sediment (Figure 3.12 g) and although significant differences were measured between treatments, all diffusive fluxes are too small to reflect major changes in the sediment.

The variable pattern of the porewater diffusive flux might be due to the dominance of non-physical processes such as biological and chemical retardation processes controlling the fluxes; for example, the formation of iron (III)-phosphate complexes, nitrification and denitrification (Clavero et al., 2000; Lavery et al., 2001). All of these retardation processes may occur on timescales comparable with the 7 days of our flux experiment.

The variable pattern of porewater diffusive flux also leads to lack of consistencies between the measured nutrient fluxes and the calculated porewater diffusive fluxes. This indicates that Fick's 1<sup>st</sup> Law for diffusive fluxes calculation should be used with caution.

# 4.8 Enzyme Activities in the Sediment under Five Different Treatments and the Relationship of Enzyme Activities and Other Parameters in the Sediment

The present study shows that AP, BG, CB, chitinase and urease profiles followed a similar decreasing trend moving down core from the surface of the sediment (Figure 3.13 a-e). The trend with depth was particularly marked in the CB activities, decreasing to near zero at 9 cm depth (Figure 3.13 c). It would be expected that the highest concentrations of both total and labile organic matter would occur at the sediment surface. This influx of fresh organic matter stimulates microbial activity in surface sediment (Boetius et al., 2000), leading to a corresponding increase in enzymatic decomposition rates (Souza, 2009). As a result, enzyme activities are highest in the surface or near surface sediments and decrease with depth (Meyer-Reil, 1986; Bowman et al., 2003; Liu et al., 2008). With increasing sediment depth, the quantity and quality of decomposable organic matter is reduced, and the remaining material becomes more and more resistant to microbial attack (Fabiano and Danovaro, 1998; Hill et al., 2006).

Factors that influenced enzyme activities in this study are shown in Figure 4.2. Each factor will be discussed in turn. This section also refers to the proposed framework in Figure 4.1 a-e to relate between enzymes and nutrients. Increasing AP, CB and chitinase activities were observed in the oxic treatments relative to that in the initial treatments (Figure 3.13 a, c and d). The inverse relationship of AP to porewater phosphate concentrations showed that AP is high when porewater phosphate concentration is low (Figure 3.14 a-b). There are two possible mechanisms that may cause this. Firstly, high phosphate concentration is known to act as an inhibitor of AP, based on the catalysis of AP to produce phosphate from organic phosphorus (section 1.6.1). Secondly, at high phosphate concentrations, there is a reduced requirement for the microorganisms to produce AP (Newman and Reddy, 1993; Zhou et al., 2008). In this study, increasing phosphate concentrations were observed in the deepest layers of anoxic sediment and presumably anaerobic microorganisms are not experiencing phosphate limitation, so the expression of AP was repressed (Steenbergh et al., 2011). When phosphate concentration are low in

the water, microorganisms or plankton may produce phosphatase to hydrolyse organic phosphorus (Newman and Reddy, 1993; Takano et al., 2006; Deng et al., 2009; Chen et al., 2011). Phosphate is immobilised by co-precipitation of Fe (III) in oxic condition (Figure 4.1 a), so AP must be released to allow microorganisms to overcome the resulting lower phosphate concentrations. AP is lower in the hypoxic, 4DA and anoxic treatments (Figure 3.14 a-b; Figure 4.1 c-e) that may result from AP inhibition in the presence of phosphate concentrations (Jones, 2002) as well as by inhibition of the reduced metal from Fe (III) to Fe (II) in the sediment (Jordan et al., 2008).

CB and chitinase activities increased in the oxic treatments relative to the anoxic treatments (Figure 3.13 c and d). There are two possible explanations for CB and chitinase activities in this study. Firstly, anoxia inhibits the activity of aerobic benthic organisms using chitin and cellulose as carbon sources. Secondly, a number of aerobic benthic organisms are killed in the anoxic treatment, so alternative carbon sources are available apart from CB and chitin. BG catalyses cellobiose to glucose (under section 1.6.3), which could be an alternative carbon source in the treatment. But, anoxia also reduces BG activity (Figure 3.13 b). Therefore, the patterns observed in CB, chitinase and BG activities might imply that anoxia inhibits the activity of aerobic benthic organisms using chitin, cellulose and BG as carbon sources (Figure 4.2).

Additionally, weak relationship between CB and TOC (Figure 3.16 g-h), suggests that other factors than TOC contributes to its activity. Chitinase activities were inversely correlated to porewater ammonium concentrations (Figure 3.16 i-j; Figure 4.1 and Figure 4.2), suggesting the relationship of chitinase in the nitrogen cycle. Chitinase activity is linked to the acquisition of nitrogen (Hill et al., 2012). Higher BG activities in the oxic treatment, suggests labile carbon acquisition as glucose is a very labile carbon source (Hill et al., 2012).



Figure 4.2 Factors affecting enzyme activities in the sediment of this study. C-carbon, PW  $NH_4^+$  - porewater ammonium, BG-  $\beta$ -glucosidase, CB- cellobiosidase, TOC- total organic carbon, TN- total nitrogen, AP- alkaline phosphatase, PW  $PO_4^{3-}$ -porewater phosphate.

Interestingly, urease activities significantly increased in the anoxic treatments relative to the oxic treatments at all depths from 1 to 7 cm depths in the sediment (Figure 3.13 e). What could produce such high urease activities? There was an inverse relationship between urease activities with porewater ammonium concentrations (Figure 3.14 c-d), reflecting a repression of urease activity by porewater ammonium concentrations. Solomon et al., (2010) have reported that urease activity was repressed by ammonium concentration in Chesapeake Bay due to less demand for ammonium via urease pathway. At the same time, high urease activities were positively related with total organic carbon (Figure 3.14 e-f) and total nitrogen, implying that organic carbon and nitrogen stimulates the activity of bacteria with urease enzymes (Figure 4.2). The positive correlation between TOC and TN content and urease activities indicates urease dependence on polymeric substrate content (Souza, 2009).

Similar to the anoxic treatments, urease activity was enhanced in the hypoxic and 4DA treatments. Possibly, the facultative and obligate anaerobic bacteria living in the sediment may have produced more urea in the sediment (Satoh, 1980) which 118 might exceed the high ammonium concentrations in the hypoxic and 4DA treatments. Urea can also increase through organism excretion and from release of the sediments (Lund and Blackburn, 1989; Therkildsen et al., 1996). Urease activity was not affected in the oxic and 1DA treatments. This is consistent with Zantua and Bremner, (1977) and Bremner and Mulvaney, (1978) who reported that oxygen had little, if any effect on urease activities in soil.

Overall, the response in the sediment of 4DA treatments followed closely to that in the anoxic treatment whereby 1DA treatments resembled oxic treatments (Figure 3.15 a). This pattern implies the importance of the duration of anoxia in regulating the nutrients and enzyme activities in estuarine sediment. The influence of depth profiles were prominent in all enzyme activities (Figure 3.15 b), most activities were highest on the surface, suggesting labile organic matter and less degraded with increasing depth (Kahkonen and Hakulinen, 2012). Enzyme activities were positively related to one another except urease (Figure 3.16 a, c and e). This indicates the close associations of AP, CB, BG and chitinase are related to the acquisition of carbon, nitrogen and phosphorus in fine-grained sediment (Hill et al., 2012). The fact that CB and chitinase activities were higher in the surface of the sediment (2 cm) than deeper layers (7-9 cm) in Figure 3.16 h and j indicates that there was highly active cycling of carbon and nitrogen in the surface sediment.

### 4.9 Conclusions

There were clear responses of nutrient concentrations in the overlying water to the oxic, hypoxic, 1DA, 4DA and anoxic treatments. These must reflect the combined effect of nutrient exchanges with the sediment and, for nitrogen, transformation between chemical forms. Concentrations of nutrients in the porewater were less straight forward to interpret. This leads to an apparent disagreement between measured nutrient fluxes in the overlying water and the calculated porewater diffusive fluxes. This is influenced by chemical and biological processes occurring within the sediment that control the release of nutrients. Overall, looking at the fluxes estimated from nutrient concentrations in the overlying water (Figure 3.6 a-d), the cores act as sinks for nitrate, minor sources for nitrite and phosphate and major

sources of ammonium in the overlying water. Examining porewater concentrations alone would give the opposite conclusions, indicating that porewater is a source for nitrate and a sink for nitrite, ammonium and phosphate (Figure 3.12 a, c, e and g). Referring to each treatment, nitrite accumulates in the oxic treatment (Figure 4.1 a). Nitrate is retained in the overlying water of 1DA treatment (Figure 4.1 c). Ammonium is retained in the hypoxic, 4DA and anoxic treatment (Figure 4.1 b, d and e), exacerbating eutrophication problems in oxygen depleted treatments. Phosphate is released into the overlying water in low oxygen conditions. In oxygenated conditions, phosphate is directed into the sediments, driven by the redox chemistry of iron.

Higher enzyme activities were found in the surface sediment with potential easier degradable organic material. Deeper sediment has less degraded material. The response of AP was related to porewater phosphate concentrations (Figure 4.2); and chitinase was inversely related to porewater ammonium concentrations (Figure 4.2). Weak relationships of AP, CB, BG and chitinase activities to total organic carbon and total nitrogen in the sediment suggest other sources affecting these enzymes. In contrast, higher urease activities in the hypoxic, 4DA and anoxic treatments were probably influenced by both the porewater ammonium concentrations and total organic carbon in the sediment (Figure 4.2). Nevertheless, clear differences were observed in the vertical profiles of the enzyme activities. Therefore, enzyme activities can be proposed as indicator of ecological effects in the sediment in comparison to porewater nutrient concentration. Enzyme activities in the sediment are less affected to the 1 day of anoxia than the 7 days hypoxia. Based on the robustness of the system to the 1DA treatment, the take home message is that the short term anoxia of 1 day is less likely to give a negative impact to the biogeochemical functioning of the estuarine sediment. The effect of hypoxia on a longer term should be considered to determine its effect to the environmental quality.

#### **CHAPTER 5**

#### SUMMARY AND FUTURE RECOMMENDATIONS

#### 5.1 Introduction

This chapter summarises the key findings that answer the scientific questions posed at the start of this research in Section 1.7. Selection of the experimental setup and timing of sampling events throughout this research were also included. This chapter also considers the limitations, remaining issues and future research priorities.

The research conducted for this thesis set out to evaluate the oxygen concentration at which harmful effects start to occur, the rates that the system responds to different levels of oxygen saturation in the overlying water and the recovery times of system processes when re-aeration are restored. The aims were achieved through the study of nutrient fluxes in the overlying water and enzyme activities in the sediment. The motivation of this thesis was triggered by the escalating growth of hypoxia (oxygen depletion to 25% saturation) and anoxia (absence of oxygen) in shallow waters and estuaries associated with human induced eutrophication. This led to the questions of:

- 1) Does short term hypoxia and anoxia affect pH in the overlying water and vertical profiles of total organic carbon and total nitrogen in the sediments?
- 2) Are differences in oxygen saturation in the overlying water reflected by changes in bromide solute transport between sediment-water interfaces and the oxygen consumption in the sediments?
- 3) How does oxygen affect nutrient cycling, which in turn influences nutrient fluxes?
- 4) Does the duration of hypoxia and anoxia influence the nitrogen and phosphorus transformations in the estuarine sediment-water system?
- 5) What factors affect enzyme activities in sediments?

## 5.2 Summary of Findings

This section provides answers to each research question posed in Section 1.7.

1) Does short term hypoxia and anoxia affect pH in the overlying water and vertical profiles of total organic carbon and total nitrogen in the sediments?

The finding in this study indicates that short term anoxia elevates pH slightly compared with oxic treatments, equivalent to a decrease of hydrogen ion concentrations or alkalinity (section 3.2.1). Presenting the pH changes in terms of hydrogen ion concentrations showed an average difference of 4 nmoles  $L^{-1}$  between oxic and anoxic treatments. This is an average difference of 0.2 pH. The actual magnitude of the increase may seem small, but pH is based on a logarithmic scale, so even a change of 0.2 units makes a substantial difference to hydrogen ion concentrations. But, since the anoxic treatment increases pH by 0.2 units, anoxia does not markedly enhance acidification problems. Only if the development of hypoxia is associated with a release of carbon dioxide, which can reduce the pH levels by 0.2 units (Mucci et al., 2011) will the physiological and biological activities of benthic organism would be affected (section 4.4).

In terms of total organic carbon (TOC) and total nitrogen (TN) distribution in the sediment, the reduction of TOC was 1% in the oxic treatment, 0.5% in the hypoxic treatment and 0.1% in the anoxic treatment relative to the initial treatments (Figure 3.2 a). Minor differences of TN were observed between treatments, less than 0.1% (Figure 3.2 b). The obvious pattern of vertical profiles of TOC and TN, suggests the decreasing substrate availability with depth (Kristensen, 2000). The smaller loss of TOC in the anoxic treatment in our study relative to the oxic treatment suggests preservation of organic carbon at low oxygen concentrations, but the loss is not feasible within a timescale of 7 days. CN ratio in the oxic and 4DA treatments at 7 cm depth (Figure 3.2 c) shows the closest value to the Redfield ratio of 6.6 and suggests labile carbon, but this is not possible at 7 cm depth where the sediment is highly unlikely to be exposed to rapid changes. The TOC and TN pattern is inconclusive and requires further study (section 4.4). 2) Are differences in oxygen saturation in the overlying water reflected by changes in bromide solute transport between sediment-water interfaces and the oxygen consumption in the sediments?

Higher solute transport between sediment and overlying water in the oxic treatments using bromide as a tracer, implies that water movement is controlled by irrigation fluxes (Table 3.3) whereas in the hypoxic and anoxic treatments, it is dominated by diffusive fluxes. Under Table 3.4, highest oxygen diffusive fluxes in the oxic treatment; reflect bacterial activity in the upper milimetres of the sediments relative to the anoxic treatments.

3) How does oxygen affect nutrient cycling, which in turn influences nutrient fluxes?

The nitrogen and phosphorus cycles are strongly regulated by oxygen. As mentioned in section 3.5.2, nitrate concentrations are high when dissolved oxygen saturation is high and started decreasing when dissolved oxygen saturations were low. The largest nitrate flux into the sediment occurred when the dissolved oxygen was low (section 3.5.2). In oxygenated condition, nitrate is the thermodynamically stable form of nitrogen (Thamdrup and Dalsgaard, 2008).

Nitrite accumulates in oxic conditions, but in anoxic conditions this is reduced further to ammonium (Figure 3.9 c and e). Nitrite concentrations increase in the overlying water and ammonium concentration increases initially in the oxic treatment because the volume of overlying water is small relative to the quantities being released from the sediment in the cores during incubations. At low oxygen saturation, ammonium and phosphate concentrations increase, corresponding to ammonium and phosphate effluxes to the overlying water but were low at high oxygen saturation (Figure 3.9 e and g). Ammonium is inevitably being produced from both nitrate and nitrite reduction in the absence of oxygen (Middelburg and Levin, 2009). When supplied with re-aeration, most of ammonium were partly oxidised to nitrite before being converted into nitrate concentrations. With oxygen, phosphate remains relatively low, showing an uptake into the sediment (Figure 3.9

g). This pattern reflects a response of phosphate immobilisation as Fe (III) dominates in oxic condition (Paytan and McLaughlin, 2007). The consumption of oxygen and the predominance of hypoxia and the extensive anoxic treatments, in turn released phosphate concentrations into the overlying water. Re-aeration was sufficient to maintain phosphate at low concentrations in the 1DA treatment and reduces phosphate concentrations in the 4DA treatment (Figure 3.9 g).

Nutrient concentrations in the porewater were more variable and are hard to interpret (Figure 3.10). Nevertheless, clear increasing ammonium and phosphate concentrations were measured with depth, as a response of low oxygen concentrations (Kristensen, 2000).

4) Does the duration of hypoxia and anoxia influence the nitrogen and phosphorus transformations in the estuarine sediment-water system?

This study demonstrates that the duration of hypoxia and anoxia are the key factors determining the response on estuarine system. The system was surprisingly robust to short term anoxia, the 1DA treatment showed a very similar response to the effect in oxic treatments (Figure 3.15 a). Apparently, the effect of a brief episode of anoxia lasting up to 24 hours did not cause a severe impact to the system and effects were largely reversible, with the system recovering to normal condition within the 7 days of treatment. The effects observed in the 1DA treatment were less marked than in continuous hypoxia of 7 days. This was evidenced particularly by a smaller ammonium concentration increase in the 1DA treatment than that in the hypoxic treatment (Figure 3.5 d). Both hypoxia and longer duration (4 days) anoxia followed by re-aeration caused substantial negative effects on the system, with subsequent biogeochemical effects. Ammonium and phosphate effluxes were shown to dominate the nutrient fluxes in the 4DA treatment (Figure 3.6 c and d). In a system without oxygen (anoxia), molecular diffusion controls the responses of nitrogen and phosphorus dynamic in the system (Quintana et al., 2007; Hedman et al., 2011). Ammonium and phosphate dominate in the overlying water and porewater, exacerbating nutrient problems. So, sediment biogeochemistry has some resilience towards short term anoxic events, but more prolonged hypoxic or anoxic events that are continued for only a few days can have major effects on ecosystem function.

5) What factors affect enzyme activities in the sediments?

Increasing alkaline phosphatase (AP), cellobiohydrolase (CB), beta-glucosidase (BG) and chitinase activity were observed in the oxic treatments relative to that in the initial treatment, but urease showed the opposite pattern (section 3.6). The response of AP depends on the phosphorus requirement of microorganism (section 4.8). In the oxic treatments, phosphorus solubility is limited by coprecipitation with iron III (Middelburg and Levin, 2009; Chen et al., 2011). Therefore AP are synthesised to overcome phosphate limitations. Vertical profiles of CB and chitinase activities were highest on the surface and decrease with depth, due to shifts of organic matter input and quality as it goes deeper (Fabiano and Danovaro, 1998; Zonneveld et al., 2010). CB, BG and chitinase activities are lowest in the anoxic treatment, possibly because anoxia inhibits the activity of aerobic organisms using these enzymes as a carbon source (section 4.8). Increasing urease activity in the absence of oxygen may be related to elevated urea concentration from excretion by bacteria and death of organisms in anoxic condition that mitigates the repressive effect of high ammonium concentration on urease (Ruan et al. 2009; Mortensen et al. 2011).

## 5.3 Selection of Experimental Setup and Research Timeline

This section explains the decision to carry out the experiment as described in section 2. 4 and reflects on the research timeline. To address the aims and research questions outlined, oxygen concentrations were manipulated in 5 different treatments, oxic (96% saturation in the overlying water), hypoxia (25% saturation), 1DA (24 hours without oxygen followed by re-aeration of up to 6 days), 4DA (4 days without oxygen and subsequent re-aeration for 3 days) and anoxia (0% saturation). Although short term hypoxia and anoxia are generally confined to the warmer summer months, its occurrence and intensity tends to vary on daily to weekly time scales in estuarine system associated with periodic fluctuations in sunlight and tides as well as rain and wind events (Wenner et al., 2004; Taylor et al., 2009). Hence, the

significance of evaluating short term effect of anoxia within 1 day to 7 days. Although the sampling area selected for this study (Breydon Water) was not naturally hypoxic, the location was chosen with practicality and sample integrity in mind. Considering the cost of the gas mixtures ordered and the time consuming process of trying to set up the experiments and maintaining the oxygen concentrations, this study was focused on the effect of episodic anoxia, which normally occurs within 7 days. This study also test the effect of different duration of episodic anoxia; 1 day anoxia and 4 days anoxia and the re-aeration effects to the sediment biogeochemistry. The effects of the treatments to nutrients and enzyme activities were substantial, that it is reasonable to run the experiment for just 7 days.

Throughout this research, three sampling events were conducted, the first one (around Great Yarmouth) in August 2008 to survey the sampling location for the sediments and to test out the method of analysing nutrients using microplate analysis. But, problem was encountered with the nutrient analysis and no results were derived. I intercalated in February 2009 to October 2009 due to medical conditions and returned to Malaysia throughout the intercalation period. A second sampling campaign was carried out in November 2009 to test out the gas setup and preliminary experiment of using oxygen minielectrode, bromide tracer experiment and nutrient analysis using the autoanalyser but, the minielectrode did not function, and gas setup did not work. A third sampling was carried out in June 2010, and the data for this study were presented in Chapter 3 under the results section. Only one sampling event and one laboratory manipulation was carried out for this research mainly due to time and funding restrictions.

### 5.4 Recommendation for Future Research

This section explains the contribution and the limitations of this thesis. Future studies are also proposed. This experimental setup or laboratory simulation provides a guideline for nutrients and enzymes regulations applicable to estuarine system showing signs of short term (days to a week) oxygen depletion. Despite of that, the limitation of this study lies largely on the inability to reflect measurement exactly to the field situation. Ideally, all measurements are preferred to be carried out in the field but, this is not practically possible in the intertidal without substantially larger logistic resources. Even if it were practically possible to set up the incubations *in situ*, there will be complications because the water changes rapidly and in the case of Breydon Water, the sediments are exposed to the air for up to 2 days during periods of neap tides (section 2.3). Therefore, a realistic assessment of the consequences of short term hypoxia and anoxia could be achieved by conducting seasonal field sampling for a year taking into account the effect of tides in intertidal sediments.

Good quality data were produced by minimising the effect of variability (Figure 3.5) by using field sediment and nutrient free artificial seawater. However, by using nutrient free artificial seawater, this eliminates *in-situ* production of the water. To overcome this, *in-situ* seawater could be included as a control as recommended by NICE protocol handbook (Dalsgaard et al., 2000). This study used artificial seawater because water taken from shallow areas of estuaries may have been exposed to large diurnal changes in light, temperature, nutrients, dissolved oxygen that could affect the experimental results of this study (section 2.4).

Clear patterns of lower hydrogen ion concentrations in the anoxic treatment were shown, corresponding to higher pH in anoxic treatments relative to lower pH in oxic treatment (section 3.2.1). A consistent pattern of rapid usage of TOC and TN in the surface relative to deeper depths was also found (section 3.2.3). Future studies should consider measuring carbon dioxide, sulphide concentrations and alkalinity in the overlying water and sediments, which might provide a more detailed picture of what is happening.

The negative consequences of short term hypoxia and anoxia on the nitrogen and phosphorus cycles were clearly shown in an experiment that runs for only one week (section 3.5.1). But, since hypoxia and anoxia in the field often occurs as repeated events of different durations, future studies should consider the resilience of benthic organisms, nutrient fluxes and enzyme activities to repeated perturbations.

A clear and consistent pattern of enzyme activities between treatments (Figure 3.13) were found. This suggests that enzyme activities could be used as a

reliable indicator of ecological effects in sediment in place of nutrient fluxes. Incorporation of DNA based techniques would also be useful to allow more exact identification of the aerobic and anaerobic microorganism responsible and the sources of these enzymes in terms of reaching a better management of estuarine ecosystem. For example, by identifying chitin decomposing bacteria for chitinase activity and measuring urea to better understand the pattern observed for urease activity.

This study provides an interesting approach to determine what processes govern the sediment-water interface by using bromide as a tracer (section 3.3). But, the number of worms in each core should be included to further understand what is happening in each treatment. The worms were not counted with the aim to prevent disruption of the vertical sediment profiles of nutrient concentration and a range of biogeochemical processes in the sediment when counting the animals at the same time. A further study should consider using the conservative approach by adding an increasing number of individual to intact sediment cores to assess the effect of anoxia on biogeochemical processes at different densities of the polychaete worm in estuaries.

So, overall, the research between the link of oxygen and sediment biogeochemistry is very important from the perspective of ecosystem functions by providing ecotechnology measures using oxygen to overcome estuarine ecosystem imbalance. A preliminary threshold of the severity of short term effects was also examined. In this viewpoint, the research questions explored in this thesis may offer beneficial values in terms of ecosystem management. Hypoxia and anoxia up to 7 days alter the oxygen penetration depth, bromide fluxes and enhanced sediment effluxes of ammonium and phosphate but short term anoxia up to 24 hours has only mild biogeochemical effect. Therefore, the duration of hypoxic and anoxic events on sediment biogeochemistry should be incorporated as an important factor influencing estuarine ecosystem studies if we are to further our understanding and predictive capabilities of the response of estuarine ecosystem to perturbations such as eutrophication or climate change.

#### REFERENCES

- Abell, G. C. J., Banks, J., Ross, D. J., Keane, J. P., Robert, S. S., Revill, A. T. and Volkman, J. K. (2011). Effects of estuarine sediment hypoxia on nitrogen fluxes and ammonia oxidizer gene transcription. FEMS Microbiology Ecology 75(1): 111-122.
- Abril, G, H. Etcheber, B. Delille, M. Frankignoulle, A.V. Borges. (2003). Carbonate dissolution in the turbid and eutrophic Loire estuary. Marine Ecology Progress Series 259:129-138.
- Alef, K. and Nannipieri, P. (1995). Enzyme activities. Methods in Applied Soil Microbiology and Biochemistry. London, Academic Press: 311-373.
- Allison, S. D., Gartner, T. B., Holland, K., Weintraub, M. and Sinsabaugh, R. L. (2007). Soil enzymes: linking proteomics and ecological process, ASM Press.
- Alongi, D. M., Pfitzner, J., Trott, L. A., Tirendi, F., Dixon, P. and Klumpp, D. W. (2005). Rapid sediment accumulation and microbial mineralization in forests of the mangrove *Kandelia candel* in the Jiulongjiang Estuary, China. Estuarine, Coastal and Shelf Science 63(4): 605-618.
- Al-Rousan, S., M. Rasheed and M. Badran (2004): Nutrient regeneration from sediments and interstitial waters from the northern Gulf of Aqaba. *Scientia Marina.*, 68, 483–490.
- Arnosti, C. (2011). Microbial Extracellular Enzymes and the Marine Carbon Cycle. Annual Review of Marine Science **3**(1): 401-425.
- Arnosti, C. and Holmer, M. (2003). Carbon cycling in a continental margin sediment: contrasts between organic matter characteristics and remineralisation rates and pathways. Estuarine, Coastal and Shelf Science **58**(1): 197-208.
- Arnosti, C., Ziervogel, K., Ocampo, L. and Ghobrial, S. (2009). Enzyme activities in the water column and in shallow permeable sediments from the northeastern Gulf of Mexico. Estuarine, Coastal and Shelf Science 84(2): 202-208.
- Ayyakkannu, K. and Chandramohan, D. (1971). Occurrence and distribution of phosphate solubilizing bacteria and phosphatase in marine sediments at Porto Novo. Marine Biology 11(3): 201-205.
- Azua, I., Unanue, M., Ayo, B., Artolozaga, I., Arrieta, J. M. and Iriberri, J. (2003). Influence of organic matter quality in the cleavage of polymers by marine bacterial communities. Journal of Plankton Research 25(12): 1451-1460.
- Baban, S. M. J. (1997). Environmental Monitoring of Estuaries; Estimating and Mapping Various Environmental Indicators in Breydon Water Estuary, U.K., Using Landsat TM Imagery. Estuar. Coast Shelf Sci. 44: 589-598.
- Bartoli, M., Vezzulli, L., Nizzoli, D., Azzoni, R., Porrello, S., Moreno, M., Fabiano, M. and Viaroli, P. (2009). Short-term effect of oxic to anoxic transition on benthic microbial activity and solute fluxes in organic-rich phytotreatment ponds. Hydrobiologia 629(1): 123-136.
- Beier, S. (2010). Bacterial Degradation and Use of Chitin in Aquatic Habitats. Uppsala University.
- Beier, S. and Bertilsson, S. (2011). Uncoupling of chitinase activity and uptake of hydrolysis products in freshwater bacterioplankton. Limnol. Oceanogr. 56(4): 1179-1188.
- Belias, C., Dassenakis, M. and Scoullos, M. (2007). Study of the N, P and Si fluxes between fish farm sediment and seawater. Results of simulation experiments
employing a benthic chamber under various redox conditions. Marine Chemistry **103**(3-4): 266-275.

- Benoit P., Gratton, Y. and Mucci, A. (2006). Modeling of dissolved oxygen levels in the bottom waters of the Lower St. Lawrence Estuary: coupling of benthic and pelagic processes. Marine Chemistry **102**:13-32.
- Berg, P., Rysgaard, S., Funch, P. and Sejr, M. K. (2001). Effects of bioturbation on solutes and solids in marine sediments. Aquatic Microbial Ecology 26(1): 81-94.
- Berman, T. and Bronk, D. A. (2003). Dissolved organic nitrogen: a dynamic participant in aquatic ecosystems Aquatic Microbial Ecology **31**(3): 279-305.
- Berner, R. A. (1980). Early Diagenesis: A Theoretical Approach, Princeton University Press.
- Bhaskar, P. V. and Bhosle, N. B. (2008). Bacterial production, glucosidase activity and particle-associated carbohydrates in Dona Paula bay, west coast of India. Estuarine, Coastal and Shelf Science **80**(3): 413-424.
- Bhatia, Y., Mishra, S. and Bisaria, V. S. (2002). Microbial β-Glucosidases: Cloning, Properties, and Applications. Critical Reviews in Biotechnology 22(4): 375-407.
- Bianchi, T. S., DiMarco, S. F., Cowan Jr, J. H., Hetland, R. D., Chapman, P., Day, J.W. and Allison, M. A. (2010). The science of hypoxia in the Northern Gulf of Mexico: A review. Science of The Total Environment 408(7): 1471-1484.
- Bianchi, T. S. (2007). Biogeochemistry of Estuaries, Oxford University Press.
- Bianchi, T. S., Filley, T., Dria, K. and Hatcher, P. G. (2004). Temporal variability in sources of dissolved organic carbon in the lower Mississippi river. Geochimica et Cosmochimica Acta **68**(5): 959-967.
- Bianchi, T. S., Johansson, B. and Elmgren, R. (2000). Breakdown of phytoplankton pigments in Baltic sediments: effects of anoxia and loss of deposit-feeding macrofauna. Journal of experimental marine biology and ecology 251(2): 161-183.
- Boetius, A. (1995). Microbial hydrolytic enzyme activities in deep-sea sediments. Helgoland Marine Research **49**(1): 177-187.
- Boetius, A., Ferdelman, T. and Lochte, K. (2000). Bacterial activity in sediments of the deep Arabian Sea in relation to vertical flux. Deep Sea Research Part II: Topical Studies in Oceanography **47**(14): 2835-2875.
- Boudreau, B. P. (1997). Diagenetic Models and Their Implementation. Modelling Transport and Reactions in Aquatic Sediments. Berlin, Heidelberg, New York, London, Paris, Tokyo, Hong Kong:, Springer-Verlag.
- Boudreau, B. P. and Meysman, F. J. R. (2006). Predicted tortuosity of muds. Geology **34**(8): 693-696.
- Bowman, J. P., McCammon, S. A., Gibson, J. A. E., Robertson, L. and Nichols, P. D. (2003). Prokaryotic Metabolic Activity and Community Structure in Antarctic Continental Shelf Sediments. Applied and Environmental Microbiology 69(5): 2448-2462.
- Bradshaw, C., Kumblad, L. and Fagrell, A. (2006). The use of tracers to evaluate the importance of bioturbation in remobilising contaminants in Baltic sediments. Estuarine, Coastal and Shelf Science **66**(1-2): 123-134.
- Braga, E. D. S. and Torres, J. L. R. (2010). The Role Of Urea In The Coastal Nitrogen Biogeochemical Cycle As A Sign Of Environmental Health. Safety, Health and Environment World Congress, São Paulo, Brazil.

- Bremner, J. M. and Mulvaney, R. L. (1978). Urease activity in soils. Soil Enzymes. R. G. Burns. New York, Academic Press: 149-196.
- Burdige, D. J. (2007). Preservation of organic matter in marine sediments: controls, mechanisms, and an imbalance in sediment organic carbon budgets? Chemical Reviews-Columbus 107(2): 467-485.(Bianchi, Johansson et al. 2000)
- Burdige, D. J. (2011). Temperature dependence of organic matter remineralization in deeply-buried marine sediments. Earth and Planetary Science Letters **311**(3-4): 396-410.
- Cai, W.-J., Hu, X., Huang, W.-J., Murrell, M. C., Lehrter, J. C., Lohrenz, S. E., Chou, W.-C., Zhai, W., Hollibaugh, J. T., Wang, Y., Zhao, P., Guo, X., Gundersen, K., Dai, M. and Gong, G.-C. (2011). Acidification of subsurface coastal waters enhanced by eutrophication. Nature Geosci 4(11): 766-770.
- Cai, W.-J. and Sayles, F. L. (1996). Oxygen penetration depths and fluxes in marine sediments. Marine Chemistry **52**(2): 123-131.
- Caffrey, J. M., Hollibaugh, J. T., Bano, N. and Haskins, J. (2010). Effects of upwelling on short-term variability in microbial and biogeochemical processes in estuarine sediments from Elkhorn Slough, California, USA. Aquatic Microbial Ecology 58(3): 261-271.
- Caldwell, B. A. (2005). Enzyme activities as a component of soil biodiversity: A review. Pedobiologia **49**: 637-644.
- Camargo, J. A. and Alonso, Ã. I. (2006). Ecological and toxicological effects of inorganic nitrogen pollution in aquatic ecosystems: A global assessment. Environment International **32**(6): 831-849.
- Canfield, D. E., Glazer, A. N., and Falkowski, P. G. (2010). The evolution and future of Earth's nitrogen cycle. Science, **330**(6001):192-196.
- Caruso, G. (2010). Leucine Aminopeptidase,  $\beta$ -Glucosidase and Alkaline Phosphatase Activity Rates and Their Significance in Nutrient Cycles in Some Coastal Mediterranean Sites. Marine Drugs **8**(4): 916-940.
- Cavari, B. Z. and Hadas, O. (1979). Heterotrophic activity, glucose uptake and primary productivity in Lake Kinneret. Freshwater Biology **9**(4): 329-338.
- Chan, F., Barth, J. A., Lubchenco, J., Kirincich, A., Weeks, H., Peterson, W. T. and Menge, B. A. (2008). Emergence of Anoxia in the California Current Large Marine Ecosystem. Science **319**(5865): 920.
- Charles, S. (2004). The Estuarine Ecosystem: Ecology, Threats and Management. Marine Pollution Bulletin. D. S. McLusky and M. Elliott, Oxford University Press. 49: 879.
- Chen, J., Lu, S., Zhao, Y., Wang, W. and Huang, M. (2011). Effects of overlying water aeration on phosphorus fractions and alkaline phosphatase activity in surface sediment. Journal of Environmental Sciences **23**(2): 206-21.
- Church, T. M. (1975). Marine Chemistry in the Coastal Environment. American Chemical Society.
- Clavero, V., Izquierdo, J. J., Fernández, J. A. and Niell, F. X. (2000). Seasonal fluxes of phosphate and ammonium across the sediment-water interface in a shallow small estuary (Palmones River, southern Spain). Marine Ecology Progress Series 198: 51-60.
- Conley, D. J., Björck, S., Bonsdorff, E., Carstensen, J., Destouni, G., Gustafsson, B. G., Hietanen, S., Kortekaas, M., Kuosa, H., Markus Meier, H. E., Müller-Karulis, B., Nordberg, K., Norkko, A., Nürnberg, G., Pitkänen, H., Rabalais,

N. N., Rosenberg, R., Savchuk, O. P., Slomp, C. P., Voss, M., Wulff, F. and Zillén, L. (2009). Hypoxia-Related Processes in the Baltic Sea. Environmental Science & Technology **43**(10): 3412-3420.

- Conley, D. J., Carstensen, J., Ærtebjerg, G., Christensen, P. B., Dalsgaard, T., Hansen, J. L. S. and Josefson, A. B. (2007). Long-Term Changes And Impacts Of Hypoxia In Danish Coastal Waters. Ecological Applications 17(sp5): \$165-\$184.
- Coolen, M. J. L. and Overmann, J. (2000). Functional Exoenzymes as Indicators of Metabolically Active Bacteria in 124,000-Year-Old Sapropel Layers of the Eastern Mediterranean Sea. Applied and Environmental Microbiology 66(6): 2589-2598.
- Cotner, J., Sada, R., Bootsma, H., Johengen, T., Cavaletto, J. and Gardner, W. (2000). Nutrient limitation of heterotrophic bacteria in Florida Bay. Estuaries and Coasts **23**(5): 611-620.
- Cowie, G. L. and Hedges, J. I. (1994). Biochemical indicators of diagenetic alteration in natural organic matter mixtures. Nature **369**(6478): 304-307.
- Cunha, A., Almeida, A., Coelho, F. J. R. C., Gomes, N. C. M., Oliveira, V. and Santos, A. L. (2010). Bacterial Extracellular Enzymatic Activity in Globally Changing Aquatic Ecosystems. Current Research, Technology and Education Topics in Applied Microbiology and Microbial Biotechnology. A. M. Vilas, FORMATEX.
- Cutts, N.D., Murby, P. and Hemingway, K.L. (2008). Habitat Loss in the Humber Estuary, UK. Institute of Estuarine & Coastal Studies (IECS), University of Hull, UK. Report produced as part of the European Interreg IIIB HARBASINS project.
- D'Avanzo, C. and Kremer, J. (1994). Diel oxygen dynamics and anoxic events in an eutrophic estuary of Waquoit Bay, Massachusetts. Estuaries and Coasts **17**(1): 131-139.
- Dai, M., Guo, X., Zhai, W., Yuan, L., Wang, B., Wang, L., Cai, P., Tang, T. and Cai, W.-J. (2006). Oxygen depletion in the upper reach of the Pearl River estuary during a winter drought. Marine Chemistry 102(1-2): 159-169.
- Dalsgaard, T., Nielsen, L. P., Brotas, V., Viaroli, P., Underwood, G., Nedwell, D., Sundbäck, K., Rysgaard, S., Miles, A., Bartoli, M., Dong, L., Thornton, D. C. O., Ottosen, L. D. M., Castaldelli, G. and Risgaard-Petersen, N. (2000). Protocol handbook for NICE - Nitrogen Cycling in Estuaries A project under the EU research programme: Marine Science and Technology (MAST III). T. Dalsgaard. Denmark, National Environmental Research Institute: 62.
- Danovaro, R., Armeni, M., Dell'Anno, A., Fabiano, M., Manini, E., Marrale, D., Pusceddu, A. and Vanucci, S. (2001). Small-Scale Distribution of Bacteria, Enzymatic Activities, and Organic Matter in Coastal Sediments. Microbial Ecology 42(2): 177-185.
- De, T. K. Sarkar, T. K. De, M. Maity, T.K. Mukherjee, A. and Das, S. (2012). Abundance and occurrence of phosphate solubilizing bacteria and phosphatase in Sediment of Hooghly estuary, north east coast of Bay of Bengal, India. Journal of Coastal Development. 15. pp. 9-16.
- De Cesare, F., Garzillo, A. M. V., Buonocore, V. and Badalucco, L. (2000). Use of sonication for measuring acid phosphatase activity in soil. Soil Biology and Biochemistry **32**(6): 825-832.

- DeForest, J. L. (2009). The influence of time, storage temperature, and substrate age on potential soil enzyme activity in acidic forest soils using MUB-linked substrates and I-DOPA. Soil Biology and Biochemistry **41**(6): 1180-1186.
- Degobbis, D., Gilmartin, M. and Orio, A. A. (1986). The relation of regeneration in the sediments of the northern adriatic to eutrophication, with special reference to the lagoon of Venice. Science of The Total Environment **56**(0): 201-210.
- Deng, J.-j., Huang, X.-f., Hu, J.-w., Li, C.-x., Yi, Y. and Long, J. (2009). Distribution of several microorganisms and activity of alkaline phosphatase in sediments from Baihua Lake. Asia-Pacific Journal of Chemical Engineering 4(5): 711-716.
- Denis, L. and Grenz, C. (2003). Spatial variability in oxygen and nutrient fluxes at the sediment-water interface on the continental shelf in the Gulf of Lions (NW Mediterranean). Oceanologica Acta **26**(4): 373-389.
- Diaz, R. J. and Rosenberg, R. (2008). Spreading Dead Zones and Consequences for Marine Ecosystems. Science **321**(5891): 926-929.
- Doney, S. C. (2010). The Growing Human Footprint on Coastal and Open-Ocean Biogeochemistry. Science **328**(5985): 1512-1516.
- Donderski, W. and Trzebiatowska, M. (1999). Chitinase Activity Production by Planktonic, Benthic and Epiphytic Bacteria Inhabiting the Moty Bay of the Jeziorak Lake (Poland). Polish Journal of Environmental Studies **8**(4): 215-220.
- Dong, L. F., Smith, C. J., Papaspyrou, S., Stott, A., Osborn, A. M. and Nedwell, D. B. (2009). Changes in Benthic Denitrification, Nitrate Ammonification, and Anammox Process Rates and Nitrate and Nitrite Reductase Gene Abundances along an Estuarine Nutrient Gradient (the Colne Estuary, United Kingdom). Applied and Environmental Microbiology **75**(10): 3171-3179.
- Dornhoffer, T., Waldbusser, G. G. and Meile, C. (2012). Burrow patchiness and oxygen fluxes in bioirrigated sediments. Journal of Experimental Marine Biology and Ecology **412**: 81-86.
- Duddridge, J. E. and Wainwright, M. (1982). Enzyme activity and kinetics in substrate-amended river sediments. Water Research **16**(3): 329-334.
- Dumont, H.J. (1998). The Caspian Lake: History, biota, structure, and function. Limnology and Oceanography **43**:44-52.
- Emeis, K. C., Struck, U., Leipe, T., Pollehne, F., Kunzendorf, H. and Christiansen, C. (2000). Changes in the C, N, P burial rates in some Baltic Sea sediments over the last 150 years-relevance to P regeneration rates and the phosphorus cycle. Marine Geology 167(1-2): 43-59.
- Fabiano, M. and Danovaro, R. (1998). Enzymatic Activity, Bacterial Distribution, and Organic Matter Composition in Sediments of the Ross Sea (Antarctica). Applied and Environmental Microbiology **64**(10): 3838-3845.
- Fernandes, M., Mount, G. and Tanner, J. (2007). Source of organic matter in sediments from Louth Bay, South Australia. S. A. S. Publication.
- Field, A. (2009). Discovering Statistics using SPSS, SAGE Publications Inc.
- Forster, S., Glud, R. N., Gundersen, J. K. and Huettel, M. (1999). *In situ* Study of Bromide Tracer and Oxygen Flux in Coastal Sediments. Estuarine, Coastal and Shelf Science **49**(6): 813-827.
- Forster, S., Graf, G., Kitlar, J. and Powilleit, M. (1995). Effects of bioturbation in oxic and hypoxic conditions: a microcosm experiment with a North Sea sediment community. Marine Ecology Progress Series **116**(153-161).

- Freeman, C., Liska, G., Ostle, N. J., Lock, M. A., Hughes, S., Reynolds, B. and Hudson, J. (1997). Enzymes and biogeochemical cycling in wetlands during a simulated drought. Biogeochemistry **39**(2): 177-187.
- Galloway, J. N., Dentener, F. J., Capone, D. G., Boyer, E. W., Howarth, R. W., Seitzinger, S. P., Asner, G. P., Cleveland, C. C., Green, P. A., Holland, E. A., Karl, D. M., Michaels, A. F., Porter, J. H., Townsend, A. R. and Vöosmarty, C. J. (2004). Nitrogen Cycles: Past, Present, and Future. Biogeochemistry 70(2): 153-226.
- Gamenick, I., Jahn, A., Vopel, K. and Giere, O. (1996). Hypoxia and sulphide as structuring factors in a macrozoobenthic community on the Baltic Sea shore: colonisation studies and tolerance experiments. Marine Ecology Progress Series 144: 73-85.
- German, D. P., Marcelo, K. R. B., Stone, M. M. and Allison, S. D. (2011). The Michaelis–Menten kinetics of soil extracellular enzymes in response to temperature: a cross-latitudinal study. Global Change Biology 18(4): 1468-1479.
- Gianfreda, L., Antonietta Rao, M., Piotrowska, A., Palumbo, G. and Colombo, C. (2005). Soil enzyme activities as affected by anthropogenic alterations: intensive agricultural practices and organic pollution. Science of The Total Environment **341**(1-3): 265-279.
- Gilbert, D., Rabalais, N. N., Diaz, R. J. and Zhang, J. (2010). Evidence for greater oxygen decline rates in the coastal ocean than in the open ocean. Biogeosciences **7**(7): 2283-2296.
- Giles, H., Pilditch, C. A., Nodder, S. D., Zeldis, J. R. and Currie, K. (2007). Benthic oxygen fluxes and sediment properties on the northeastern New Zealand continental shelf. Continental Shelf Research **27**(18): 2373-2388.
- Gillespie, D., Ed. (2005). Sediment, Water and Porewater Sampling. A System of Hierarchical Monitoring Methods for assessing changes in the biological and physical state of intertidal areas: Book of Protocols, HIMOM.
- Glud, R. N. (2008). Oxygen dynamics of marine sediments. Marine Biology Research, Taylor & Francis. **4:** 243-289.
- Glud, R. N., Berg, P., Hume, A., Batty, P., Blicher, M.E., Lennert, K. and Rysgaard, S. (2010). Benthic O2 exchange across hard-bottom substrates quantified by eddy correlation in a sub-Arctic fjord. Marine Ecology Progress Series 417: 1-12.
- Glud, R., Gundersen, J. K., Barker Jorgensen, B., Revsbech, N. P. and Schulz, H. D. (1994). Diffusive and total oxygen uptake of deep-sea sediments in the eastern South Atlantic Ocean:*in situ* and laboratory measurements. Deep Sea Research Part I: Oceanographic Research Papers **41**(11-12): 1767-1788.
- Glud, R. N., Ramsing, N. B., Gundersen, J. K. and Klimant, I. (1996). Planar optrodes: a new tool for fine scale measurements of two-dimensional O2 distribution in benthic communities. Marine Ecology Progress Series 140: 217-226.
- Gluyas, J. and Coleman, M. (1992). Material flux and porosity changes during sediment diagenesis. Nature **356**(6364): 52-54.
- Golterman, H. L. (2001). Phosphate release from anoxic sediments or `What did Mortimer really write?'. Hydrobiologia **450**(1): 99-106.
- Gooday, G. W. (1990). Physiology of microbial degradation of chitin and chitosan. Biodegradation 1(2): 177-190.

- Goodrich, T. D. and Morita, R. Y. (1977). Bacterial chitinase in the stomachs of marine fishes from Yaquina Bay, Oregon, USA. Marine Biology **41**(4): 355-360.
- Grandy, A. S., Neff, J. C. and Weintraub, M. N. (2007). Carbon structure and enzyme activities in alpine and forest ecosystems. Soil Biology and Biochemistry **39**(11): 2701-2711.
- Gray, J. S., Wu, R. S.-S. and Or, Y. Y. (2002). Effects of hypoxia and organic enrichment on the coastal marine environment. Marine Ecology Progress Series **238**: 249-279.
- Green, M. A., Aller, R. C., Cochran, J. K., Lee, C. and Aller, J. Y. (2002). Bioturbation in shelf/slope sediments off Cape Hatteras, North Carolina: the use of 234Th, Chl-a, and Br<sup>-</sup> to evaluate rates of particle and solute transport. Deep Sea Research Part II: Topical Studies in Oceanography 49(20): 4627-4644.
- Greenwood, N., Parker, E. R., Fernand, L., Sivyer, D. B., Weston, K., Painting, S. J., Kroeger, S., Forster, R. M., Lees, H. E., Mills, D. K. and Laane, R. W. P. M. (2010). Detection of low bottom water oxygen concentrations in the North Sea; implications for monitoring and assessment of ecosystem health. Biogeosciences 7(4): 1357-1373.
- Haag, I., Schmid, G. and Westrich, B. (2005). *In situ* measurement and simulation of dissolved oxygen and nutrient transfer across sediment-water interface of River Neckar, Germany. RMZ-Materials and Geoenvironment 52(1): 51-54.
- Hagy, J., Boynton, W., Keefe, C. and Wood, K. (2004). Hypoxia in Chesapeake Bay, 1950–2001: Long-term change in relation to nutrient loading and river flow. Estuaries and Coasts **27**(4): 634-658.
- Hakulinen, R., Kähkönen, M. A. and Salkinoja-Salonen, M. (2005). Vertical distribution of sediment enzyme activities involved in the cycling of carbon, nitrogen, phosphorus and sulphur in three boreal rural lakes. Water Research 39(11): 2319-2326.
- Hammond, D. E., Cummins, K. M., McManus, J., Berelson, W. M., Smith, G. and Spagnoli, F. (2004). Methods for measuring benthic nutrient flux on the California Margin: Comparing shipboard core incubations to *in situ* lander results. Limnol. Oceanogr. 2: 146-159.
- Hansen, K. and Kristensen, E. (1998). The impact of the polychaete *Nereis diversicolor* and enrichment with macroalgal (*Chaetomorpha linum*) detritus on benthic metabolism and nutrient dynamics in organic-poor and organic-rich sediment. Journal of Experimental Marine Biology and Ecology **231**(2): 201-223.
- Hearn, C. J. and Robson, B. J. (2001). Inter-annual Variability of Bottom Hypoxia in Shallow Mediterranean Estuaries. Estuarine, Coastal and Shelf Science **52**(5): 643-657.
- Hedges, J. I., Baldock, J. A., Gelinas, Y., Lee, C., Peterson, M. and Wakeham, S. G. (2001). Evidence for non-selective preservation of organic matter in sinking marine particles. Nature 409(6822): 801-804.
- Hedges, J. I. and Keil, R. G. (1995). Sedimentary organic matter preservation: an assessment and speculative synthesis. Marine Chemistry **49**(2-3): 81-115.
- Hedman, J. E., Gunnarsson, J. S., Samuelsson, G. and Gilbert, F. (2011). Particle reworking and solute transport by the sediment-living polychaetes

*Marenzelleria neglecta* and *Hediste diversicolor*. Journal of Experimental Marine Biology and Ecology **407**(2): 294-301.

- Heilskov, A. C. and Holmer, M. (2001). Effects of benthic fauna on organic matter mineralization in fish-farm sediments: importance of size and abundance. ICES Journal of Marine Science: Journal du Conseil 58(2): 427-434.
- Helly, J. J. and Levin, L. A. (2004). Global distribution of naturally occurring marine hypoxia on continental margins. Deep Sea Research Part I: Oceanographic Research Papers 51(9): 1159-1168.
- Herbert, R. A. (1999). Nitrogen cycling in coastal marine ecosystems. FEMS Microbiology Reviews 23(5): 563-590.
- Hietanen, S. and Lukkari, K. (2007). Effects of short-term anoxia on benthic denitrification, nutrient fluxes and phosphorus forms in coastal Baltic sediment. Aquatic Microbial Ecology **49**(3): 293-302.
- Hietanen, S., Jäntti, H., Buizert, C., Labrenz, M. and Voss, M. (2012). Hypoxia and nitrogen processing in the Baltic Sea water column. Limnology and Oceanography 57(1): 325.
- Hill, B. H., Elonen, C. M., Seifert, L. R., May, A. A. and Tarquinio, E. (2012). Microbial enzyme stoichiometry and nutrient limitation in US streams and rivers. Ecological Indicators 18(0): 540-551.
- Hill, B., Elonen, C., Jicha, T., Bolgrien, D. and Moffett, M. (2006). Sediment microbial enzyme activity as an indicator of nutrient limitation in the great rivers of the Upper Mississippi River basin. Biogeochemistry 97(2): 195-209.
- Hillman, K., Gooday, G. W. and Prosser, J. I. (1989). The mineralization of chitin in the sediments of the Ythan estuary, Aberdeenshire, Scotland. Estuarine, Coastal and Shelf Science **29**(6): 601-612.
- Hinga, K. R. (2002). Effects of pH on coastal marine phytoplankton. Marine Ecology Progress Series **238**: 281-300.
- Hobbie, J. E. (2000). Estuarine science: a synthetic approach to research and practice. Washington Island Press.
- Hoppe, H.-G. (1993). Use of Fluorogenic Model Substrates for Extracellular Enzyme Activity (EEA) Measurement of Bacteria. Handbook of Methods in Aquatic Microbial Ecology. New York, Springer-Verlag.
- Hoppe, H.-G. (2003). Phosphatase activity in the sea. Hydrobiologia **493**(1): 187-200.
- Hoppe, H.-G., Carol, A. and Gerhard, H. (2002). Ecological Significance of Bacterial Enzymes in the Marine Environment. Enzymes in the Environment, CRC Press.
- Howarth, R., Chan, F., Conley, D. J., Garnier, J., Doney, S. C., Marino, R. and Billen, G. (2011). Coupled biogeochemical cycles: eutrophication and hypoxia in temperate estuaries and coastal marine ecosystems. Frontiers in Ecology and the Environment 9(1): 18-26.
- Hulth, S., Aller, R. C., Canfield, D. E., Dalsgaard, T., Engström, P., Gilbert, F., Sundbäck, K. and Thamdrup, B. (2005). Nitrogen removal in marine environments: recent findings and future research challenges. Marine Chemistry 94(1-4): 125-145.
- Hulth, S., Blackburn, H.T. and Hall, P.O.J. (1994). Arctic sediments (Svalbard): consumption and microdistribution of oxygen.Marine Chemistry **46**(3): 293-316.

- Hupfer, M. and Lewandowski, J. (2008). Oxygen Controls the Phosphorus Release from Lake Sediments–a Long-Lasting Paradigm in Limnology. International Review of Hydrobiology 93(4-5): 415-432.
- Ingalls, A. E., Aller, R. C., Lee, C. and Wakeham, S. G. (2004). Organic matter diagenesis in shallow water carbonate sediments. Geochimica et Cosmochimica Acta **68**(21): 4363-4379.
- Jäntti, H. and Hietanen, S. (2012). The Effects of Hypoxia on Sediment Nitrogen Cycling in the Baltic Sea. AMBIO **41**(2): 161-169.
- Jenkins, C. (2005). Nutrient flux assessment in the Port waterways. Adelaide, S. Aust., Environment Protection Authority.
- Jetten, M. S. M. (2008). The microbial nitrogen cycle. Environmental microbiology **10**(11): 2903-2909.
- Jjemba, P. K. (2004). Environmental microbiology: principles and applications, Science.
- Jiang, C., Hu, J., Huang, X., Li, C., Deng, J., Zhang, J. and Liu, F. (2011). Phosphorus speciation in sediments of Lake Hongfeng, China. Chinese Journal of Oceanology and Limnology 29(1): 53-62.
- Jiang, X., Jin, X., Yao, Y., Li, L. and Wu, F. (2006). Effects of oxygen on the release and distribution of phosphorus in the sediments under the light condition. Environmental Pollution **141**(3): 482-487.
- Jonasson, L., Hansen, J. L. S., Wan, Z. and She, J. (2012). The impacts of physical processes on oxygen variations in the North Sea-Baltic Sea transition zone. Ocean Science **8**(1): 37-48.
- Jones, P. D. (2006). Water quality and fisheries in the Mersey estuary, England: A historical perspective. Marine Pollution Bulletin **53**(1-4): 144-154.
- Jones, R. D. (2002). Phosphorus cycling. Manual of environmental microbiology. C. J. Hurst. Washington, D.C., ASM Press: 450, 453.
- Jordan, T. E., Cornwell, J. C., Boynton, W. R. and Anderson, J. T. (2008). Changes in phosphorus biogeochemistry along an estuarine salinity gradient: The iron conveyer belt. Limnol. Oceanogr. **53**(1): 172-184.
- Jorgensen, N. O. G. (2006). Uptake of urea by estuarine bacteria. Aquatic Microbial Ecology **42**(3): 227-242.
- Kähkönen, M. A. and Hakulinen, R. (2012). Stratification of selected hydrolytic enzyme activities in the sediment in two lakes in Finland. International Journal of Sediment Research **27**(2): 250-258.
- Kandeler, E., Stemmer, M. and Klimanek, E.-M. (1999). Response of soil microbial biomass, urease and xylanase within particle size fractions to long-term soil management. Soil Biology and Biochemistry **31**(2): 261-273.
- Kang, H., Kim, S.-Y., Fenner, N. and Freeman, C. (2005). Shifts of soil enzyme activities in wetlands exposed to elevated CO2. Science of the Total Environment **337**(1-3): 207-212.
- Karlson, K., Hulth, S. and Rosenberg, R. (2007). Density of *Monoporeia affinis* and biogeochemistry in Baltic Sea sediments. Journal of Experimental Marine Biology and Ecology 344(2): 123-135.
- Katsev, S., Chaillou, G., Sundby, B. and Mucci, A. (2007). Effects of progressive oxygen depletion on sediment diagenesis and fluxes: A model for the lower St. Lawrence River Estuary. Limnol. Oceanogr. 52(6): 2555-2568.
- Kazemi, H., Morris, J., Anderson, S., Gantzer, C. and Buyanovsky, G. (1998). Measurement of bromide ion used as a solute-transport monitor via

epithermal neutron activation analysis. Journal of Radioanalytical and Nuclear Chemistry **235**(1): 249-254.

- Keefe, S. H., Barber, L. B., Runkel, R. L., Ryan, J. N., McKnight, D. M. and Wass, R. D. (2004). Conservative and reactive solute transport in constructed wetlands. Water Resour. Res. 40(1): 12.
- Kelley, A. M., Fay, P. A., Polley, H. W., Gill, R. A. and Jackson, R. B. (2011). Atmospheric CO2 and soil extracellular enzyme activity: a meta-analysis and CO2 gradient experiment. Ecosphere 2(8): 1-20.
- Kelly-Gerreyn, B. A., Hydes, D. J. and Waniek, J. J. (2005). Control of the diffusive boundary layer on benthic fluxes: a model study. Marine Ecology Progress Series 292: 61-74.
- Kemp, W. M., Testa, J. M., Conley, D. J., Gilbert, D. and Hagy, J. D. (2009). Temporal responses of coastal hypoxia to nutrient loading and physical controls. Biogeosciences 6(12): 2985-3008.
- Kemp, W. M., Boynton, W. R., Adolf, J. E., Boesch, D. F., Boicourt, W. C., Brush, G., Cornwell, J. C., Fisher, T. R., Glibert, P. M., Hagy, J. D., Harding, L. W., Houde, E. D., Kimmel, D. G., Miller, W. D., Newell, R. I. E., Roman, M. R., Smith, E. M. and Stevenson, J. C. (2005). Eutrophication of Chesapeake Bay: historical trends and ecological interactions. Marine Ecology Progress Series 303: 1-29.
- Keyhani, N. O. and Roseman, S. (1999). Physiological aspects of chitin catabolism in marine bacteria. Biochimica et Biophysica Acta **1473**(1): 108-122.
- Kim, K. and Kim, D. (2007). Seasonal and spatial variability of sediment oxygen fluxes in the Beobsan intertidal flat of Taean Bay, mid-western Korean Peninsula. Geosciences Journal **11**(4): 323-329.
- King, G. M. (1986). Characterization of β-Glucosidase Activity in Intertidal Marine Sediments. Applied and Environmental Microbiology **51**(2): 373-380.
- Kirchman, D. L. and White, J. (1999). Hydrolysis and mineralization of chitin in the Delaware Estuary. Aquatic Microbial Ecology **18**(2): 187-196.
- Koch, M. S., Kletou, D. C. and Tursi, R. (2009). Alkaline phosphatase activity of water column fractions and seagrass in a tropical carbonate estuary, Florida Bay. Estuarine, Coastal and Shelf Science 83(4): 403-413.
- Kodama K., T. Horiguchi, G. Kume, S. Nagayama, T. Shimizu, H. Shiraishi, M. Morita, M. Shimizu. (2006). Effects of hypoxia on early life history of the stomatopod *Oratosquilla oratoria* in a coastal sea. Marine Ecology Progress Series 324:197-206.
- Kollner, K. E., Carstens, D., Keller, E., Vazquez, F., Schubert, C. J., Zeyer, J. and Bürgmann, H. (2012). Bacterial Chitin Hydrolysis in Two Lakes with Contrasting Trophic Statuses. Applied and Environmental Microbiology 78(3): 695-704.
- Koster, M., Dahlke, S. and Meyer-Reil, L.-A. (1997). Microbiological studies along a gradient of eutrophication in a shallow coastal inlet in the southern Baltic Sea (Nordrugensche Bodden). Marine Ecology Progress Series **152**: 27-39.
- Kristensen, E. (2000). Organic matter diagenesis at the oxic/anoxic interface in coastal marine sediments, with emphasis on the role of burrowing animals. Hydrobiologia **426**(1): 1-24.
- Kristensen, K. and Hansen, K. (1999). Transport of carbon dioxide and ammonium in bioturbated (*Nereis diversicolor*) coastal, marine sediments. Biogeochemistry **45**(2): 147-168.

- Kristensen, E. and Blackburn, T. H. (1987). The fate of organic carbon and nitrogen in experimental marine sediment systems: Influence of bioturbation and anoxia. Journal of Marine Research **45**(1): 231-257.
- Langston, W. J., Chesman, B. S. and Burt, G. R. (2006). Characterisation of European Marine Sites. Mersey Estuary SPA. Marine Biological Association of the United Kingdom. Occasional Publications **18**: 185pp.
- Lansard, B., Rabouille, C. and Massias, D. (2003). Variability in benthic oxygen fluxes during the winter-spring transition in coastal sediments: an estimation by in situ micro-electrodes and laboratory mini-electrodes. Oceanologica Acta **26**(3): 269-279.
- Larson, F. and Sundback, K. (2008). Role of microphytobenthos in recovery of functions in a shallow-water sediment system after hypoxic events. Marine Ecology Progress Series **357**: 1-16.
- Lavery, P. S., Oldham, C. E. and Ghisalberti, M. (2001). The use of Fick's First Law for predicting porewater nutrient fluxes under diffusive conditions. Hydrological Processes **15**(13): 2435-2451.
- LeCleir, G. R. and Hollibaugh, J. T. (2006). Chitinolytic bacteria from alkaline hypersaline Mono Lake, California, USA. Aquatic Microbial Ecology **42**(3): 255-264.
- Lee, C. (1992). Controls on organic carbon preservation: The use of stratified water bodies to compare intrinsic rates of decomposition in oxic and anoxic systems. Geochimica et Cosmochimica Acta **56**(8): 3323-3335.
- Leloup, J., Loy, A., Knab, N. J., Borowski, C., Wagner, M. and Jørgensen, B. B. (2007). Diversity and abundance of sulfate-reducing microorganisms in the sulfate and methane zones of a marine sediment, Black Sea. Environmental microbiology **9**(1): 131-142.
- Lerat, Y., Lasserre, P. and Corre, P. l. (1990). Seasonal changes in pore water concentrations of nutrients and their diffusive fluxes at the sediment-water interface. Journal of Experimental Marine Biology and Ecology **135**(2): 135-160.
- Leschine, S. B. (1995). Cellulose Degradation in Anaerobic Environments. Annual Review of Microbiology **49**: 399-426.
- Lewis, B. L., Glazer, B. T., Montbriand, P. J., Luther III, G. W., Nuzzio, D. B., Deering, T., Ma, S. and Theberge, S. (2007). Short-term and interannual variability of redox-sensitive chemical parameters in hypoxic/anoxic bottom waters of the Chesapeake Bay. Marine Chemistry 105(3-4): 296-308.
- Li, Y. And Gregory, S. (1974). Diffusion of ions in sea water and in sediments. Geochim. Cosmochim. Acta 38:703-714.
- Libes, S. M. (2009). An introduction to marine biogeochemistry. London UK, Academic Press.
- Litchman, E. and Nguyen, B. L. V. (2008). Alkaline Phosphatase Activity As A Function Of Internal Phosphorus Concentration In Freshwater Phytoplankton. Journal of Phycology **44**(6): 1379-1383.
- Liu, H., Tian, Y., Zheng, T., Yan, C. and Hong, H. (2008). Studies of glucosidase activities from surface sediments in mangrove swamp. Journal of Experimental Marine Biology and Ecology 367: 111-117.
- Liu, S., Zhang, J. and Jiang, W. (2003). Pore Water Nutrient Regeneration in Shallow Coastal Bohai Sea, China. Journal of Oceanography **59**(3): 377-385.

- Lund, B. A. and Blackburn, T. H. (1989). Urea turnover in a coastal marine sediment sediment measured by a L<sub>4</sub>C-urea short term incub. Journal of Microbiological Methods **9**: 297-308.
- MacLellan, J. (2010). Innovative Strategies to Enhance Enzymatic Hydrolysis: A Review. MMG 445 Basic Biotechnology **6**(1): 31-35.
- Macreadie, P. I., Ross, D. J., Longmore, A. R. and Keough, M. J. (2006). Denitrification measurements of sediments using cores and chambers. Marine Ecology Progress Series 326: 49-59.
- Madsen, E. L. (2008). Environmental Microbiology: From Genomes to Biogeochemistry, John Wiley & Sons.
- Maes, J., Stevens, M. and Breine, J. (2007). Modelling the migration opportunities of diadromous fish species along a gradient of dissolved oxygen concentration in a European tidal watershed. Estuarine, Coastal and Shelf Science 75(1-2): 151-162.
- Maki, M., Leung, K. T. and Qin, W. (2009). The prospects of cellulase-producing bacteria for the bioconversion of lignocellulosic biomass. Int J Biol Sci **5**(5): 500-516.
- Maksymowska-Brossard, D. and Piekarek-Jankowska, H. (2001). Seasonal variability of benthic ammonium release in the surface sediments of the Gulf of Gdansk (southern Baltic Sea).Oceanologia, 43.
- Mallet, C., Basset, M., Fonty, G., Desvilettes, C., Bourdier, G. and Debroas, D. (2004). Microbial Population Dynamics in the Sediments of a Eutrophic Lake (Aydat, France) and Characterization of Some Heterotrophic Bacterial Isolates. Microbial Ecology 48(1): 66-77.
- Mallet, C. and Debroas, D. (2001). Regulation of  $\beta$ -and  $\alpha$ -glycolytic activities in the sediments of a eutrophic lake. Microbial Ecology **41**(2): 106-113.
- Manning, P., Saunders, M., Bardgett, R. D., Bonkowski, M., Bradford, M. A., Ellis, R. J., Kandeler, E., Marhan, S. and Tscherko, D. (2008). Direct and indirect effects of nitrogen deposition on litter decomposition. Soil Biology & Biochemistry 40: 688-698.
- Martin, W. R. and Banta, G. T. (1992). The measurement of sediment irrigation rates: A comparison of the Br tracer and 222Rn/226Ra disequilibrium techniques. Journal of Marine Research **50**(1): 125-154.
- Martínez-Crego, B., Romero, J. and Alcoverro, T. (2006). The use of surface alkaline phosphatase activity in the seagrass *Posidonia oceanica* as a biomarker of eutrophication. Marine Ecology **27**(4): 381-387.
- Marx, M. C., Wood, M. and Jarvis, S. C. (2001). A microplate fluorimetric assay for the study of enzyme diversity in soils. Soil Biology and Biochemistry **33**(12-13): 1633-1640.
- Matisof, G. and Wang, X. (1998). Solute transport in sediments by freshwater infaunal bioirrigators. Limnol. Oceanogr. **43**(7): 1487-14.
- Mayer, L. M. (1989). Extracellular proteolytic enzyme activity in sediments of an intertidal mudflat. Limnol. Oceanogr. **34**(6): 973-981.
- Mee, L.D. (2001). Eutrophication in the Black Sea and a basin wide approach to control it. In: Von Bodungen, B., Turner, R.K. Science and Integrated Coastal Management. Dahlem University Press, Berlin, pp. 71–91. Meile, C. and Cappellen, P. v. (2003). Global Estimates of Enhanced Solute Transport in Marine Sediments. Limnol. Oceanogr. 48(2).

- Meile, C., Berg, P., Cappellen, P. V. and Tuncay, K. (2005). Solute-specific pore water irrigation: Implications for chemical cycling in early diagenesis. Journal of Marine Research **63**(3): 601-621.
- Meile, C. and Cappellen, P. V. (2003). Global Estimates of Enhanced Solute Transport in Marine Sediments. Limnol. Oceanogr. **48**(2).
- Meile, C., Koretsky, C. M. and Cappellen, P. V. (2001). Quantifying Bioirrigation in Aquatic Sediments: An Inverse Modeling Approach. Limnol. Oceanogr. 46(1): 164-177.
- Meyers, P. A. (1994). Preservation of elemental and isotopic source identification of sedimentary organic matter. Chemical Geology **114**(3-4): 289-302.
- Meyers, P. A. (1997). Organic geochemical proxies of paleoceanographic, paleolimnologic, and paleoclimatic processes. Organic Geochemistry **27**(5-6): 213-250.
- Meyers, P. A., Teranes, J. L., Last, W. M. and Smol, J. P. (2002). Sediment Organic Matter: Tracking Environmental Change Using Lake Sediments, Springer Netherlands. **2:** 239-269.
- Meyers, P. A. (2003). Applications of organic geochemistry to paleolimnological reconstructions: a summary of examples from the Laurentian Great Lakes. Organic geochemistry **34**(2): 261-289.
- Meyer-Reil, L.-A. (1986). Measurement of hydrolytic activity and incorporation of dissolved organic substrates by microorganisms in marine sediments. Marine Ecology Progress Series **31**: 143-149.
- Meyer-Reil, L. A. and Koster, M. (2000). Eutrophication of Marine Waters: Effects on Benthic Microbial Communities. Marine Pollution Bulletin **41**(1): 255-263.
- Middelburg, J. J. and Levin, L. A. (2009). Coastal hypoxia and sediment biogeochemistry. Biogeosciences 6(7): 1273-1293.
- Montagna, P. A. and Froeschke, J. (2009). Long-term biological effects of coastal hypoxia in Corpus Christi Bay, Texas, USA. Journal of Experimental Marine Biology and Ecology **381, Supplement**: S21-S30.
- Mortensen, B. M., Haber, M. J., DeJong, J. T., Caslake, L. F. and Nelson, D. C. (2011). Effects of environmental factors on microbial induced calcium carbonate precipitation. Journal of Applied Microbiology **111**(2): 338-349.
- Mortimer, R. J. G., Krom, M. D., Watson, P. G., Frickers, P. E., Davey, J. T. and Clifton, R. J. (1998). Sediment-Water Exchange of Nutrients in the Intertidal Zone of the Humber Estuary, UK. Marine Pollution Bulletin 37(3-7): 261-279.
- Mouret, A., Anschutz, P., Deflandre, B., Chaillou, G., Hyacinthe, C., Deborde, J., Etcheber, H., Jouanneau, J.-M., Grémare, A. and Lecroart, P. (2010). Oxygen and organic carbon fluxes in sediments of the Bay of Biscay. Deep Sea Research Part I: Oceanographic Research Papers **57**(4): 528-540.
- Mucci, A., Starr, M., Gilbert, D., and Sundby, B. (2011). Acidification of Lower St. Lawrence Estuary Bottom Waters. Atmosphere-Ocean **49**(3): 206-218.
- Mudge, S. M. and Duce, C. E. (2005). Identifying the source, transport path and sinks of sewage derived organic matter. Environmental Pollution **136**(2): 209-220.
- Murrell, M. C. and Lehrter, J. C. (2010). Sediment and Lower Water Column Oxygen Consumption in the Seasonally Hypoxic Region of the Louisiana Continental Shelf. Estuaries and Coasts.

- Naqvi, S., Bange, H., Farias, L., Monteiro, P., Scranton, M. and Zhang, J. (2010). Marine hypoxia/anoxia as a source of CH<sub>4</sub> and N<sub>2</sub>O. Biogeosciences **7**(7): 2159-2190.
- Nedwell, D. B., Dong, L. F., Sage, A. and Underwood, G. J. C. (2002). Variations of the nutrients loads to the mainland UK estuaries: Correlation with catchment areas, urbanization and coastal Eutrophication. Estuarine Coastal and Shelf Science 54(6): 951-970.
- Neubacher, E. C., Parker, R. E., and Trimmer, M. (2011). Short-term hypoxia alters the balance of the nitrogen cycle in coastal sediments. Limnology and Oceanography, **56**(2): 651.
- Newman, S. and Reddy, K. R. (1993). Alkaline phosphatase activity in the sedimentwater column of a hypereutrophic lake. Journal of Environmental Quality **22**(4).
- Niemi, R. M. and Vepsäläinen, M. (2005). Stability of the fluorogenic enzyme substrates and pH optima of enzyme activities in different Finnish soils. Journal of Microbiological Methods **60**(2): 195-205.
- Nobre, A.M., J.G. Ferreira, A. Newton, T. Simas, J.D. Icely, R. Neves. (2005). Management of coastal eutrophication: Integration of field data, ecosystemscale simulations and screening models. Journal of Marine Systems 56:375-390.
- Norfolk County Council (2011). Minerals Site Specific Allocations & Waste Site Specific Allocations. Development Plan Documents. Habitats Regulations Assessment Appendices B and C. Regulation 61 of the Conservation of Habitats and Species Regulations 2010. Mott McDonald Ltd.
- Ohrstrom, P., Hamed, Y., Persson, M. and Berndtsson, R. (2004). Characterizing unsaturated solute transport by simultaneous use of dye and bromide. Journal of Hydrology **289**(1-4): 23-35.
- Olander, L. P. and Vitousek, P. M. (2000). Regulation of soil phosphatase and chitinase activity by N and P availability. Biogeochemistry **49**(2): 175-191.
- Olausson, E. and Cato, I. (1980). Chemistry and Biogeochemistry of Estuaries. New York, Wiley.
- Ozaki, K., Tajima, S. and Tajika, E. (2011). Conditions required for oceanic anoxia/euxinia: Constraints from a one-dimensional ocean biogeochemical cycle model. Earth and Planetary Science Letters **304**(1-2): 270-279.
- Park, K., Kim, C.-K. and Schroeder, W. (2007). Temporal variability in summertime bottom hypoxia in shallow areas of Mobile Bay, Alabama. Estuaries and Coasts 30(1): 54-65.
- Paytan, A. and McLaughlin, K. (2007). The Oceanic Phosphorus Cycle. Chemical Reviews 107(2): 563-576.
- Peers, G. S., Milligan, A. J. and Harrison, P. J. (2000). Assay Optimization And Regulation Of Urease Activity In Two Marine Diatoms. Journal of Phycology 36(3): 523-528.
- Piehler, M. F. and Smyth, A. R. (2011). Habitat-specific distinctions in estuarine denitrification affect both ecosystem function and services. Ecosphere 2(1): 1-12.
- Podgorska, B. and Mudryk, Z. J. (2003). Distribution and enzymatic activity of heterotrophic bacteria decomposing selected macromolecular compounds in a Baltic Sea sandy beach. Estuarine, Coastal and Shelf Science **56**(3): 539-546.

- Poremba, K. (1995). Hydrolytic enzymatic activity in deep-sea sediments, Blackwell Publishing Ltd. **16:** 213-222.
- Poulicek, M., Gaill, F. and Goffinet, G. (1998). Chitin Biodegradation in Marine Environments. Nitrogen-Containing Macromolecules in the Bio- and Geosphere, American Chemical Society. **707**: 163-210.
- Prokopenko, M. G., Sigman, D. M., Berelson, W. M., Hammond, D. E., Barnett, B., Chong, L. and Townsend-Small, A. (2011). Denitrification in anoxic sediments supported by biological nitrate transport. Geochimica et Cosmochimica Acta 75(22): 7180-7199.
- Quintana, C. O., Hansen, T., Delefosse, M., Banta, G. and Kristensen, E. (2011). Burrow ventilation and associated porewater irrigation by the polychaete *Marenzelleria viridis*. Journal of Experimental Marine Biology and Ecology 397(2): 179-187.
- Quintana, C. O., Tang, M. and Kristensen, E. (2007). Simultaneous study of particle reworking, irrigation transport and reaction rates in sediment bioturbated by the polychaetes *Heteromastus* and *Marenzelleria*. Journal of Experimental Marine Biology and Ecology 352(2): 392-406.
- Rabalais, N. N. (2010). Eutrophication of Estuarine and Coastal Ecosystems. Environmental Microbiology, John Wiley & Sons, Inc.: 115-135.
- Rabalais, N. N., Diaz, R. J., Levin, L. A., Turner, R. E., Gilbert, D. and Zhang, J. (2010). Dynamics and distribution of natural and human-caused hypoxia. Biogeosciences 7: 585-619.
- Rabalais, N. N., Turner, R. E. and Wiseman, W. J. (2002). Gulf of Mexico hypoxia, aka "The dead zone". Annual Review of Ecology and Systematics 33: 235-263.
- Rahmansyah, M., Antonius, S. and Sulistinah, N. (2006). Phosphatase And Urease Instability Caused By Pesticides Present In Soil Improved By Grounded Rice Straw. ARPN Journal of Agricultural and Biological Science 4(2): 56-62.
- Ramsing, N. and Gundersen, J. (2001). Seawater and Gases Tabulated physical parameters of interest to people working with microsensors in marine systems. Limnol. Oceanogr.
- Rasheed, M. (2004). Nutrient Fluxes from sediments of the northern Gulf of Aqaba under various anthropogenic activities. Lebanese Science Journal **5**: 3-16.
- Rasheed, M., Al-Rousan, S., Manasrah, R. and Al-Horani, F. (2006). Nutrient fluxes from deep sediment support nutrient budget in the oligotrophic waters of the Gulf of Aqaba. Journal of Oceanography **62**(1): 83-89.
- Rasmussen, H. and Jørgensen, B. B. (1992). Microelectrode studies of seasonal oxygen uptake in a coastal sediment: role of molecular diffusion. Mar. Ecol. Prog. Ser. 81: 289–303.
- Rasmussen, P. H., Knudsen, I. M. B., Elmholt, S. and Jensen, D. F. (2002). Relationship between soil cellulolytic activity and suppression of seedling blight of barley in arable soils. Applied Soil Ecology **19**(1): 91-96.
- Reimers, C. E., Özkan-Haller, H. T., Berg, P., Devol, A., McCann-Grosvenor, K. and Sanders, R. D. (2012). Benthic oxygen consumption rates during hypoxic conditions on the Oregon continental shelf: Evaluation of the eddy correlation method. J. Geophys. Res. **117**(C2): 18.
- Revsbech, N. P., Sorensen, J., Blackburn, T. H. and Lomholt, J. P. (1980). Distribution Of Oxygen In Marine-Sediments Measured With Microelectrodes. Limnology and Oceanography 25(3): 403-411.

- Riadas, O. and Pinkas, R. (1997). Arylsulfatase and alkaline phosphatase (APASE) activity in sediments of Lake Kinneret, Israel. Water, Air, & Soil Pollution **99**(1): 671-679.
- Riedel, B., Zuschin, M., Haselmair, A. and Stachowitsch, M. (2008). Oxygen depletion under glass: Behavioural responses of benthic macrofauna to induced anoxia in the Northern Adriatic. Journal of Experimental Marine Biology and Ecology 367(1): 17-27.
- Riedinger, N., Brunner, B., Formolo, M. J., Solomon, E., Kasten, S., Strasser, M. and Ferdelman, T. G. (2010). Oxidative sulfur cycling in the deep biosphere of the Nankai Trough, Japan. Geology 38(9): 851-854.
- Rowe, G., Kaegi, M., Morse, J., Boland, G. and Escobar Briones, E. (2002). Sediment community metabolism associated with continental shelf hypoxia, Northern Gulf of Mexico. Estuaries and Coasts **25**(6): 1097-1106.
- Rowe, G. T., Morse, J., Nunnally, C. and Boland, G. S. (2008). Sediment community oxygen consumption in the deep Gulf of Mexico. Deep Sea Research Part II: Topical Studies in Oceanography **55**(24-26): 2686-2691.
- Røy, H., Hüttel, M. and Jørgensen, B. B. (2002). The Role of Small-Scale Sediment Topography for Oxygen Flux across the Diffusive Boundary Layer. Limnology and Oceanography 47(3): 837-847.
- Rozan, T. F., Taillefert, M., Trouwborst, R. E., Glazer, B. T., Ma, S., Herszage, J., Valdes, L. M., Price, K. S. and III, G. W. L. (2002). Iron–sulfur–phosphorus cycling in the sediments of a shallow coastal bay: Implications for sediment nutrient release and benthic macroalgal blooms. Limnol. Oceanogr. 47(5): 1346–1354.
- Ruan, A., He, R., Xu, S. and Lin, T. (2009). Effect of dissolved oxygen on nitrogen purification of microbial ecosystem in sediments. Journal of Environmental Science and Health, Part A 44(4): 397-405.
- Rubao, J., C.S. Chen, W.B. Judith, J.S. David, B. Dmitry, L.F. Gary, H.J. Thomas, V. Henry, E. Brian, C. James, G. Wayne and Marie, B. (2002). Influence of suspended sediments on the ecosystem in Lake Michigan: a 3-D coupled biophysical modeling experiment. Ecological Modelling 152: 169-190.
- Sabil, N., Aissouni, Y., Coletti-Previero, M.-A., Donazzolo, R., D'Ippolito, R. and Pavoni, B. (1995). Immobilized Enzymes and Heavy Metals in Sediments of Venice Internal Canals. Environmental Technology 16(8): 765-774.
- Sabri, A. (1977). Phosphate flux in Breydon Water estuary. University of East Anglia.
- Sagasti, A., Duffy, J. D. and Schaffner, L. S. (2003). Estuarine epifauna recruit despite periodic hypoxia stress. Marine Biology **142**(1): 111-122.
- Saiya-Cork, K. R., Sinsabaugh, R. L. and Zak, D. R. (2002). The effects of long term nitrogen deposition on extracellular enzyme activity in an *Acer saccharum* forest soil. Soil Biology and Biochemistry **34**(9): 1309-1315.
- Sakamoto, K. and Toyohara, H. (2009). A comparative study of cellulase and hemicellulase activities of brackish water clam *Corbicula japonica* with those of other marine Veneroida bivalves. Journal of Experimental Biology 212(17): 2812-2818.
- Sanaullah, M., Blagodatskaya, E., Chabbi, A., Rumpel, C. and Kuzyakov, Y. (2011). Drought effects on microbial biomass and enzyme activities in the rhizosphere of grasses depend on plant community composition. Applied Soil Ecology 48(1): 38-44.

- Satoh, Y. (1980). Production of Urea by Bacterial Decomposition of Organic Matter Including Phytoplankton. Internationale Revue der gesamten Hydrobiologie und Hydrographie 65(2): 295-301.
- Sayles, F. L. and Martin, W. R. (1995). *In situ* tracer studies of solute transport across the sedimentwater interface at the Bermuda Time Series site. Deep Sea Research Part I: Oceanographic Research Papers **42**(1): 31-52.
- Schneider, B., Schlitzer, R., Fischer, G. and Nöthig, E.-M. (2003). Depth-dependent elemental compositions of particulate organic matter (POM) in the ocean. Global Biogeochem. Cycles 17(2): 1032.
- Selman, M., Zachary, S., Suzie, G. and Diaz, R. (2008). Eutrophication and Hypoxia in Coastal Areas: A Global Assessment of the State of Knowledge. WRI Policy Note(1): 1-6.
- Shand, I. (2005). The Biogeochemical Origins and Plant Availability of Potentially Toxic Elements in Sediments from Thames Estuary. Cranfield University.
- Shi, X., Qian, S., Kong, F., Zhang, M. and Yu, Y. (2011). Differences in growth and alkaline phosphatase activity between *Microcystis aeruginosa* and *Chlorella pyrenoidosa* in response to media with different organic phosphorus.
- Shivaraman, N. and Shivaraman, G. (2003). Anammox A novel microbial process for ammonium removal. Current Science 84(12):1507-1508 **84**(12): 1508.
- Shull, D. H. and Mayer, L. M. (2002). Dissolution of particle-reactive radionuclides in deposit-feeder digestive fluids. Limnol. Oceanogr. 47(5): 1530-1536.
- Silke, J., O'Beirn, F. and Cronin, M. (2005). *Karenia mikimotoi:* an exceptional dinoflagellate bloom in western Irish waters, summer 2005. Marine Environment and Health Series, No 21. Marine Institute, Marine Environment and Food Safety Services: Galway. p. 48.
- Sinsabaugh, R. L., Gallo, M. E., Lauber, C., Waldrop, M. P. and Zak, D. R. (2005). Extracellular Enzyme Activities and Soil Organic Matter Dynamics for Northern Hardwood Forests receiving Simulated Nitrogen Deposition. Biogeochemistry 75(2): 201-215.
- Sinsabaugh, R. L., Reynolds, H. and Long, T. M. (2000). Rapid assay for amidohydrolase (urease) activity in environmental samples. Soil Biology and Biochemistry 32(14): 2095-2097.
- Skoog, A. C. and Arias-Esquivel, V. A. (2009). The effect of induced anoxia and reoxygenation on benthic fluxes of organic carbon, phosphate, iron, and manganese. Science of The Total Environment 407(23): 6085-6092.
- Snajdr, J., Valášková, V., Merhautová, V., Cajthaml, T. Å. and Baldrian, P. (2008). Activity and spatial distribution of lignocellulose-degrading enzymes during forest soil colonization by saprotrophic basidiomycetes. Enzyme and Microbial Technology 43(2): 186-192.
- Sobczak, W. V., Cloern, J. E., Jassby, A. D. and Müller-Solger, A. B. (2002). Bioavailability of organic matter in a highly disturbed estuary: The role of detrital and algal resources. Proceedings of the National Academy of Sciences 99(12): 8101-8105.
- Soetaert, K., Middelburg, J. J., Heip, C., Meire, P., Damme, S. V. and Maris, T. (2006). Long-term change in dissolved inorganic nutrients in the heterotrophic Scheldt estuary (Belgium, The Netherlands). Limnol. Oceanogr. 51(1 (part 2)): 409–423.

- Soetaert, K., Middelburg, J. J., Herman, P. M. J. and Buis, K. (2000). On the coupling of benthic and pelagic biogeochemical models. Earth-Science Reviews **51**(1-4): 173-201.
- Soetaert, K., Vanaverbeke, J., Heip, C., Herman, P. M. J., Middelburg, J. J., Sandee, A. and Duineveld, G. (1997). Nematode distribution in ocean margin sediments of the Goban Spur (northeast Atlantic) in relation to sediment geochemistry. Deep Sea Research Part I: Oceanographic Research Papers 44(9-10): 1671-1683.
- Solomon, C., Collier, J., Berg, G. and Glibert, P. (2010). Role of urea in microbial metabolism in aquatic systems: a biochemical and molecular review. Aquatic Microbial Ecology **59**(1): 67-88.
- Søndergaard, M., Jensen, J. P. and Jeppesen, E. (2003). Role of sediment and internal loading of phosphorus in shallow lakes. Hydrobiologia **506-509**(1): 135-145.
- Souza, C., Almeida, B., Colwell, R. and Rivera, I. (2010). The Importance of Chitin in the Marine Environment. Marine Biotechnology **13**(5): 823-830.
- Souza, A. C. R.(2009). Activity and Kinetics of Microbial Extracellular Enzymes in Organic-Poor Sands of a South Texas Estuary. University of Texas.
- Steckbauer, A., Duarte, C. M., Carstensen, J., Vaquer-Sunyer, R. and Conley, D. J. (2011). Ecosystem impacts of hypoxia: thresholds of hypoxia and pathways to recovery. Environmental Research Letters 6(2): 1-12.
- Steenbergh, A. K., Bodelier, P. L. E., Hoogveld, H. L., Slomp, C. P. and Laanbroek, H. J. (2011). Phosphatases relieve carbon limitation of microbial activity in Baltic Sea sediments along a redox-gradient. Limnol. Oceanogr. 56(6): 2018-2026.
- Stemmer, M., Gerzabek, M. H. and Kandeler, E. (1998). Organic matter and enzyme activity in particle-size fractions of soils obtained after low-energy sonication. Soil Biology and Biochemistry **30**(1): 9-17.
- Stevens, P., Blewett, D. and Casey, J. (2006). Short-term effects of a low dissolved oxygen event on estuarine fish assemblages following the passage of hurricane Charley. Estuaries and Coasts 29(6): 997-1003.
- Sullivan, A. B., Snyder, D. M. and Rounds, S. A. (2010). Controls on biochemical oxygen demand in the upper Klamath River, Oregon. Chemical Geology 269(1-2): 12-21.
- Swiontek-Brzezinska, M., Lalke-Porczyk, E. and Donderski, W. (2007). Chitinolytic activity of bacteria and fungi isolated from shrimp exoskeletons. Oceanological and Hydrobiological Studies **36**(3): 101-111.
- Takano, Y., Mori, H., Kaneko, T., Ishikawa, Y., Marumo, K. and Kobayashi, K. (2006). Phosphatase and microbial activity with biochemical indicators in semi-permafrost active layer sediments over the past 10,000 years. Applied Geochemistry 21(1): 48-57.
- Talbot, V., Giuliano, L., Bruni, V. and Bianchi, M. (1997). Bacterial abundance, production and ectoproteolytic activity in the Strait of Magellan. Marine Ecology Progress Series 154: 293-302.
- Taylor, J. P., Wilson, B., Mills, M. S. and Burns, R. G. (2002). Comparison of microbial numbers and enzymatic activities in surface soils and subsoils using various techniques. Soil Biology and Biochemistry 34(3): 387-401.
- Taylor, G. T., Thunell, R., Varela, R., Benitez-Nelson, C. and Scranton, M. I. (2009). Hydrolytic ectoenzyme activity associated with suspended and sinking

organic particles within the anoxic Cariaco Basin. Deep Sea Research Part I: Oceanographic Research Papers **56**(8): 1266-1283.

- Tengberg, A., Almroth, E. and Hall, P. (2003). Resuspension and its effects on organic carbon recycling and nutrient exchange in coastal sediments: in situ measurements using new experimental technology. Journal of Experimental Marine Biology and Ecology 285-286(0): 119-142.
- Tengberg, A., Stahl, H., Gust, G., Müller, V., Arning, U., Andersson, H. and Hall, P. O. J. (2004). Intercalibration of benthic flux chambers I. Accuracy of flux measurements and influence of chamber hydrodynamics. Progress in Oceanography 60(1): 1-28.
- Thamdrup, B. and Dalsgaard, T. (2008). Nitrogen Cycling in Sediments. Microbial Ecology of the Oceans, John Wiley & Sons, Inc.: 527-568.
- Therkildsen, M., King, G. and Lomstein, B. (1996). Urea production and turnover following the addition of AMP, CMP, RNA and a protein mixture to a marine sediment. Aquatic Microbial Ecology **10**(2): 173-179.
- Therkildsen, M. S. and Lomstein, B. A. (1994). Seasonal variation in sediment urea turnover in a shallow estuary. Marine Ecology Progress Series **109**: 77-82.
- Turner, B. L., Hopkins, D. W., Haygarth, P. M. and Ostle, N. (2002). β-Glucosidase activity in pasture soils. Applied Soil Ecology **20**(2): 157-162.
- Turner, R. E., Rabalais, N. N. and Justic, D. (2012). Predicting summer hypoxia in the northern Gulf of Mexico: Redux. Marine Pollution Bulletin 64(2): 319-324.
- Tyler, R., Brady, D. and Targett, T. (2009). Temporal and Spatial Dynamics of Diel-Cycling Hypoxia in Estuarine Tributaries. Estuaries and Coasts **32**(1): 123-145.
- Ueda, N., H. Tsutsumi, M. Yamada, K. Hanamoto, S. Montani. (2000). Impacts of oxygen-deficient water on the macrobenthic fauna of Dokai Bay and on adjacent intertidal flats, in Kitakyushu, Japan. Marine Pollution Bulletin 40:906-913.
- Uncles, R., Joint, I. and Stephens, J. (1998). Transport and retention of suspended particulate matter and bacteria in the Humber-Ouse Estuary, United Kingdom, and their relationship to hypoxia and anoxia. Estuaries and Coasts **21**(4): 597-612.
- Vaquer-Sunyer, R. and Duarte, C. M. (2008). Thresholds of hypoxia for marine biodiversity. Proceedings of the National Academy of Sciences 105(40): 15452-15457.
- Verado, D. J., Froelich, P. N. and McIntyre, A. (1990). Determination of Organic-Carbon and Nitrogen in Marine-Sediments using the Carlo-Erba-NA-1500 Analyzer. Deep-Sea Research 37: 157-165.
- Wang, X.-C. and Qin, L. (2006). Beta-Glucosidase Activity in Paddy Soils of the Taihu Lake Region, China. Pedosphere **16**(1): 118-124.
- Ward, B. B., Capone, D. G. and Zehr, J. P. (2007). What's new in the nitrogen cycle?. Oceanography **20**(2): 109.
- Warren, R. A. J. (1996). Microbial hydrolysis of polysaccharides. Annual Review of Microbiology 50: 183-212.
- Washbourne, I., Crenshaw, C. and Baker, M. (2011). Dissimilatory nitrate reduction pathways in an oligotrophic freshwater ecosystem: spatial and temporal trends. Aquatic Microbial Ecology **65**(1): 55-64.

- Wenner, E., Sanger, D., Arendt, M., Holland, A. F. and Chen, Y. (2004). Variability in Dissolved Oxygen and Other Water-Quality Variables Within the National Estuarine Research Reserve System. Journal of Coastal Research: 17-38.
- Wilczek, S., Fischer, H. and Pusch, M. (2005). Regulation and Seasonal Dynamics of Extracellular Enzyme Activities in the Sediments of a Large Lowland River. Microbial Ecology 50(2): 253-267.
- Wild, C., Rasheed, M., Jantzen, C., Cook, P., Struck, U., Huettel, M. and Boetius, A. (2005). Benthic metabolism and degradation of natural particulate organic matter in carbonate and silicate reef sands of the northern Red Sea. Marine Ecology Progress Series 298: 69-78.
- Wirth, S. J. and Wolf, G. A. (1992). Micro-plate colourimetric assay for Endo -acting cellulase, xylanase, chitinase, 1,3-β-glucanase and amylase extracted from forest soil horizons. Soil Biology and Biochemistry **24**(6): 511-519.
- Witte, U., Wenzhofer, F., Sommer, S., Boetius, A., Heinz, P., Aberle, N., Sand, M., Cremer, A., Abraham, W. R., Jorgensen, B. B. and Pfannkuche, O. (2003). *In situ* experimental evidence of the fate of a phytodetritus pulse at the abyssal sea floor. Nature **424**(6950): 763-766.
- Wittmann, C., Kähkonen, M. A., Ilvesniemi, H., Kurola, J. and Salkinoja-Salonen, M. S. (2004). Areal activities and stratification of hydrolytic enzymes involved in the biochemical cycles of carbon, nitrogen, sulphur and phosphorus in podsolized boreal forest soils. Soil Biology and Biochemistry 36(3): 425-433.
- Wojewodzic, M., Kyle, M., Elser, J., Hessen, D. and Andersen, T. (2011). Joint effect of phosphorus limitation and temperature on alkaline phosphatase activity and somatic growth in *Daphnia magna*. Oecologia **165**(4): 837-846.
- Wright, A. L. and Reddy, K. R. (2001). Phosphorus Loading Effects on Extracellular Enzyme Activity in Everglades Wetland Soils. Soil Sci. Soc. Am. J. 65(2): 588-595.
- Wu, R. S. S. (2002). Hypoxia: from molecular responses to ecosystem responses. Marine Pollution Bulletin 45(1-12): 35-45.
- Yiyong, Z., Song, C., Cao, X., Li, J., Chen, G., Xia, Z. and Jiang, P. (2008). Phosphorus fractions and alkaline phosphatase activity in sediments of a large eutrophic Chinese lake (Lake Taihu). Hydrobiologia **599**(1): 119-125.
- Zantua, M. I. and Bremner, J. M. (1977). Stability of urease in soils. Soil Biology and Biochemistry **9**(2): 135-140.
- Zhang, T., Wang, X. and Jin, X. (2007). Variations of alkaline phosphatase activity and P fractions in sediments of a shallow Chinese eutrophic lake (Lake Taihu). Environmental Pollution **150**(2): 288-294.
- Zhou, Y., Song, C., Cao, X., Li, J., Chen, G., Xia, Z. and Jiang, P. (2008). Phosphorus fractions and alkaline phosphatase activity in sediments of a large eutrophic Chinese lake (Lake Taihu). Hydrobiologia **599**(1): 119-125.
- Ziervogel, K. and Arnosti, C. (2009). Enzyme activities in the Delaware Estuary affected by elevated suspended sediment load. Estuarine, Coastal and Shelf Science **84**(2): 253-258.
- Zimmerman, A. R. and Canuel, E. A. (2000). A geochemical record of eutrophication and anoxia in Chesapeake Bay sediments: anthropogenic influence on organic matter composition. Marine Chemistry **69**(1-2): 117-137.

Zonneveld, K. A. F., Versteegh, G. J. M., Kasten, S., Eglinton, T. I., Emeis, K. C., Huguet, C., Koch, B. P., Lange, d. G. J., Leeuw, D. J. W., Middelburg, J. J., Mollenhauer, G., Prahl, F. G., Rethemeyer, J. and Wakeham, S. G. (2010). Selective preservation of organic matter in marine environments; processes and impact on the sedimentary record. Biogeosciences 7: 483 - 511.